Effects of changes in land use and climate on water availability of a tropical catchment

Ī

Hero Marhaento

Effects of changes in land use and climate on water availability of a tropical catchment

Hero Marhaento

Graduation committee:

prof. dr. G.P.M.R. Dewulf	University of Twente, chairman and secretary
prof. dr. ir. A.Y. Hoekstra	University of Twente, promotor
dr. ir. M.J. Booij	University of Twente, co-promotor
dr. ing. T.H.M. Rientjes	University of Twente, co-promotor
prof. dr. ir. J.C.J. Kwadijk	University of Twente
prof. dr. V.G. Jetten	University of Twente
prof. dr. ir. R. Uijlenhoet	Wageningen University & Research
prof. dr hab. R.J. Romanowicz	Polish Academy of Sciences
univprof. P. Reggiani, Ph.D.	University of Siegen

This research was funded by DIKTI-scholarship from Directorate General of Higher Education, Ministry of Research, Technology and Higher Education of the Republic of Indonesia and supported by University of Twente, The Netherlands and Universitas Gadjah Mada, Indonesia.

Cover design: Hero Marhaento

Copyright © 2018 by Hero Marhaento, Enschede, The Netherlands

All right reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, without the prior permission in writing from the proprietor.

Printed by Ipskamp Printing, Enschede

ISBN: 978-90-365-4491-7 DOI: 10.3990/1.9789036544917 URL: https://doi.org/10.3990/1.9789036544917

EFFECTS OF CHANGES IN LAND USE AND CLIMATE ON WATER AVAILABILITY OF A TROPICAL CATCHMENT

DISSERTATION

to obtain the degree of doctor at the University of Twente, on the authority of the rector magnificus, prof. dr. T.T.M. Palstra, on account of the decision of the graduation committee, to be publicly defended on Thursday 22 February 2018 at 14:45 hrs

by

Hero Marhaento

Born on April 5th, 1982 in Yogyakarta, Indonesia This dissertation has been approved by:

prof. dr. ir. A.Y. Hoekstrapromotordr. ir. M.J. Booijco-promotordr. ing. T.H.M. Rientjesco-promotor

Verily, after hardship comes ease [Quran 94:5]

I dedicate this book to my parents, wife and son

Contents

Ackno	wledgem	nents	xi
Summ	ary		xv
Same	nvatting .		xviii
Chapte	er 1. Intro	oduction	1
1.1.	Backord	bund	. 1
1.2.	Research objective		
1.3.	Researc	ch questions	3
1.4.	Researc	ch approach	4
1.5.	Thesis of	outline	5
Chapte	er 2: Attri	ibution of changes in streamflow to land use change and	
climate	e change	e in a mesoscale tropical catchment in Java. Indonesia	7
Abstra	ct		7
2.1.	Introduc	tion	8
2.2.	Study a	dv area and data availability	
	2.2.1.	Catchment description	9
	2.2.2.	Discharge data	10
	2.2.3.	Precipitation and climatological data	12
	2.2.4.	Spatial data	13
2.3.	Methods	s	13
	2.3.1.	Separating effects of land use and climate change on	
		streamflow	13
	2.3.2.	Validation of attribution assessment	17
		2.3.2.1. Trend analysis of climate variables	18
		2.3.2.2. Land use change analysis	18
2.4.	Results		19
	2.4.1.	Attribution of changes in streamflow to land use change	
		and climate change	19
	2.4.2.	Trend analysis of climate variables	20
	2.4.3.	Land use change detection	21
2.5.	Discuss	ion	22
2.6.	Conclus	sions	25
Chapte	er 3: Attri	ibution of changes in the water balance of a tropical	
catchn	nent to la	and use change using the SWAT model	27
Abstra	ct		27
3.1.	Introduc	tion	28
3.2.	Study a	rea and data	30
	3.2.1.	Catchment description	30
	3.2.2.	Hydro-climatic conditions	31

	3.2.3.	Data colle	ction	31
3.3.	Method	s		33
	3.3.1.	Land use	changes	33
	3.3.2.	SWAT mo	odel set up and calibration	34
		3.3.2.1.	Water balance estimation	34
		3.3.2.2.	Model setup	34
		3.3.2.3.	Model calibration	36
	3.3.3.	SWAT sin	nulations	37
3.4.	Results			39
	3.4.1.	Land use	change analysis	39
	3.4.2.	Model cal	ibration and simulations	40
		3.4.2.1.	Model calibration	40
		3.4.2.2.	Performance of model simulations with and	
			without land use changes	42
	3.4.3.	Changes	in water balance ratios	43
	3.4.4.	Effect of la	and use change on the water balance	46
3.5.	Discuss	ion	-	46
3.6.	Conclus	sions		50
Chant	or 1. Llug	Irological r	concerns to future land use change and elimete	
chap	lei 4. пус	nicol octob	esponse to ruture land use change and climate	51
Abotr	je in a tro	ipical catch		51
ADSIL	Introduc			51
4.1.	Study A	1001		52
4.2.		Cotobmon	ata Availability	54 54
	4.2.1.			54
12	4.Z.Z.			55
4.3.			abanga madal	50
	4.3.1.			50
		4.3.1.1.	Business as usual (BAO) scenario	57
	400	4.3.1.Z.		60
	4.3.Z.			01
	4.3.3.			01
	4.3.4.	Future nyo	arological responses to land use and climate	~~~
	Deculto	change so		62
4.4.	Results			63
	4.4.1.	Land use		63
	4.4.2.	Climate ci	hange	65
	4.4.3.	Future nyo		66
		4.4.3.1.	Effects of land use change	66
		4.4.3.2.	Effects of climate change	67
		4.4.3.3.	Combined effects of land use change and	
			climate change	69

4.5.	Discus	sion	
	4.5.1.	Effects of future land use change	
	4.5.2.	Effects of future climate change	
	4.5.3.	Combined effects of future land use change and climate	
	4.5.4.	Limitations and uncertainties	
4.6.	Conclu	isions	
Chap	ter 5: Se	nsitivity of streamflow characteristics to different spatial	
land	use confi	igurations in a tropical catchment	
Abstr	act		
5.1.	Introduction		
5.2.	Study a	area and data availability	
	5.2.1.	Catchment description	
	5.2.2.	Data availability	
5.3.	Method	۲	
	5.3.1.	Reconstruction of past land use change	
	5.3.2.	Hydrological model simulations	
	5.3.3.	Changes in land use patterns and streamflow	
	5 24	Simulation of land use scenarios	
E /	Docult		
5.4.		Past land use evolution	
	5.4.1.	Correlations between land use patterns and streamflow	
	5.4.Z.	correlations between land use patterns and streamnow	
	512	Characteristics	
	5.4.5.		
F F	Diagua		
5.5. 5.6	Conclus		
5.0.	Conciu	ISIONS	
Chap	ter 6: Co	onclusions	
6.1.	Overview of the main findings		
6.2.	Limitat	ions and future outlook	
6.3.	Contrib	oution to scientific advancement	
Refer	rences		
Abou	t the auth	hor	
List o	f publica	tions	

Acknowledgements

When you do a PhD and start writing this section, you must realize that your seemingly never-ending PhD journey is almost over. For me, it feels like hearing your child crying for the first time in the delivery room; thrilled yet very relieved! ^(C) Thus, this is time to think of and to thank all the amazing people who guided and helped me during this roller-coaster journey.

My first acknowledgement goes to my daily supervisor dr. ir. Martijn J. Booij. Martijn, our first meeting was in the training and selection workshop of DIKTIscholarship held by LabMath-Indonesia in Bandung in January 2013. When you mentioned my name as one of the two candidates who got the scholarship, that was a really life-changing moment, not only for me but also for my family, for which I am truly indebted to you. A 'new life' as a PhD candidate began in September 2013, which was quite challenging since I came from different scientific background. Long story short, I could not have accomplished all this without your relentless support, guidance and scientific intuitive solutions, for which I am truly thankful. Indeed it has really been a pleasure working with you.

I am deeply grateful to my supervisor and promotor, prof. dr. ir. Arjen Y. Hoekstra. Arjen, your high quality standard on academic research has pushed me forward. Without your endless motivation and support, this dissertation might not be completed. I enjoyed our discussions related to the research, and the small talk about many things during lunch time. Thank you for all enthusiasm, trust and constructive comments.

I would like to express my great appreciation to my third supervisor, dr. ing. Tom H.M. Rientjes. Tom, your knowledge and scientific experience is the basis of the results obtained in this dissertation. The intensive discussions with you and Martijn, especially when we were working on the second paper will never be forgotten. My knowledge and skills on hydrological modelling have enhanced significantly ever since. For this, I thank for your immense support and constructive advice.

This thesis would never exist without the help of several people in particular during data collections: mbak Vicky from Bengawan Solo River Basin Organization (BBWS Bengawan Solo), mas Sigit from Bengawan Solo Watershed Management Agency (BP DAS Bengawan Solo) and mas Sena from Indonesian Agency for Meteorological, Climatological and Geophysics (BMKG). My gratitude also goes to Prima, Saddam and Wayan, the three-musketeers who always gave me extra hands whenever I needed help with the field data, while I was thousands miles away from the study area. I would also thank to all colleagues in the Faculty of Forestry, Universitas Gadjah Mada especially mas

Oka, mas Imron, bu Lies, mas Taufik and mbak Fajar for their valuable support during my research.

The research was possible due to funding from DIKTI-scholarship from the Directorate General of Higher Education, Ministry of Research, Technology and Higher Education of the Republic of Indonesia, supported by University of Twente (UT), Universitas Gadjah Mada (UGM), and LabMath-Indonesia (LMI). I would like to convey my sincere thanks to all who helped me through the administrative procedures regarding the funding. Special recognitions are aimed to prof. Brenny from LMI, mbak Rifa and pak Rohman from Faculty of Forestry UGM, bu Emmy and pak Gugup from the rectorate office of UGM.

It is my pleasure to acknowledge all my colleagues in the Water Engineering and Management (WEM) group. The list can be very long, thus I will mention just a few. My sincere gratitude goes to my comrade Andry and his family for their warm friendship. My special thanks go to my paranimfen: Hatem and Pepijn, who helped me during preparation of my defense, and also to Anouk who helped me with the samenvatting. Thanks go to my former office mates: Jane, Zhuo La and Alejandro, and currently: Pepijn and Anouk. I will never forget the time we shared stories about many things like football, food, weather and supervisors ©. I really appreciate our friendship. I shared duties handling the group lunch talk together with Johan and Matthijs, thanks guys! In addition, I always enjoyed the time when I had small conversations outside of research with Abebe, Hatem, Joep S., Rick, Marcella, Maarten, Charlotte, Hamideh, Bunyod, Lara, Louisa, Ranran, Guori, Tariq, Filipe, Leonardo, Michiel, Juliette, Denie, Pieter and others. And, of course, my sincere gratitude to Joke for helping me with all the scholarship letters, reimbursements, and other things needed to support my research. I also thank Anke and Monique for their kind support. I will surely miss the WEM group gatherings such as the group outings, the Christmas lunches and the birthday celebrations.

Outside from the scientific environment, the life in the Netherlands was very enjoyable with close friendships from many Indonesian people and communities. The list is exhaustive!

I am indebted to thank the Indonesian Students Association in Enschede (PPIE), whom helped me and family during the first arrival and to settle in: pak Yusuf, mbak Ratna, pak Rahman, pak Nasrullah, pak Bayu, and all ITC'ers 2013. I would like to thank for their supports all Indonesian PhD students in UT: mas/mbak Andry, Aji, Aulia, Sunu, Taufiq, Dhadhang, Ari, Nicco, Habib, Muthia, Khafid, Riswan, Dewi, Novi, Heksi, Ifha, Dwi M., Rizal, Dwi, Deby, Kunaifi, Kamia, Lulu, Pesi, Nova, Diah, Irena, Ayu, Dwi R., Jarot and others. I wish all of you a successful PhD journey. Also, thanks to Apri, Aris, Aufar, Rusydi, Rindia, Nadia, Acny, Akbar, Aden, Yosia, Yasir, Faisal, Fajar Eko, Adlan, Wildan, Giri,

Okky, Gibran, Ihrom, and many more for their cordial friendship during my stay in Enschede.

I am grateful for the friendship in the Universitas Gadjah Mada Alumni Association in the Netherlands (Kagama-NL), which I chaired the organization in 2 years (2015-2017). I and my family feel very comfortable whenever we had a gathering with the Kagama-NL. I really appreciate the sincere friendship with all of you: mas/mbak Richo, Erlis, Ari S, Aryanti, Dita, Erda, Bambang, Dian, Yayok, Icha, Nanang, Yuda, Hakim, Tiwi, Metta, Didik, Iwan, Agung, Tulus, Pugo, Mila, Nur, Arie, and many more. I also appreciate the support from the Indonesian Embassy for Kagama-NL especially from the ambassador H.E. I Gusti Agung Wesaka Puja, the Deputy Chief of Mission pak Ibnu Wahyutomo, attaché of Educational and Cultural Affairs prof. Bambang Hari Wibisono, former chief of Information and Cultural Affairs mas Azis Nurwahyudi, and mas Usman.

Finally, I dedicate this work to my beloved family. I am highly indebted to my parents: bapak Dr. Noorhadi Rahardjo and ibu Tri Sudarminah, and my two sisters: dek Asty and dek Cahya. I could not have become what I am now without your love and care, for which I feel so blessed. My sincere gratitude also goes to my parents in law: bapak Suwardi and ibu Suradiyah for their support and never ending prayers. To my wife, Astuti Tri Padmaningsih and my son, Arga Danadyaksa. O Lord, I am speechless. Astuti, you are the reason I dare to take this journey. It was not an easy decision when we decided to leave our hometown moving to a new place in a completely different environment, especially when everything was settled: jobs, house, families and many more. It was a really tough life during the first year we stayed in Enschede as we both were (very) busy with our education and at the same time took care of our son, Arga. But, with all your commitment and sacrifice, we were able to overcome all the difficulties. Moreover, I was very proud when you ultimately succeeded finishing your MSc degree with cum laude in ITC!. I am so lucky to have you always by my side; thank you for being my other half. To my adorable son, Arga, you might be too young to understand how this journey changed our life. Keep shining dude, take all the benefits from this incredible 4.5 year experience living far away from our hometown. But, keep your seatbelt fasten, because our wonderful time will continue, Insyaa Allah.

Beyond everything, I must acknowledge and thank Almighty Allah for giving me the opportunity to undertake this PhD and to persevere and complete it satisfactorily. Without His blessings, this achievement would not be possible.

Hero Marhaento

Summmary

Introduction. Land use changes such as deforestation and urbanization influence the hydrology of catchments and hence water availability. Together with climate change, land use changes can affect the frequency of floods or droughts and thus threaten local or regional socio-economic development. For Indonesia, the effects of changes in land use and climate have been projected to cause a food crisis and eventually increase the degree of poverty in the future. In order to mitigate future risks, knowledge on the extent and directions of land use change and climate change impacts on water availability is essential.

The objective of this research is to assess the effects of land use change and climate change on the water availability in the Samin catchment (278 km²) in Java, Indonesia. The research is divided into four parts (papers). The first two papers address the attribution of observed changes in hydrological processes to land use change and climate change, using a data-based approach (first paper) and a modelling approach (second paper). In the third paper, the future hydrological response due to expected land use change and climate change is estimated. Finally, in the fourth paper, the effects of different land use patterns on streamflow characteristics are assessed.

Attribution of changes in streamflow to land use change and climate change in a mesoscale tropical catchment in Java, Indonesia. Changes in the streamflow of the study catchment have been attributed to land use change and climate change using a method based on the changes in the proportion of excess water relative to changes in the proportion of excess energy. The results show that 72% of the increase in streamflow in the period 1990-2013 can be attributed to land use change and 28% to climate change. The results were corroborated by statistical trend analyses (i.e. Mann-Kendall trend analysis and Sen's slope estimator) and land use change analysis based on two Landsat imageries for 1994 and 2013. The results of the statistical trend analysis are in the same direction as the results of the attribution analysis, where climate change was relatively small compared to the significant land uses changes due to deforestation during the period 1994-2013. It was concluded that changes in streamflow in the study catchment can be mainly attributed to land use change rather than climate change.

Attribution of changes in the water balance of a tropical catchment to land use change using the SWAT model. Changes in the water balance of the study catchment have been attributed to land use change using the Soil Water Assessment Tool (SWAT) and a baseline-altered method. The simulation period 1990–2013 was divided into four equal periods to represent baseline conditions and altered land use conditions. A SWAT model was calibrated for the baseline

period and applied to the altered periods with and without land use change acquired from Landsat and Aster satellite imageries for the years 1994, 2000, 2006 and 2013. The results show that the model performance for simulations with land use change is better than the model performance without land use change, confirming that land use change is an explanatory factor for observed changes in the water balance. Land use changes during 1994–2013, which included a decrease in forest area from 48.7% to 16.9%, an increase in agriculture area from 39.2% to 45.4% and an increase in settlement area from 9.8% to 34.3%, have resulted in an increase of the ratio of streamflow to precipitation from 35.7% to 44.6%, a decrease in the ratio of surface runoff to streamflow from 26.6% to 37.5% and a decrease in the ratio of base flow to streamflow from 40% to 31.1%. At sub-catchment level, the effects of land use changes on the water balance varied in different sub-catchments depending on the scale of changes in forest and settlement areas.

Hydrological response to future land use change and climate change in a tropical catchment. Future hydrological response in the study catchment has been simulated using the validated SWAT model, predicted land use distributions for the period 2030-2050, and outputs from a bias-corrected Regional Climate Model (RCM) and six Global Climate Models (GCMs) to include climate model uncertainty. Two land use change scenarios, namely a business-as-usual (BAU) scenario, where no measures are taken to control land use change, and a controlled (CON) scenario, where the future land use follows the land use planning, were used in the simulations together with two climate change scenarios, namely Representative Concentration Pathways (RCPs) 4.5 and 8.5. It was predicted that, in the period 2000-2050, settlement and agriculture areas of the study catchment will increase by 33.9% and 3.5%, respectively, under the BAU scenario, whereas agriculture area and evergreen forest will increase by 15.2% and 10.2%, respectively, under the CON scenario. In comparison to the baseline conditions (1983-2005), the mean annual maximum and minimum temperature in 2030-2050 are expected to increase by an average of +1°C, while predicted changes in the mean annual precipitation range from -20% to +19% under RCP 4.5 and from -25% to +15% under RCP 8.5. The results show that land use change and climate change individually will cause changes in the water balance components, but that more pronounced changes are expected if the drivers are combined in the modelling approach, in particular for changes in annual streamflow and surface runoff. Furthermore, under the CON scenario the annual streamflow and surface runoff could be potentially reduced by up to 10% and 30%, respectively indicating the effects of applied land use planning.

Sensitivity of streamflow characteristics to different spatial land use configurations in a tropical catchment. While the previous chapters have focused on the impact of changes in types of land use on hydrological processes, this chapter assesses the effects of changes in the spatial configuration of land use on streamflow characteristics. The land use distribution in the period 1982-2013 was reconstructed based on satellite images and used to estimate land use pattern characteristics for the years 1982, 1994, 2000, 2006 and 2013. A validated SWAT model was employed using the land use distributions for the mentioned years as inputs and a correlation analysis was applied to identify relations between changes in land use pattern characteristics and simulated streamflow characteristics. Furthermore, a future land use scenario analysis was carried out to assess the sensitivity of streamflow characteristics to different land use patterns. The results show that changes in the percentages of different land use types and the physical connectivity between patches of similar land use types appear as dominant drivers for changes in streamflow characteristics. When the relative presence of different land use types is fixed but the physical connectivity of patches is changed, simulation results indicate that an increase in the settlement connectivity can result in an increase in the ratio of surface runoff to streamflow and a decrease in the ratio of dry-season streamflow to wet-season streamflow, and vice versa, while changes in the forest connectivity have less impact on streamflow characteristics.

Conclusions. The research provides insight in the impacts of changes in land use and climate on water availability in a tropical catchment in the past and future. Beside contributing to the advancement of the scientific field of hydrology through proposed methods used in this thesis, findings of this study can be useful for supporting the authorities for mitigating future water stress through appropriate land use management.

Samenvatting

Introductie. Veranderingen in landgebruik zoals ontbossing en verstedelijking beïnvloeden de hvdroloaie van stroomgebieden en daarmee de waterbeschikbaarheid. Samen met klimaatveranderingen kunnen veranderingen in landgebruik invloed hebben op de frequentie van desastreuze gebeurtenissen (bijv. overstromingen en droogte), wat kan leiden tot bedreigingen voor lokale, regionale en globale socio-economische ontwikkeling. In Indonesië zijn de gevolgen van veranderingen in landgebruik en klimaat de oorzaken van een voedselcrisis wat uiteindelijk resulteert in een toename van de mate van armoede in de toekomst. Kennis over de grootte en richting van landgebruiks- en klimaatveranderingen is noodzakelijk toekomstige om risico's op waterbeschikbaarheid te beperken.

De doelstelling van dit onderzoek is om de effecten van veranderingen in landgebruik en klimaat op waterbeschikbaarheid in het stroomgebied van de Samin (278 km²) in Java. Indonesië, te beoordelen. Dit onderzoek is opgesplitst in vier delen (artikelen). De eerste twee artikelen richten zich op de toekenning van waargenomen veranderingen in hydrologische processen op veranderingen in landgebruik en klimaat, gebruikmakende van een data-gedreven benadering (eerste artikel) en een modelleerbenadering (tweede paper). In het derde artikel worden de toekomstige hydrologische reacties ten gevolge van verwachte landgebruiks- en klimaatveranderingen geschat. In het vierde artikel worden de effecten verschillende landgebruik van patronen van op de afvoerkarakteristieken beoordeeld.

Toekenning van veranderingen in afvoer aan landgebruiksveranderingen en klimaatveranderingen in een mesoschaal stroomgebied in Java, Indonesië. Veranderingen in de afvoer van het studiegebied zijn toegekend aan landgebruiks- en klimaatveranderingen gebruikmakende van een methode gebaseerd op de veranderingen in het aandeel van overtollig water ten opzichte van veranderingen in het aandeel van overtollige energie. De resultaten laten zien dat 72% van de toename in afvoer in de periode 1990-2013 kan worden toegeschreven aan veranderingen in landgebruik en 28% kan worden toegekend aan klimaatverandering. De resultaten werden bevestigd door statische trendanalyses (Mann-Kendall trend analyse en Sen's slope estimator) en een analyse van landgebruiksveranderingen gebaseerd op Landsat satelietbeelden uit 1994 en 2013. De resultaten van de statische trendanalyses komen overeen met de resultaten van de toekenningsanalyse waar klimaatveranderingen relatief klein waren in vergelijking met de significante landgebruiksveranderingen ten gevolge van ontbossing in de periode 1994-2013. Er werd geconcludeerd dat de veranderingen in afvoer in het studiegebied voornamelijk kunnen worden toegeschreven aan de veranderingen in landgebruik in plaats van klimaatveranderingen.

Toekenning van veranderingen in de waterbalans van een tropisch stroomgebied aan landgebruiksveranderingen gebruikmakende van het SWAT model. Veranderingen in de waterbalans van het studiegebied zijn toegekend aan landgebruiksveranderingen gebruikmakende van de Soil Water Assessment Tool (SWAT) en een gewijzigde baseline methode. De 1990-2013 simulatieperiode was opgesplitst in vier gelijke periodes om de baseline condities en gewijzigde condities in landgebruik weer te geven. Een SWAT model was gekalibreerd en gevalideerd voor de baseline periode en toegepast op de gewijzigde periodes met en zonder landgebruiksveranderingen verkregen van Landsat en Aster satellietbeelden voor de jaren 1994, 2000, 2006 en 2014. De modelprestaties resultaten laten zien dat voor de simulaties met zijn landgebruiksveranderingen beter dan de modelprestaties zonder veranderingen in landgebruik. waarmee wordt bevestiad dat landgebruiksveranderingen een verklarende factor is voor de waargenomen veranderingen in de waterbalans. Veranderingen in landgebruik tijdens de periode 1994-2013 met een afname van bosgebieden van 48,7% naar 16,9%, een toename in landbouwgebieden van 39,2% naar 45,4% en een toename in verstedelijking van 9,8% naar 34,4%, resulteerde in een toename van de ratio van afvoer tot neerslag van 35,7% naar 44,6%, een afname van de ratio van evapotranspiratie tot neerslag van 60% naar 54,8%, een toename van de ratio van oppervlakte-afvoer tot afvoer van 26.6% naar 37.5%, en een afname van de ratio basisafvoer tot afvoer van 40% tot 31,1%. De effecten van landgebruiksveranderingen op de waterbalans varieerden voor de verschillende deelstroomgebieden afhankelijk van de mate van veranderingen in bosgebieden en verstedelijking.

Hydrologische reactie op toekomstige veranderingen in landgebruik en klimaat in een tropisch stroomgebied. De toekomstige hydrologische reactie in het studiegebied is gesimuleerd met behulp van het gevalideerde SWAT model. Een voorspelde verdeling van landgebruik in de periode 2030-2050 is gebruikt als invoer, en uitvoer van een bias gecorrigeerd Regional Climate Model (RCM) en uitvoer van zes Global Climate Models(GCMs) zijn gebruikt om de onzekerheid verdisconteren. van het klimaatmodel te Twee landgebruikscenario's, namelijk een business-as-usual (BAU) scenario waarin geen maatregelen zijn genomen om landgebruiksveranderingen te beheersen, en een gecontroleerde (CON) scenario waarin toekomstige landgebruik de landgebruiksplanning volgt, waren gebruikt in de simulaties samen met twee klimaatveranderingscenario's, namelijk de Representative Concentration Pathways (RCPs) 4.5 en 8.5. Er werd voorspeld dat in 2050 verstedelijking en

landbouwgrond in het studiegebied met respectievelijk 33,9% en 3,5% zullen toenemen in het BAU scenario en landbouwgrond en groenblijvend bos zullen toenemen met respectievelijk 15,2% en 10,2% in het CON scenario. In vergelijking met de baseline condities (1983-2005) zal de verwachte gemiddelde jaarlijkse maximum en minimum temperatuur in 2030-2050 toenemen met een gemiddelde van +1°C, terwijl veranderingen in de gemiddelde jaarlijkse neerslag variëren tussen -20% en +19% onder RCP 4.5 en tussen -25% en +15% onder RCP 8.5. De resultaten laten zien dat veranderingen in zowel landgebruik als in klimaat leiden tot veranderingen in de waterbalanscomponenten, maar de verwachte veranderingen zijn groter als beide oorzaken worden gecombineerd, met name voor veranderingen in de jaarlijkse afvoer en oppervlakte-afvoer. Daarnaast kunnen de jaarlijkse afvoer en oppervlakte-afvoer mogelijk afnemen met respectievelijk 10% tot 30% in het CON scenario.

Gevoeligheid van afvoerkarakteristieken voor verschillende ruimtelijke landgebruikconfiguraties in een tropisch stroomgebied. Terwijl in de vorige hoofdstukken de focus was op het effect van veranderingen in type landgebruik op hydrologische processen, beoordeelt dit hoofdstuk het effect van ruimtelijke configuraties veranderingen in de van landgebruik op afvoerkarakteristieken. De landgebruiksverdeling uit de periode 1982-2013 was gereconstrueerd met behulp van satellietbeelden en gebruikt om een schatting te maken van de karakteristieken van landgebruikspatronen voor de jaren 1982, 1994, 2000, 2006 en 2013. Een gevalideerd SWAT model was toegepast gebruikmakende van de landgebruiksverdelingen voor de genoemde jaren als invoer en een correlatieanalyse was toegepast om de relaties tussen veranderingen in karakteristieken van landgebruikspatronen en gesimuleerde afvoerkarakteristieken bepalen. Daarnaast was toekomstige te een landgebruiksscenario analvse uitaevoerd de aevoeliaheid van om afvoerkarakteristieken voor verschillende landgebruikspatronen te bepalen. De resultaten laten zien dat veranderingen in de percentages van verschillende landgebruikstypes en de fysische verbinding tussen gebieden met hetzelfde landgebruik de meest dominante oorzaken blijken te zijn voor veranderingen van de afvoerkarakteristieken. Uit simulaties blijkt dat wanneer het relatieve voorkomen van verschillende landgebruikstypes is vastgelegd, maar de fysische locaties van de gebieden zijn veranderd, een toename in verbindingen van stedelijke gebieden kan leiden tot een toename in de ratio van oppervlakteafvoer tot afvoer en tot een afname van de ratio van de afvoer in het droge seizoen tot de afvoer in het natte seizoen en vice versa, terwijl veranderingen in verbindingen van bosgebieden minder impact hebben op de afvoerkarakteristieken.

Conclusie. Het onderzoek geeft inzicht in het effect van veranderingen in landgebruik en klimaat op waterbeschikbaarheid in een tropisch stroomgebied voor het verleden en de toekomst. Daarnaast draagt het bij aan de vooruitgang van het wetenschappelijke inzicht op het gebied van hydrologie door middel van de gebruikte methodes in dit proefschrift en kunnen de bevindingen van deze studie nuttig zijn om autoriteiten te ondersteunen voor het verzachten van toekomstige waterstress door middel van passende ruimtelijke ordening.

Chapter 1 Introduction

1.1. Background

Effects of land use change and climate change on the local, regional, and global water balance have been of interest among hydrologists for decades. Numerous studies have been carried out to quantify and simulate the effects of land use change on hydrological processes in catchments under various climatic conditions using methods such as paired catchment studies (i.e. comparison studies between control catchments and catchments under treatment) (e.g. Bosch and Hewlett, 1982; Brown et al., 2005; Suryatmojo et al., 2011; Zhao et al., 2012), a single model framework (e.g. Niehoff et al., 2002; Ashagrie et al., 2006; Breuer et al., 2009; Menzel et al., 2009; Suarez et al., 2013) and multiple models (ensemble modeling) (e.g. Huisman et al., 2009). Most studies found that, at the local scale, effects of land use change on hydrological processes are more pronounced than the effects of climate change (Bosch and Hewlett, 1982; Brown et al., 2005, O'connell et al., 2007; Wohl et al., 2012; Gallo et al., 2015). For large scales (>100 km²), the effects of land use change on hydrological processes are often contradictory and inconsistent because of e.g. climatic interference (Calder et al., 2001; Blöschl et al., 2007; Beck et al., 2013). Several studies show that the combination of land use and climate changes does not necessarily result in accelerating changes in hydrological processes (Legesse et al., 2003; Khoi & Suetsugi, 2014), since the effects of both drivers may also offset each other (Zhang et al, 2016). In addition, feedback mechanisms between land use change and climate change may be operative in larger catchments, however to what extent and in which direction is not very well understood (Pielke, 2005; Blöschl et al., 2007; Wohl et al., 2012).

One of the major challenges in the study of hydrology is to assess the attribution of changes in water availability to land use and climate changes (Romanowicz & Booij, 2011). A widespread belief exists in the hydrological community that forest reduction due to agricultural and settlement developments is the main cause of an increasing number of disasters (floods and droughts) (Andréassian, 2004; Marfai et al., 2008). It is generally agreed that forest reduction may significantly reduce canopy interception and soil infiltration capacity, resulting in a larger fraction of precipitation being transformed into surface runoff rather than recharge to groundwater (Bruijnzeel, 1989; 2004; Ibanez et al., 2002; Ogden et al.; 2013). However, some studies in tropical regions found that the influence of climate change (particularly changes in temperature and precipitation) on water

availability in catchments is larger than the influence of land use change (Legesse et al., 2003; Khoi & Suetsugi, 2014). In addition, several reports argue that forests act like a 'pump' that consumes (evaporates) a lot of water during the dry season rather than like a 'sponge' that supplies water to streamflow in the dry season (Calder et al., 1998; Calder, 2001). These findings call for further investigations on the effects of land use change and climate change on hydrological processes of catchments. Knowledge on the relative impacts of land use change and climate change on water availability can provide a better understanding of the single effect of land use change and climate change, and thus will be helpful in estimating the effectiveness of land use management practices at landscape level.

Hydrology in tropical regions differs from other regions in having greater energy inputs with large spatial and temporal variability and a fast rate of change including human-induced change (Wohl et al., 2012). Several studies have projected that land use and climate conditions of tropical regions in the future are generally characterized by a continuous reduction of tropical forest area due to cropland expansions, an increase in the average temperature (similarly to all regions in the world) and changes in the spatial and temporal precipitation variability at an uncertain magnitude (Thomson et al., 2010; Wohl et al., 2012; Nobre et al., 2016). As a result, the frequencies of droughts and floods in tropical regions related to land use and climate change are predicted to increase, resulting in threats to local or regional socio-economic development (IPCC, 2007; 2012). For Indonesia, it was projected that land use change and climate change may cause a food crisis in the future due to a decrease in crop production resulting from a warming climate (Amien et al., 1996; Naylor et al., 2007; Syaukat, 2011). Furthermore, poverty in Indonesia could increase in the future because food insecurity usually correlates with poverty (Slater et al., 2007; Syaukat, 2011). For these reasons, the Government of the Republic of Indonesia (Gol) has initiated numerous policies that engage all sectors to mitigate future risks associated with land use change and climate change (MoE, 2010). In the water resources sector, Gol has enacted National Law No. 7/2004 on water resources which aims at the integration between land use planning and water management (Fulazzaky, 2014; Putra & Han, 2014).

Integration between land use planning and water management has been acknowledged worldwide as a measure to achieve sustainable water resources use (Wheater & Evans, 2009). Land use planning can be an effective way to mitigate future risks associated with changes in land use and climate because different land use distributions may result in different water use and water storage characteristics (Bruijnzeel, 1989, 2004; Legesse et al., 2003; Memarian et al., 2014). However, it is a challenge to effectively implement the integration

between land use planning and water management. Population growth followed by an increasing demand for land for food and settlement increases the pressure on land, since more land is likely allocated for economic purposes (e.g. agricultural and industrial area expansions) than for soil and water protection purposes (Carter et al., 2005; Wheater & Evans, 2009). In addition, land use planning is often applied without information on its effectiveness to reduce risks for future water stress (Carter et al., 2005; Fulazzaky, 2014). For these reasons, it is necessary to gain knowledge on the extent and directions of future hydrological processes as affected by ongoing trends of land use change under various climate conditions.

Summarizing, the effects of land use change and climate change on water availability in tropical regions are not well understood yet and call for further investigations. This study intends to provide a better insight in the impacts of land use change and climate change on water availability in tropical regions, in particular for the chosen study catchment in Java, Indonesia. Even though this research was conducted for a single catchment, it is thought to represent problems that are characteristic for tropical developing countries, particularly in South-East Asia, where quite similar climate and land use conditions can be found.

1.2. Research objective

The main objective of this thesis is to assess the effects of land use change and climate change on the water availability in the Samin catchment (278 km²) in Java, Indonesia. The Samin catchment is selected as the study catchment because of data availability and the fact that it is representative for many catchments in Java, but also for other catchments of Indonesia and South-East Asia (in terms of experiencing massive land use changes during the last 30 years).

1.3. Research questions

In order to achieve the research objective, the following research questions are identified.

- 1. What is the attribution of changes in streamflow to land use change and climate change?
- 2. What is the attribution of changes in water balance components such as evapotranspiration, streamflow, surface runoff, lateral flow and base flow to land use change?

- 3. What is the future hydrological response of the study catchment under different land use change and climate change scenarios?
- 4. What are the effects of different spatial land use configurations on streamflow characteristics?

1.4. Research approach

This study started with an extensive literature study that resulted in an overview of methods to attribute changes in water availability to land use change and climate change. In general, two groups of methods were identified namely databased approaches and modelling approaches. While modelling approaches are able to simulate hydrological processes under changing conditions (i.e. land use change and climate change), they are accompanied by the need for large amounts of data and time-consuming model calibration and validation. Databased approaches require less data and are relatively faster than modelling approaches, but suffer from lacking the ability to describe underlying hydrological processes under land use and climate changes. However, both approaches may provide mutual benefits to assess the attribution of changes in water availability to land use change and climate change. In this study, both approaches were applied for the same study catchment (**Q1** and **Q2**) in order to give insight on the effects of land use and climate changes on water availability assessed using different approaches.

The data-based approach only uses observed time series of precipitation, streamflow and calculated potential evapotranspiration. Furthermore, a quantitative measure is developed to use those datasets to attribute changes in streamflow to land use change and climate change based on the changes in the proportion of excess water relative to changes in the proportion of excess energy (Q1). The results of the data-based approach will provide general knowledge on the relationship between land use change, climate change and streamflow alteration. To obtain a better understanding of the effects of land use change on the other water balance components such as actual evapotranspiration and the fraction of precipitation becoming surface runoff, lateral flow and base flow, a modelling approach is required (Q2). In this study, the Soil and Water Assessment Tool (SWAT) (Arnold et al., 1998) was used to simulate hydrological processes. The SWAT model was calibrated and validated using observed streamflow data and simulated with and without land use change conditions to investigate whether land use change is the main driver for changes in streamflow.

With a good knowledge on the relations between past land use change, climate change and their impacts on hydrological processes in the study catchment,

future water availability of the study catchment can be predicted (Q3). The validated SWAT model was run using as inputs simulated future land use distributions following a business-as-usual scenario and a land use planning scenario, and the outputs of climate models for two climate change scenarios namely Representative Concentration Pathway (RCP) 4.5 and 8.5. As a result, the individual and combined impacts of land use change and climate change on water availability including the effectiveness of applied land use planning can be assessed.

While previous research activities focused primarily on assessing the impacts of changes in land use type on water availability, it is interesting to know how changes in the spatial land use configurations may affect streamflow characteristics as well (Q4). For this purpose, the validated model was run under different land use scenarios with different spatial pattern characteristics, while the percentages of different land use types were fixed. In this scenario analysis, the land use planning was used as a baseline scenario so that the results can be used to explore potential hydrological impacts of alternative land use planning.

1.5. Thesis outline

Chapter 2 quantifies the relative contribution of land use change and climate change to streamflow alteration in the study catchment based on the relations precipitation, actual evapotranspiration between and potential evapotranspiration. Hydro-climatic data covering the period 1990-2013 and land use data acquired from Landsat satellite imageries for the years 1994 and 2013 were used and analyzed using a method based on the changes in the proportion of excess water relative to changes in the proportion of excess energy. Furthermore, land use change analysis and statistical trend analysis (i.e. the Mann-Kendall trend analysis and Sen's slope estimator) for the mean annual precipitation, potential evapotranspiration and streamflow were carried out to corroborate the attribution results.

Chapter 3 analyses the attribution of changes in the water balance of the study catchment to land use change using the Soil Water Assessment Tool (SWAT) and the baseline-altered method. The simulation period 1990–2013 was divided into four equal periods to represent baseline conditions and altered land use conditions. The past land use evolution from 1994-2013 was acquired from Landsat and Aster satellite imageries and used as inputs in the validated SWAT model with and without land use change to investigate the contribution of land use change to changes in streamflow. Furthermore, causal relationships between land use change and water balance components were investigated.

Chapter 4 analyses potential changes in hydrological processes to expected future changes in land use and climate. Hydrological processes in the future period 2030-2050 were simulated under both land use change and climate change scenarios. Two land use change scenarios, namely a business-as-usual (BAU) scenario, where no measures are taken to control land use change, and a controlled (CON) scenario, where the future land use follows the governmental land use planning, were used in the simulations together with two climate change scenarios, namely Representative Concentration Pathways (RCPs) 4.5 and 8.5. For climate change, the output of a bias-corrected Regional Climate Model and the outputs of six Global Climate Models were used to include climate model uncertainty.

Chapter 5 analyses the impacts of changes in land use pattern characteristics on streamflow characteristics of the study catchment. Changes in fifteen landscape metrics from past land use distributions in the period 1982-2013 were investigated to determine which land use pattern characteristics significantly affected the ratio of surface runoff to streamflow and the ratio of streamflow in the dry season to streamflow in the wet season. Furthermore, a validated SWAT model was applied using inputs from different land use pattern scenarios based on the land use planning to explore potential hydrological impacts of different land use patterns.

Chapter 6 concludes the thesis by summarizing the main findings, providing research limitations and future research directions, and explaining how this thesis contributes to hydrological science.

Chapter 2

Attribution of changes in streamflow to land use change and climate change in a mesoscale tropical catchment in Java, Indonesia¹

Abstract

Changes in the streamflow of the Samin catchment (277.9 km²) in Java, Indonesia, have been attributed to land use change and climate change. Hydroclimatic data covering the period 1990-2013 and land use data acquired from Landsat satellite imageries for the years 1994 and 2013 were analysed. A quantitative measure is developed to attribute streamflow changes to land use and climate changes based on the changes in the proportion of excess water relative to changes in the proportion of excess energy. The results show that 72% of the increase in streamflow might be attributed to land use change. The results are corroborated by a land use change analysis and two statistical trend analyses namely the Mann-Kendall trend analysis and Sen's slope estimator for mean annual discharge, precipitation and potential evapotranspiration. The results of the statistical trend analysis are in the same direction as the results of the attribution analysis, where climate change was relatively minor compared to significant land uses change due to deforestation during the period 1994-2013. This study concludes that changes in streamflow can be mainly attributed to land use change rather than climate change for the study catchment.

¹ This chapter is based on a paper that has been published as: Marhaento, H., Booij, M. J., & Hoekstra, A. Y. (2017). Attribution of changes in streamflow to land use change and climate change in a mesoscale tropical catchment in Java, Indonesia. *Hydrology Research*, *48*(4), 1143-1155

2.1. Introduction

Hydrology in tropical regions differs from that in other regions in having greater energy inputs and faster rates of change, including human-induced changes (Wohl et al., 2012). Despite high annual precipitation, water availability is often insufficient for human use in tropical regions because of seasonality, droughts, and increasing water demands resulting from rapid population growth. Bruijnzeel (1990) and Douglas (1999) argue that high rates of deforestation, urbanization and intensive land tillage, which are commonly found in tropical regions, have large impacts on water availability.

Bosch and Hewlett (1982) and Brown et al. (2005) reviewed the results of numerous catchment model experiments (e.g. paired catchment studies) throughout the world, including in the tropics, and found that changes in land use type through deforestation and afforestation can significantly affect the mean annual flow and the variability of annual flow (flow duration and seasonal flow). The annual water yield in tropical regions probably increases with deforestation, with maximum gains in water yield following total clearing (Bruijnzeel, 1990). However, these clear signals of how land use change affects hydrology were mostly found for small catchments. Evidence of land use change effects on water availability in larger catchments (>100 km²) in tropical regions is less consistent (Costa et al., 2003; Beck et al., 2013).

Apart from land use changes, climate change is the other main driver that influences water availability in tropical regions. Several studies have argued that climate change (particularly changes in temperature and precipitation) has a larger influence on water availability than land use change (Legesse et al., 2003; Khoi & Suetsugi, 2014; Yan et al., 2015). Blöschl et al. (2007) argue that climate change impacts on water availability vary depending on the spatial scale, due to direct and indirect influences through feedback mechanisms between land use and climate changes. Hejazi & Moglen (2008) found that the combination of land use change and climate change might result in more significant hydrological changes than either driver acting alone.

A major challenge in the study of tropical hydrology is to assess the attribution of changes in water availability to land use and climate changes (Romanowicz & Booij, 2011). A widespread belief exists among hydrologists in tropical countries that land use changes (e.g. deforestation) are the main cause of an increasing number of floods (Andréassian, 2004). Only a quantitative approach that combines the effects of land use and climate change can provide a better understanding of the single effect of land use change. Knowledge on the relative impacts of changes in land use and climate on water availability will be helpful in

estimating the effectiveness of land use management practices at the landscape level.

According to Zhang et al. (2012), there are two ways to distinguish the impacts of land use and climate changes on hydrology: a modelling and a non-modelling approach. The modelling approach has been widely used to measure the relative effects of land use and climate change on hydrology (Li et al., 2009; Khoi & Suetsugi, 2014; Zhan et al., 2013). However, the ability to simulate realistic conditions is accompanied by the need for large amounts of data. Several non-modelling approaches were introduced to assess the contribution of land use and climate changes on hydrology. Wei and Zhang (2010) and Zhang et al. (2012) used the modified double mass curve to exclude the effect of climate change on runoff generation in a deforested area. Tomer and Schilling (2009), Ye et al. (2013) and Renner et al. (2014) used a coupled water-energy budget approach to distinguish relative impacts of land use and climate change on watershed hydrology. A classical non-modelling approach is to employ trend analysis and change detection methods (Rientjes et al., 2011; Zhang et al., 2014).

This study aims to attribute changes in streamflow to land use and climate changes in the Samin catchment in Java, Indonesia. A non-modelling approach is used to achieve the research objective. An adaptation of the Tomer & Schilling (2009) approach is proposed to distinguish the impacts of land use and climate change on streamflow based on the relations between precipitation, actual evapotranspiration and potential evapotranspiration. Subsequently, statistical trend analysis (i.e. the Mann-Kendall trend analysis and Sen's slope estimator) and land use change analysis were carried out to validate the attribution results. The measures used for attribution analysis and the validation of the attribution results by means of statistical analyses and land use change analysis are the novelty of the present study. The study area and data availability are then described, followed by an explanation of the methods used in the study. Subsequent sections then discuss the key findings of the study and finally, conclusions are drawn.

2.2. Study area and data availability

2.2.1. Catchment description

The Samin River is one of the tributaries of the Bengawan Solo River, which plays an important role to support life within its surrounding area. It is located in the western part of Central Java Province, Indonesia. The catchment area of the Samin River extends over 277.9 km² and is located between latitude 7.6^{0} – 7.7^{0}

South and longitude 110.8⁰-111.2⁰ East (see Figure 2.1). The highest part of the catchment is located in the Lawu Mountain with an altitude of 3,175 meter above mean sea level and the lowest part is located near the Bengawan Solo river with an altitude of 84 meter above mean sea level. The average slope in the Samin catchment is 10.2%, and the stream density is around 2.2 km/km². According to the global soil map from the Harmonized World Soil Database (FAO/IIASA/ISRIC/ISSCAS/JRC, 2012), two soil classes namely Luvisols and Andosols are dominant in the Samin catchment, which occupy 57% and 43% of the study area, respectively. Luvisols are developed from parent material of accumulated silicate clay and Andosols are developed from parent material of the volcanic Lawu Mountain. Seasons in the Samin catchment are influenced by monsoon winds, where the dry season is influenced by Australian continental wind masses and generally extends from May to October and the wet season is influenced by Asian and Pacific Ocean wind masses and generally extends from November to April.



Figure 2.1. Location of the Samin catchment in Java, Indonesia.

2.2.2. Discharge data

The Bengawan Solo River Basin office provided daily water level data of the Samin catchment for the period 1990–2013. The daily discharge data have been

obtained by converting daily water level data to discharge values using the rating curves provided by the Bengawan Solo River Basin office. To test the reliability of the dataset, a quality check has been performed. A data screening process and a visual check of the hydrograph were carried out to identify missing and unrealistic values (outliers). An absolute error was found in the measured water level data where all daily water level data were systematically overestimated in the periods 1995-2008 and 2009-2013 (see Figure 2.2). The data provider confirmed that this error is probably due to a change of the gauge location.



Figure 2.2. Original daily water level data acquired from the data provider. The arrow shows a systematic error (shifting upward) in the water level data. Data for the entire year 2007 are missing.

A correction of the water level was carried out based on the height difference between the lowest water level of both error periods. The annual minimum 7-day average was used to define the lowest value in both periods. Correction values of -0.6 meter and -0.4 meter were found for the daily water level data within the period 1995–2008 and 2009–2013, respectively. Subsequently, the missing discharge data were completed using a non-linear recession model (Wittenberg, 1994). A non-linear recession model was selected after Pearson's test showed low correlation coefficients between the Samin discharge station and adjacent discharge stations (i.e. Keduang and Pidekso stations), which inhibited the use of widely used regional regression models to estimate the discharge. Note that this method was applied to fill-in data of maximum fifteen consecutive days of missing discharge values. Streamflow data that were unavailable for more than fifteen consecutive days were excluded from the analysis. This fill-in procedure concerns less than 5% of the data. The discharge data included missing daily discharge values for the entire year 2007.

2.2.3. Precipitation and Climatological data

Daily precipitation from eleven precipitation stations and meteorological data from three meteorological stations (Adi Sumarmo station, Pabelan station and Jatisrono station) were provided by the Bengawan Solo River Basin Organization for the period 1990–2013. Outliers and missing values of precipitation and meteorological data were identified and corrected. The data were checked for errors related to data processing (e.g. human errors) since most of the precipitation and meteorological data were manually recorded from the gauges. Doubtful precipitation values, such as negative precipitation values, unrealistic values and missing data were corrected using the normal ratio method (Paulhus & Kohler, 1952).

To obtain catchment average precipitation depths, daily precipitation values were averaged using the Thiessen polygon approach with elevation correction (TEC). The TEC method was selected after the results from the TEC approach were compared with three other widely known interpolation methods namely Inverse Distance Weighting (IDW), Ordinary Kriging (OK) and Ordinary Co-kriging (OCK) using 72 randomly selected sample points of mean monthly precipitation. It was found that the Root Mean Square Error (RMSE) of TEC of 67 mm was comparable with the RMSE of IDW (56 mm), OK (69 mm) and OCK (60 mm) and for all methods the $R^2 > 0.8$. Moreover, the TEC method is the simplest method to compute average precipitation values. The elevation correction for the TEC approach is based on a simple linear regression between the mean annual precipitation and elevation of thirteen precipitation stations in the surrounding catchment. This resulted in a correction factor for the Thiessen polygon method of 153 mm increase of annual precipitation per 100 m increase of elevation.

The reference evapotranspiration (ET_0) was calculated in each meteorological stations using the Penman-Monteith method as recommended by the Food and Agricultural Organization (Smith and Allen, 1992). However, the daily meteorological data for Pabelan station and Jatisrono station were only available from 2008 to 2013. To complete the meteorological values in these stations, daily meteorological data from the National Centers for Environmental Prediction Climate Forecast System Reanalysis (Saha et al., 2010) were used. They provide daily climate data at a resolution of $0.25^0 \times 0.25^0$ from 1979 to 2010. Furthermore, daily ET₀ for the study catchment were averaged using the Thiessen polygon approach. An elevation correction for ET₀ was not used since the data availability was not sufficient to determine the correlation between potential evapotranspiration and elevation. However, the elevation gap between the mean elevation of the catchment and the meteorological stations is minor. Figure 2.3 shows the mean annual precipitation, potential evapotranspiration and discharge of the study catchment.



Figure 2.3. Mean annual precipitation, potential evapotranspiration and discharge for the period 1990–2013 in the Samin catchment. The data include a missing discharge value for the year 2007.

2.2.4. Spatial data

Landsat imageries from 1994 and 2013 were available for the study area through the United States Geological Survey archives (USGS, 2016). The data scene (path/row) number is 119/65 and the acquisition dates are September 1, 1994 and October 7, 2013. Both images have cloud cover of less than 5% so are sufficient for further analysis of land use images classification. The catchment boundaries and the stream network of the study area were delineated based on a Digital Elevation Model (DEM) from a contour map with a Contour Interval (CI) of 12.5 meters that was available from the Geospatial Information Agency of Indonesia.

2.3. Methods

2.3.1. Separating effects of land use and climate change on streamflow

This study extends the idea of Tomer and Schilling (2009) who distinguish the impacts of land use and climate change on hydrology using the changes in the proportion of excess water relative to changes in the proportion of excess energy. The amount of excess water within the system (i.e. catchment) can be expressed as precipitation (P) minus actual evapotranspiration (ET) and the amount of excess energy as potential evapotranspiration (ET₀) minus actual evapotranspiration (ET). The amounts of excess water and excess energy
divided by the available water and energy amounts result in dimensionless values P_{ex} and E_{ex} on a scale of 0 to 1, which can be expressed as follows:

$$E_{ex} = 1 - ET/ET_0$$
 (eq. 2.2)

where P_{ex} is the proportion of excess water, E_{ex} the proportion of excess energy, P the precipitation, ET₀ the potential evapotranspiration and ET the actual evapotranspiration.

The Tomer and Schilling (2009) framework follows two basic assumptions for separating land use and climate change impacts on hydrology based on excess water and energy. First, land use changes will affect ET, which will decrease or increase P_{ex} and E_{ex} simultaneously because ET is in the numerator of both fractions. As a result, P_{ex} and E_{ex} will move creating an angle close to 45[°] or 225[°] compared to the x-axis (see Figure 2.4) if there is a change in land use while climate is unchanged (i.e. $\Delta P \sim 0$ and $\Delta ET_0 \sim 0$). A movement creating an angle of 45[°] indicates a decrease of water and energy consumption (e.g. less ET because of a less vegetated area), while a movement creating an angle of 225⁰ indicates an increase of water and energy consumption (e.g. more ET because of a more densely vegetated area). Second, climate change will affect P and/or ET₀, which will be reflected by a change in the ratio of P to ET₀. If the ratio of P to ET₀ increases while ET remains unchanged (i.e. no land use changes), the P_{ex} value will increase and/or the Eex value will decrease, and vice versa, creating a movement along a line with an angle close to 135° or 315° compared to the xaxis (see Figure 2.4). Within the framework, a change in streamflow can be equally attributed to land use change and climate change if movements of Pex and E_{ex} are parallel to the P_{ex} axis or E_{ex} axis. A reference is made to Tomer and Schilling (2009) for a more detailed explanation about the concept.

However, Renner et al. (2014) argue that the Tomer and Schilling (2009) concept cannot be applied to all hydro-climatic conditions and works only for a region where precipitation equals evaporative demand. They proposed an adaptation of the concept by considering the aridity index (ET_0/P) to determine the climatic state of the study catchment. Within their improved concept, a land use change impact on hydrology is defined as a change in ET, but with a constant aridity, and a climate change impact on hydrology is defined as a change of P_{ex} and E_{ex} for the same aridity index is considered as a land use change impact and a change of P_{ex} and E_{ex} moving away from a constant aridity index is considered as a climate change impact.

This study extended the framework adapted by Renner et al. (2014) by developing quantitative measures to estimate the land use and climate change impacts on streamflow alteration based on the changes of P_{ex} and E_{ex} . The period of analysis 1990-2013 for which hydro-meteorological time series were available was divided into two periods: a baseline and an altered period, when land use change and climate change might have contributed to streamflow change. The years 1990-1997 and 2006-2013 were regarded as the baseline period and the altered period, respectively, since during the period 1998-2003 significant land use changes have occurred due to deforestation. In 1998, which is considered to be the starting year of the 'reformation era', many local communities reclaimed their customary rights inside state forests and converted forest area to other land uses as alternative sources of livelihood after the economic crisis in Indonesia (Resosudarmo et al., 2012). Subsequently, Pex and E_{ex} for the baseline period and altered period, later symbolized as point M₁ (P_{ex1}, E_{ex1}) and M₂ (P_{ex2}, E_{ex2}) , were compared and the change of P_{ex} and E_{ex} relative to the long-term aridity index ($\overline{ET_0}/\overline{P}$) of the study catchment (Figure 2.4) were determined.

The magnitude of land use and climate change impacts that causes a shift from point M_1 (P_{ex1}, E_{ex1}) to M_2 (P_{ex2}, E_{ex2}) is estimated based on three measures: (1) the resultant length (R); (2) the angle (θ) of change; and (3) the attribution (in %) to land use change and climate change. The resultant length (R) indicates the magnitude of the changes of excess water and energy where a higher resultant length (R) represents a higher magnitude of changes of excess water and energy. A higher change of excess water and energy then corresponds to a higher rate of land use and climate change impacts on streamflow change. The magnitude of the resultant length (R) from M_1 to M_2 can be calculated based on Phytagoras' theorem as follows:

$$R = \sqrt{(E_{ex2} - E_{ex1})^2 + (P_{ex2} - P_{ex1})^2}$$
 (eq. 2.3)

The angle (θ) of change indicates the contribution of land use and climate changes with a higher slope reflecting a higher contribution of climate change. The angle (θ) can be calculated based on the gradient of the vector M₁-M₂ relative to the gradient of the long term aridity index using the following equations:

$$\tan(\vartheta) = \frac{\left|\frac{\overline{\mathrm{ET}_{0}}}{\overline{\mathrm{P}}} - \frac{\mathrm{P}_{\mathrm{ex2}} - \mathrm{P}_{\mathrm{ex1}}}{\mathrm{E}_{\mathrm{ex2}} - \mathrm{E}_{\mathrm{ex1}}}\right|}{1 + \left(\frac{\overline{\mathrm{ET}_{0}}}{\overline{\mathrm{P}}}\right) * \left(\frac{\mathrm{P}_{\mathrm{ex2}} - \mathrm{P}_{\mathrm{ex1}}}{\mathrm{E}_{\mathrm{ex2}} - \mathrm{E}_{\mathrm{ex1}}}\right)}$$
(eq. 2.4)

$$\theta = \arctan(\theta)$$
 (eq. 2.5)

The attribution (in %) of streamflow changes to land use change and climate change is measured by determining the length of the changes along the aridity index line and the line perpendicular to the aridity index line, which are denoted as LUC and CC respectively. The lengths of LUC and CC can be calculated as follows:

$$LUC = \mathsf{R}^* \cos \theta \tag{eq. 2.6}$$

$$CC = R^* \sin \theta$$
 (eq. 2.7)

The relative magnitudes of *LUC* and *CC* are denoted as L (%) and C (%) and calculated using the following equations.

$$L(\%) = \frac{LUC}{LUC + CC} *100\%$$
 (eq. 2.8)

$$C(\%) = \frac{CC}{LUC + CC} *100\%$$
(eq. 2.9)



Figure 2.4. Adapted Tomer and Schilling (2009) framework to illustrate how the fractions of excess water and energy respond to land use changes and climate change. The points M_1 and M_2 are the fractions of excess water and energy of the baseline period (P_{ex1}, E_{ex1}) and altered period (P_{ex2}, E_{ex2}). The contribution of land use and climate change to streamflow changes is estimated based on the changes of P_{ex} and E_{ex} relative to the long-term aridity index line ($\overline{ET_0}/\overline{P}$). For example, if the long-term aridity index is 0.8, the change along the constant aridity index line (i.e. LUC line) is attributed to land use change and the line perpendicular to this line (i.e. CC line) is attributed to climate change. The movement direction will determine whether land use change or climate change has a more dominant contribution to changes in streamflow.

2.3.2. Validation of attribution assessment

Two analyses were carried out to validate the results of the attribution analysis: a statistical trend analysis to validate the contribution of climate change to streamflow change and a land use change analysis to validate the contribution of land use change to streamflow change.

2.3.2.1. Trend analysis of climate variables

Trend analysis was performed to check whether the mean annual discharge (Q), precipitation (P) and evapotranspiration (ET_0) have significantly changed over time (long-term). It was hypothesized that if climate change has a larger contribution than land use change to streamflow alteration, the trends in climate variables (P and ET_0) will be in the same direction and have the same magnitude as the streamflow trend. The trend direction and magnitude were determined using the Mann-Kendall test and Sen's slope estimator. The Mann-Kendall test and Sen's slope estimator were selected since they are widely used to detect trends in long-time series of hydrological and climatological data (Rientjes et al., 2011; Zhang et al., 2014).

2.3.2.2. Land use change analysis

Land use change analysis was carried out to measure the rate of land use change in the study catchment, and to validate the contribution of land use change to streamflow changes. It was hypothesized that if land use change has a larger contribution than climate change to streamflow alteration, the type of change in land use will be in line with the attribution results, e.g. deforestation will affect to increase P_{ex} and E_{ex} simultaneously.

Landsat imageries from the years 1994 and 2013 were used to assess land use changes within the study area. These two imageries represent the land use condition of the baseline period (1990-1997) and altered period (2006-2013). Before image processing, a pre-processing analysis had been applied for the selected images including geometric correction to avoid distortion on map coordinates and masking analysis to obscure the area beyond our study area. After the pre-processing analysis was completed, a maximum likelihood algorithm was applied to retrieve the land cover map using a thousand sample points that were generated from an institutional land use map (scale 1:25,000) from the Geospatial Information Agency of Indonesia. The sample points were divided into two parts: half of the sample points were used to perform image classification and another half was used to perform accuracy assessment. An error matrix (Congalton, 1991) was made to calculate the accuracy using four measures: the producer's accuracy, the user's accuracy, the overall accuracy and the Kappa coefficient. The producer's accuracy is to measure how well a certain area can be classified. The user's accuracy is to measure how well labels on a map represent each category on the ground. The overall accuracy is to measure the total number of correct samples divided by the total number of samples. The Kappa coefficient is the coefficient of agreement between the classification map and the reference data. Subsequently, land use change

analysis was performed based on the area differences of each land use class from different years.

2.4. Results

2.4.1. Attribution of changes in streamflow to land use change and climate change

The results for the three measures (see Table 2.1 and Figure 2.5) show a simultaneous increase of P_{ex} and E_{ex} in the study catchment. The increase of P_{ex} and E_{ex} occurred because ET has significantly decreased, which is probably due to deforestation, while P and ET₀ remain relatively unchanged. The aridity index was found to be 0.8 and the movement of P_{ex} and E_{ex} relative to the aridity index line has created an angle of 21°. The angle is less than 45° indicating that climate change (P and ET₀) is minor and has a smaller contribution than land use change on the streamflow alteration. In addition, the change of Pex and Eex is relatively low with a Resultant value (R) of 0.1. The attribution of changes in streamflow to land use change and climate change was estimated to be about 72% and 28%, respectively. Note that the discharge data includes uncertainty during measurements that might influence the attribution results considerably. Using the original discharge data (i.e. before the mean annual discharge has been corrected by a decrease of 60% for the years 1995–2008 and a decrease of 40% for the years 2009–2013 due to a systematic error), the attribution results were found to be 98% and 2% for land use change and climate change contribution, respectively.

Table 2.1. Measures of the attribution of changes in streamflow to land use and climate changes

Period	Р	ET_0	Q	ET	P _{ex}	E _{ex}	R	θ	L	С
1990 - 1997	1,962	1,644	588	1,374	0.30	0.16				
2006 - 2013	2,072	1,639	771	1,301	0.37	0.20	0.1	21.0	72	28

P = mean annual precipitation (mm), ET_0 = mean annual potential evapotranspiration (mm), Q = mean annual discharge (mm), ET = mean annual evapotranspiration (mm), P_{ex} = excess water divided by available water, E_{ex} = excess energy divided by available energy, R = Resultant length (dimensionless), ϑ = angle of changes (degrees), L = Attribution to land use change (%), C = Attribution to climate change (%)



Figure 2.5. Change of excess water (P_{ex}) and excess energy (E_{ex}) relative to long term aridity index line ($\overline{ETo}/\overline{P}$). The arrow shows the change of P_{ex} and E_{ex} between the baseline period (1990–1997) and the altered period (2006–2013). The natural variations of P_{ex} and E_{ex} for each period are represented by the standard deviation lines.

2.4.2. Trend analysis of climate variables

The trend analysis was carried out for mean annual climate variables (i.e. P and ET_0) and discharge (Q). The results of the Mann-Kendall test and Sen's slope estimator (see Table 2.2) show that the trends in P and ET_0 are not significant while the trend in Q is significant at a significance level of 5%. The trend magnitude determined by Sen's slope for Q (i.e. Sen's slope = 12.1 mm/year) is larger than for ET_0 (i.e. Sen's slope = 1.3 mm/year) and P (i.e. Sen's slope = 2.0 mm/year). In general, the statistical results from the Mann-Kendall trend test and Sen's slope estimator showed that the mean annual precipitation and potential evapotranspiration have not significantly changed while the mean annual discharge has changed significantly. The results are in line with the attribution results, which generally revealed a small contribution of climate change to changes in streamflow.

Analysis	Precipitation	ET ₀	Discharge
Mann-Kendall Z-statistic	0.1	0.4	2.7*
p-value	1.0	0.4	0.01*
Sen's slope	2.0	1.3	12.1

Table 2.2. Results of statistical trend analysis for mean annual precipitation, mean annual potential evapotranspiration and mean annual discharge for the period 1990-2013

An asterisk indicates that trend is significant at 5%

2.4.3. Land use change detection

Following the land use classification from the Geospatial Information Agency of Indonesia, eight dominant land use classes were found in the study area: evergreen forest, mixed garden, settlement, paddy field, dryland farming, shrubs, bare land and water body. Evergreen forest is homogeneous forest area that consists of Pinus merkusii tree species. Mixed garden is community forest that consists of multipurpose trees (e.g. fruits, fuel woods, etc.) and often combined with seasonal crops on the same unit of land. Settlement is build-up area and its surroundings. Paddy field is agricultural area that consists of paddy rice fields with an intensive irrigation system. Dry land farming is agricultural area for seasonal crops production. Shrub is abandoned area covered by herbaceous plants. Bare land is rocky abandoned area without vegetation cover. Water body refers to rivers and ponds. By applying an error matrix (Congalton, 1991) using 500 (unit) samples, it was found that the average producer's accuracy is 87.6%, the average user's accuracy is 91.5%, the overall accuracy is 89.3% and the Kappa coefficient is 87.6%. According to Anderson et al. (1976), these accuracy assessment results may represent a strong agreement and high accuracy for producing a land use map.

Land use change analysis was performed based on the area differences for each land use class from different years. Furthermore, the eight land use classes were reclassified into four land use classes to have more general land use classes namely forest area (i.e. combination of evergreen forest and mixed garden), agricultural area (i.e. combination of paddy field and dry land farming), settlements and others (i.e. combination of shrub, bare land and water body) (see Table 2.3). The results show that settlements and agricultural area have increased by 24% and 6%, respectively during the period 1994–2013. These expansions caused large scale deforestation by decreasing the forest area with 32%. Since climate changes have a minor contribution to the streamflow alteration, significant changes in land use (i.e. deforestation) validate the results of the attribution analysis, which revealed a larger contribution of land use change than climate change to streamflow alteration. Figure 2.6 shows the land use maps for the years 1994 and 2013.

	1994		2013		
Land Use Class	(in hectare)	%	(in hectare)	%	Change (%)
Forest area	13,542.7	49	4,687.1	17	-32
Agriculture area	10,896.6	39	12,628.6	45	6
Settlements	2,711.6	10	9,531.5	34	24
Others	647.1	2	950.9	4	2
Total	27,798.0	100	27,798.0	100	0

 Table 2.3. Land use distribution of the Samin catchment in the year 1994 and 2013 including its change

 1994
 2013



Figure 2.6. Land use maps in the Samin catchment for 1994 (a) and 2013 (b)

2.5. Discussion

Land use changes, which are related to deforestation due to expansion of agriculture areas and settlement areas, were probably the cause of significant changes in streamflow generation. Using the three measures developed in this study, land use change has found to contribute to about 72% of the streamflow alteration in the study catchment. These results are in the same direction as the

results of the statistical trend analysis, where the annual ET_0 and P have not significantly changed (at a significance level of 5%) over the period of analysis. In contrast, the annual Q has significantly changed (at a significance level of 5%) and at the same time, land use has dramatically changed, where a large increase of settlements and agricultural area has decreased the forest area during the period 1990–2013. These findings corroborate the attribution results where changes in streamflow can be largely attributed to land use changes rather than to climate change in the Samin catchment. Numerous studies argue that a continuous decline of tree-areas in catchments may lower the infiltration rate, reduce the groundwater recharge and inhibit water to be stored in the soil (Bosch and Hewlett, 1982; Brown et al., 2005). As a result, a larger volume of precipitation was transformed into surface runoff.

Despite the fact that the impacts of land use change and climate change on streamflow alteration were evident, the magnitude of change in excess water and energy represented by the Resultant (R) length, was relatively low. Besides land use and climate change, the magnitude of change in streamflow seems to be affected by other factors for instance by the catchment size, the slope variation and the soil type. Several studies reported the impacts of land use change on streamflow generation for different catchment sizes (e.g. D'Almeida et al., 2007; Blöschl et al., 2007; Gallo et al., 2015). These studies generally argue that the magnitude of land use change impacts on hydrology became smaller with increasing catchment area. In addition to the catchment size, the slope variation may also influence the impact magnitude. Van Dijk et al. (2007) argue that a larger topographic variation results in shallower soils, less infiltration and therefore generating more runoff. Thus, the impact of land use change on streamflow generation in a catchment with a large topographic variation will be amplified and vice versa. Bruijnzeel (2004) addressed the role of soil conditions on the magnitude of land use change impacts on hydrology in tropical regions. He argues that soil protection measures following deforestation, for instance by applying the Reduce Impact Logging (RIL) technique during land clearing for agriculture or plantations, might decrease the impact magnitude of forest removal in hydrological processes. Nonetheless, underlying natural geology and soil types in a system are important to control catchment hydrological behaviour after land use has changed. A porous soil of volcanic deposits in the study area might have a lower impact magnitude than an area with similar land use change condition having a low porosity and low hydraulic conductivity. However, the influence of these factors (i.e. catchment size, slope variation and soil type) on the resultant value could not be assessed in the present study due to limited data availability in other catchments. More research is needed to test the applicability of the resultant value under different catchment conditions.

The present study proposes a framework to quantitatively assess the attribution of changes in streamflow to land use and climate changes. Although promising results were obtained, two challenges were addressed for further study.

First, the basic conceptual design proposed by Tomer and Schilling (2009) depends on rigorous assumptions, which are not realistic in the real world. The framework uses the assumption that climate change only results in changes in P and ET_0 and land use change only results in changes in ET. In this way, the basic concept neglects the natural complex system where changes in ET are caused by an interaction between climate change and land use change (Budyko, 1974; Wang, 2014; Jiang et al., 2015). Furthermore, the basic concept used the assumption of a linear correlation between the fractions of excess water and energy that is represented by a straight line in a 2-D plot. This simplification differs from the widely known Budyko curve (Budyko, 1974), but is in line with the study of Pike (1964). Renner et al. (2012) argue that the concept of Tomer and Schilling (2009) is not valid for wet catchments (i.e. P is much higher than ET_0) or dry catchments (i.e. P is much lower than ET₀) so that is not applicable in many parts of the world. The basic concept that was originally developed for a temperate climate only works for conditions where precipitation meets evaporative demand (i.e. the middle part of the Budyko curve). Using the aridity index as a correction for the basic concept (Renner et al., 2014), the results have improved but do not reduce the uncertainty inherent in the basic assumption. The proposed approach requires a condition in which changes in the water supply have the same impacts as a change in energy supply, but in opposite directions (i.e. $\Delta P = -\Delta ET_0$). Thus, for conditions where P and ET_0 changes in the same direction (i.e. both decrease or increase), the attribution of changes in streamflow to climate change will interfere with land use change impacts. The results of the present study were found convincing, because the hydro-climatic state of our study catchment met the conditions imposed. Although a sharp attribution is not possible due to the assumptions used, the movements of P_{ex} and E_{ex} compared to the aridity index line can provide a rough indication. A validation through trend analysis of climate variables (i.e. P and ET₀) and land use change analysis as applied in this study appears to be useful to verify the attribution results. The proposed method needs more practical applications across various climatic regions to make the approach more reliable and robust.

Second, this study agrees with Tomer and Schilling (2009), Ye et al. (2013) and Renner et al. (2014) that the basic concepts of excess water and energy are sensitive to the data quality, particularly for precipitation, potential evapotranspiration and streamflow data. Reliable time series of hydrological data are rarely found in developing countries, including Indonesia (Douglas, 1999). However, performing data checks for errors and making data corrections using well-established methodologies (as demonstrated in this study) can be useful to arrive at more reliable datasets. Moreover, data analysis in this study was carried out over a long time period and on an annual basis, which may reduce random errors. More convincing results, however, are expected if hydrological datasets are available for a long time period and data gauges are well distributed over the area of interest.

2.6. Conclusions

A quantitative assessment of land use and climate change contribution to streamflow alteration has been carried out using measures described in this paper. The results show that changes in streamflow of the Samin catchment during the period 1990-2013 can be attributed to land use change for 72% and climate change for 28%. The results were corroborated by the results of statistical trend analyses (Mann-Kendall trend analysis and Sen's slope estimator), and land use change analysis. The results of the statistical trend analyses show that the climate (i.e. mean annual P and ET₀) has not significantly changed while the mean annual discharge has significantly changed at a confidence level of 5%. At the same time, land use has significantly changed due to deforestation where the forest area has decreased by 32% mostly due to an increase of settlements and agricultural area of 24% and 6%, respectively. Our results are in line with the results from other tropical hydrological studies on the contribution of land use and climate change to streamflow alteration ranging from small-scale experiments (Bosch and Hewlett, 1982; Bruijnzeel, 1990; Brown et al., 2005) to large-scale modelling studies (Thanapakpawin et al., 2007; Alansi et al., 2009).

Chapter 3

Attribution of changes in the water balance of a tropical catchment to land use change using the SWAT model²

Abstract

Changes in the water balance of the Samin catchment (277.9 km²) on Java, Indonesia, can be attributed to land use change using the Soil Water Assessment Tool model. A baseline-altered method was used in which the simulation period 1990-2013 was divided into 4 equal periods to represent baseline conditions (1990-1995) and altered land use conditions (1996-2001, 2002-2007, and 2008-2013). Land use maps for 1994, 2000, 2006, and 2013 were acquired from satellite images. A Soil Water Assessment Tool model was calibrated for the baseline period and applied to the altered periods with and without land use change. Incorporating land use change resulted in a Nash-Sutcliffe efficiency of 0.7 compared to 0.6 when land use change is ignored. In addition, the model performance for simulations without land use change gradually decreased with time. Land use change appeared to be the important driver for changes in the water balance. The main land use changes during 1994–2013 are a decrease in forest area from 48.7% to 16.9%, an increase in agriculture area from 39.2% to 45.4%, and an increase in settlement area from 9.8% to 34.3%. For the catchment, this resulted in an increase of the ratio of streamflow to precipitation from 35.7% to 44.6% and a decrease in the ratio of evapotranspiration to precipitation from 60% to 54.8%. More pronounced changes can be observed for the ratio of surface runoff to streamflow (increase from 26.6% to 37.5%) and the ratio of base flow to streamflow (decrease from 40% to 31.1%), whereas changes in the ratio of lateral flow to streamflow were minor (decrease from 33.4% to 31.4%). At sub-catchment level, the effect of land use changes on the water balance varied in different sub-catchments depending on the scale of changes in forest and settlement area.

² This chapter is based on a paper that has been published as: Marhaento, H., Booij, M. J., Rientjes, T. H. M., & Hoekstra, A. Y. (2017). Attribution of changes in the water balance of a tropical catchment to land use change using the SWAT model. *Hydrological Processes*, 31(11), 2029-2040.

3.1. Introduction

Attributing changes in streamflow to land use and climate change has been of interest for decades. Numerous studies have been carried out throughout the world focusing on the relationship between changes in streamflow and changes in land use and climate. Zhang et al. (2012) distinguishes two approaches to attribute changes in streamflow to land use change: a modelling and non-modelling approach. The latter is often carried out based on paired catchment studies (Bosch and Hewlett, 1982; Brown et al., 2005) or long-term hydrological data analyses (Rientjes et al., 2011; Zhang et al., 2014; Marhaento et al., 2017b). The modelling approach can be based on a single model (Niehoff et al., 2002; Rodriguez Suarez et al., 2014) or multiple models (ensemble modeling) (Huisman et al., 2009). Surprisingly, very few studies investigated hydrological behavior under changing conditions in tropical regions (Douglas, 1999; Wohl et al., 2012). The impact of land use change and climate change on streamflow can be significant for tropical regions having greater energy inputs and faster rates of change, including human-induced changes (Wohl et al., 2012).

In the 1990s, Indonesia was among countries having the largest land use changes in terms of forest lost in the world, behind Brazil (Hansen et al., 2009). A high demand of land resources to accelerate the economic development and meet local needs for food and settlements resulted in land use changes at different scales. Several hydrological studies have been carried out in Indonesia to investigate the impacts of land use change on water resources (Valentin et al., 2008; Remondi et al., 2015), showing that deforestation and urbanization were the two land use changes mostly reported to affect streamflow. Deforestation and urbanization caused increases in peak flows enhancing the flood risk during the wet season and decreases in base flow in the dry season following the loss of forest area (Bruijnzeel, 2004).

Douglas (1999) argues that most of the hydrological studies in tropical regions, including Indonesia, focus on the impacts of land use change on streamflow (e.g. seasonal flow and annual flow) and sediment yield based on small-scale plot measurements. In addition, land use change can also affect other water balance components such as evapotranspiration, soil water content and groundwater recharge. Yet, only few studies have attempted to assess the attribution of changes in the water balance to land use change in tropical regions. Moreover, some of the results are contradictory and inconsistent, in particular for large catchments (>100 km²; Calder et al., 2001; Beck et al., 2013).

In order to assess the impacts of land use change on the water balance, a modelling approach is typically used. Models can be used to assess impacts of historic as well as future land use changes on hydrological conditions

(Thanapakpawin et al., 2007; Huisman et al., 2009). A wide range of models has been used to simulate land use change impacts on the water balance, e.g. SWAT (Arnold et al., 1998), MIKE-SHE (Refsgaard and Storm, 1995), or DHSVM (Wigmosta et al., 1994). However, the contribution of land use change to changes in the catchment water balance is still not fully understood due to climatic interferences (Wang, 2014), requiring further study. A challenge is to attribute changes in the water balance to land use change under varying climatic conditions (Romanowicz and Booij, 2011).

Against this background, the goal of this study is to attribute changes in the water balance of a tropical catchment in Indonesia to land use change using a modelling approach. First, changes in streamflow of the Samin catchment in Java, Indonesia are attributed to land use change for the time period 1990–2013, using the semi-distributed physically based SWAT model (Soil and Water Assessment Tool) (Arnold et al., 1998) and a baseline-altered method. Second, causal relationships between land use change and water balance alteration in the study catchment are assessed. A diagnostic approach is followed using two widely used statistical analyses, namely Mann-Kendall trend analysis and Pearson's test, to investigate whether the water balance has statistically changed over the study period due to land use change. This study considers the water balance in terms of five water balance ratios: the ratio of streamflow to precipitation (the runoff coefficient, Q/P), the ratio of evapotranspiration to precipitation (ET/P), the ratio of surface runoff to streamflow (Q_b/Q).

The use of attribution analysis to test whether land use change is the main driver for changes in streamflow prior to assess the impact of land use change on the overall water balance is proposed. In addition, a diagnostic approach may provide insight on the relationship between land use change and water balance alteration that is rarely investigated in tropical regions. This study complements the study of Marhaento et al. (2017b) who conducted attribution analysis of changes in streamflow to land use and climate change using a non-modelling approach for the same catchment.

The structure of the paper is as follows. Section 3.2 presents a description of the study area and the data availability. Section 3.3 describes the methods used and Sections 3.4 and 3.5 give the results and discussion of the key findings. Finally, conclusions are addressed in Section 3.6.

3.2. Study area and data

3.2.1. Catchment description

The Samin River, with a total length of about 53 km, is of the Bengawan Solo River, which plays an important role to support life within its surrounding area. The catchment area of the Samin River extends over 277.9 km² and ranges between southern latitude 7.6°-7.7° and eastern longitude 110.8°-111.2° (see Figure 3.1). The main economic activity in this catchment is agriculture, which is supported by the water availability within the catchment. The highest part of the catchment is the Lawu Mountain, with an altitude of 3,175 meter above mean sea level and the lowest part is near the Bengawan Solo river in Sukoharjo district, with an altitude of 84 meter above mean sea level. The average slope in the Samin catchment is 10.2% where higher slopes are mostly in the upstream part near the foothill of the Lawu Mountain. The stream density is around 2.2 km/km² and streams form a parallel drainage pattern, where the tributaries tend to stretch out from east to west following the surface slope. According to the soil map from the Harmonized World Soil Database (FAO/IIASA/ISRIC/ISSCAS/JRC, 2012), two soil classes, namely Luvisols and Andosols, are dominant in the Samin catchment, which cover 57% and 43% of the study area, respectively. Luvisols are developed from parent material of accumulated silicate clay that is dominant in the midstream to downstream area. These soils are characteristic for the forested area where a leafy humus horizon can be found at the top of the soil layers. Andosols are developed from parent material of the volcanic Lawu Mountain and are mainly dominant in the upstream area. These soils are typically highly porous and fertile due to volcanic deposits from the Lawu Mountain.



Figure 3.1. Samin catchment in Java, Indonesia, with the locations of hydrological gauges

3.2.2. Hydro-climatic conditions

The climate of the Samin catchment is tropical monsoon with distinct dry and wet seasons. The dry season is influenced by Australian continental masses and generally extends from May to October whereas the wet season is influenced by Asian and Pacific Ocean wind masses and generally extends from November to April. The spatial precipitation pattern likely follows the orography, with a larger amount of precipitation in the upstream area than in the downstream area. Based on a simple linear regression between the mean annual precipitation and elevation of thirteen precipitation stations in the surrounding catchment, it was found that the mean annual precipitation increases about 150 mm per 100 m increase of elevation. This is comparable with the finding of Subarna et al. (2014) who estimated an increase in the mean annual precipitation of about 140 mm per 100 m increase of elevation for the Cisangkuy catchment in West Java, Indonesia. The mean daily temperature is approximately 26 ⁰C with a mean daily minimum of about 21.5 °C and a mean daily maximum of about 30.5 °C. The daylight length is approximately 13 hours with a small difference between the wet and dry seasons resulting in a mean daily solar radiation of about 17 MJ/m². 1990–2013, the During the period annual precipitation, potential evapotranspiration and streamflow of the Samin catchment ranged from 1,500 mm - 3,000 mm, 1,400 mm - 1,700 mm and 500 mm - 1,200 mm, respectively where a high Q/P (i.e. >0.5) was found for the years 2000 and 2005 (see Figure 3.2). The trend analysis of the mean annual precipitation, potential evapotranspiration and streamflow showed an increase of approximately 48 mm, 31.2 mm and 290 mm, respectively (Marhaento et al., 2017b). Whereas longterm changes in the mean annual precipitation and potential evapotranspiration were statistically not significant, changes in the mean annual streamflow were significant at a significance level of 5%. Figure 3.2 shows the mean annual precipitation, potential evapotranspiration, and streamflow of the Samin catchment for the period 1990-2013.

3.2.3. Data collection

The SWAT model requires a Digital Elevation Model (DEM), land use data, soil data and climate data as model inputs. The DEM data were generated from a contour map with a contour interval of 12.5 meters, which was made available by the Indonesia Geospatial Information Agency. Land use information was available for a 30 m resolution for the years 1994 and 2013 from Marhaento et al. (2017b).



Figure 3.2. Mean annual precipitation, potential evapotranspiration, and streamflow for the period 1990–2013 in the Samin catchment (after Marhaento et al, 2017b)

Two additional land cover images were acquired from the Landsat satellite (USGS, 2016) for September 9, 2000 and Aster images (ERSDAC, 2016) for November 9, 2006 to represent land use for the years 2000 and 2006. Soil data were taken from the Harmonized World Soil Database (FAO et al., 2012). The soil characteristics required as SWAT input that were not available in the global soil map (i.e. available water content, saturated hydraulic conductivity and bulk density), were obtained from the Soil-Plant-Atmosphere-Water (SPAW) model (Saxton & Willey, 2005).

The SPAW model uses pedotransfer functions including information on soil texture, soil salinity, organic matter, gravel and soil compaction to determine water retention characteristics. The model has been developed based on an extensive number of laboratory datasets of soil characteristics from the USDA/NRCS National Soil Characterization database and has been validated using thousands of sample subsets (Saxton & Willey, 2005). Even though the SPAW model has been developed in the United States, the application of the SPAW model is worldwide and often performed with a model calibration (Botula et al., 2014).

For model calibration, daily streamflow data were used. The Bengawan Solo River Basin office provided daily water level data in the Samin catchment for the period 1990–2013. The daily discharge data have been obtained by converting the daily water level data to discharge values using the rating curves provided by

the Bengawan Solo River Basin office. It was assumed that the available rating curve is sufficient to represent the river flows (i.e. high flows and low flows) considering the limited information on how the rating curve has been checked and updated by the data provider. This study used the same discharge and climate data as Marhaento et al. (2017b) who have corrected erroneous water level data due to shifts in gauge location and corrected precipitation data based on the relationship between precipitation and elevation. In addition, the missing discharge data were completed using a non-linear recession model (Wittenberg, 1994) that was applied to fill-in data of maximum fifteen consecutive days of missing discharge values. Streamflow data that were unavailable for more than fifteen consecutive days were excluded from the analysis. This fill-in procedure concerns less than 5% of the data. For the missing climate data, a normal ratio method based on values from surrounding stations was used.

3.3. Methods

3.3.1. Land use changes

Because land use information for the years 1994 and 2013 is available from Marhaento et al. (2017b), additional satellite image analysis was carried out for Landsat and Aster images for the years 2000 and 2006. These images were completed by pre-processing analysis through a non-systematic geometric correction to avoid distortion on map coordinates and masking analysis to remove the area beyond the study area. The Aster image resolution was resampled from 15×15 m² grid-size to 30×30 m² grid-size to be similar to the grid size of Landsat images. A maximum likelihood approach was used to perform image classification using thousand ground-control points (GCPs; see Marhaento et al., 2017b), and then a confusion matrix was made to measure classification accuracy. The GCPs were divided into two sets: half of the GCPs were used to perform image classification and half of them were used to perform accuracy assessments by means of the overall accuracy and the Kappa coefficient. The overall accuracy is defined as the number of correct samples divided by the total number of samples. The Kappa coefficient measures the agreement between the classification map and the reference data. Land use change analysis was performed based on differences in areal coverage of each land use class between different years.

3.3.2. SWAT model set up and calibration

3.3.2.1. Water balance estimation

This study used the water quantity module of the SWAT model, a physically based semi-distributed model operating on a daily time step with proven suitability for hydrologic impact studies around the world (Wagner et al., 2013; Memarian et al., 2014). The SWAT model divides a catchment into subcatchments and then further divides each sub-catchment into hydrological response units (HRUs) for which a land-phase water balance is calculated (Neitsch et al., 2011). A HRU is defined as a lumped area within a catchment assumed to have uniform behavior and is characterized by a dominant land use type, soil type and slope conditions (Arnold et al., 1998). For each simulation time step (daily in this study), a water balance is computed for each HRU and for each sub-catchment. Runoff is routed through channels to the catchment outlet and used to calculate the catchment water balance.

This study focused on the land-phase water balance where land use change has its main impact. The components of the land-phase water balance in the SWAT model include inflows, outflows and variations in storages. Precipitation (P) is the main inflow in the model. The outflows are actual evapotranspiration (ET), surface runoff (Q_s), lateral flow (Q_l) and base flow (Q_b). There are four water storage possibilities in SWAT namely snowpack, soil moisture (SM), shallow aquifer (SA) and deep aquifer (DA). The snowpack storage was excluded in the analysis because snowfall is not relevant in the study area. Flows between storage (Perc), water movement upward from the shallow aquifer to the soil moisture storage (Revap) and deep aquifer recharge (DA_Rchg). Figure 3.3 shows the SWAT model schematization used in this study. For a more detailed description of the SWAT model, reference is made to Neitsch et al. (2011).

3.3.2.2. Model setup

The DEM was used to delineate boundaries of the catchment and was used to divide the catchment into sub-catchments. A stream network from the Indonesia Geospatial Information Agency was used to "burn-in" the stream network to create accurate flow routing and to delineate sub-catchments. This procedure resulted in 11 sub-catchments, ranging in size from 0.12 km² to 83 km². In addition, the DEM was used to generate a slope map based on the slope classification from the Guideline of Land Rehabilitation from the Ministry of Forestry (1987).



Figure 3.3. The Soil Water Assessment Tool model structure. DA = deep aquifer; ET = evapotranspiration; HRU = hydrological response unit; Q_b = base flow; Q_l = lateral flow; Q_s = surface runoff; SA = shallow aquifer; SM = soil moisture

Land use codes from the SWAT database, namely AGRR, FRSE, FRST, RICE, URMD and LBLS, were assigned to denote dryland farming, evergreen forest, mixed garden, paddy field, settlement and shrub, respectively. For the settlement, this study used the assumption that the settlement area is not fully impervious providing some pervious spaces in between the houses that are often used for house yards. Thus, the class Urban Residential Medium Density (URMD) in the SWAT model was used to assign parameters in the settlement area. URMD assumes an average of 38% impervious area in the settlement area (Neitsch et al., 2011), which is relatively similar to the settlement conditions in the study catchment. In addition, as a typical rural area in Java, the artificial drainage network in the study area is not well developed. Thus, the impacts of the artificial drainage network of the settlements on the long-term water balance of the system are assumed to be not significant. Furthermore, the SWAT crop and management database were modified for each land use type in order to match the conditions in a tropical region. It should be noted that SWAT is designed for temperate regions so that it is necessary to modify the crop parameters for application in a tropical region (Kilonzo, 2013; Griensven et al. 2014). Table 3.1 shows the adjusted values in the SWAT crop and management database applied in this study. Forest includes evergreen forest and mixed garden, whereas agriculture includes paddy field and dry-land farming.

HRUs were created by spatially overlying maps of land use, soil and slope classes. The Penman-Monteith method was used to calculate reference evapotranspiration. For runoff simulations, the Soil Conservation Service Curve Number (SCS-CN) method was selected because it has a direct link to land use types. In addition, the CN values were adjusted for slope effects based on the

equation from William (1995), since the default CN value in the SWAT model is only suitable for a slope up to 5%, which is not appropriate for the study catchment. For flow routing, the Muskingum method that models the storage volume as a combination of wedge and prism storage (Neitsch et al., 2011) was used.

	Land use/land cover class								
Variable	Fo	rest	Sh	rub	Agriculture				
	Original	Adjusted	Original	Adjusted	Original	Adjusted			
LAI_INIT (-)	0	5	0	3	0	3			
BIO_INIT (kg/ha)	0	1,000 ^a	0	0	0	0			
PHU_LT (-)	0	3,500 ^a	0	3,500 ^a	0	3,500 ^a			
GSI (mm/s)	2	7	5	5	7	7			
CANMX (mm)	0	5	0	2	0	2			

Table 3.1. Adjusted values in the SWAT crop and management database (adapted from Kilonzo, 2013).

Note. Definitions for each variable are available in Neitsch et al. (2011). ^aThe maximum value allowed in SWAT. (-) means no units

3.3.2.3. Model calibration

Model calibration aims to produce a robust SWAT model that is able to simulate land use change impacts on the water balance. Identification of SWAT parameters to be calibrated was based on the procedure from Abbaspour et al. (2015). Within the procedure, the SWAT model was executed using the default SWAT parameters. The simulated hydrograph was visually compared with the observed hydrograph. Based on the characteristics of the differences between observed and simulated hydrograph (e.g. underestimation/overestimation discharge and shifted discharge), relevant SWAT parameters were identified. Furthermore, one at a time sensitivity analysis was carried out to detect the most sensitive parameters among the relevant parameters.

The model was calibrated using the Latin Hypercube Sampling approach from the Sequential Uncertainty Fitting version 2 (SUFI-2) in the SWAT-Calibration and Uncertainty Procedure (SWAT-CUP) package. First parameter ranges were determined based on minimum and maximum values allowed in SWAT. A number of iterations were performed where each iteration consisted of 1,000 simulations with narrowed parameter ranges in subsequent calibration rounds. The new parameter ranges were determined based on the lowest and highest parameter value of the 10% best parameter sets according to the objective function value. The calibration process was stopped and the optimum parameter values were obtained after the objective function value showed insignificant changes (smaller than 0.01). Simulations for model calibration were assessed on a monthly basis and the Nash Sutcliffe Efficiency (NSE) (Eq.3.1) was used as the objective function.

$$NSE = 1 - \frac{\sum_{i=1}^{i=N} [Qs(i) - Qo(i)]^2}{\sum_{i=1}^{i=N} [Qo(i) - \overline{Qo}]^2}$$
(Eq. 3.1)

where *i* is the time step, *N* the total number of time steps, *Qs* the simulated discharge, *Qo* the observed discharge and $\overline{Q_0}$ the mean of *Qo* over the period of analysis.

3.3.3. SWAT simulations

The SWAT simulations were carried out in two steps. The first step included simulations to attribute changes in streamflow to land use change. These simulations aimed to investigate whether land use change is the main driver for changes in streamflow. This study used a baseline-altered approach, which divides the dataset into four sequential periods: the first period is regarded as the baseline period, and other periods are regarded as altered periods, periods when the hydrological regime within a catchment is expected to be changed due to land use changes. The method assumes that by calibrating a model in the baseline period and applying the calibrated model in the altered periods using updated land use maps will enable the simulation of impacts of land use change on streamflow (Li et al., 2009 and Wagner et al., 2013).

In the present study, the simulation period (1990–2013) was divided into four equal periods to accommodate gradual changes in land use in the study catchment. The first period (1990–1995) was regarded as the baseline period and the other periods (1996–2001, 2002–2007 and 2008–2013) were regarded as altered periods. The four land use maps available for the years 1994, 2000, 2006 and 2013 represent the land use conditions for each period. The calibrated SWAT model was run for the baseline period (1990–1995) using land use of the year 1994 and then was applied for the altered periods. This study carried out two simulations for the altered periods i.e. with land use change (using land use maps of 2000, 2006 and 2013) and without land use change in order to investigate the attribution of changes in streamflow to land use change. This

study followed the method from Refsgaard et al. (1989) and Lørup et al. (1998), which used model performances as an indicator of land use change impacts. Land use change is regarded as the important driver to changes in streamflow when model simulations with land use changes show a better model performance than simulations without land use change. Updating land use cover in the simulations will adapt the HRU information in the simulation and thus affect streamflow simulation. Figure 3.4 shows the hypothetical scenario of the simulations. It should be noted that the changes in the model performance values are not necessarily linearly related to changes in land use as suggested by the straight lines in the diagram. To incorporate land use change in the simulations, the Land Use Update (LUP) tool in ArcSWAT was used.



Figure 3.4. A hypothetical scenario for the model simulations to test the hypothesis regarding the effects of land use changes on the model performance. NSE = Nash–Sutcliffe efficiency

The first step of the simulations shows the land use change contribution to changes in streamflow. As a second step in the study, the changes in the mean annual water balance as a result of land uses change were assessed at both catchment level and sub-catchment level. This study focused on five water balance ratios, namely the ratio of streamflow to precipitation (the runoff coefficient, Q/P), the ratio of evapotranspiration to precipitation (ET/P), the ratio of surface runoff to streamflow (Q_s/Q), the ratio of lateral flow to streamflow (Q_l/Q) and the ratio of base flow to streamflow (Q_b/Q). A diagnostic approach using statistical analyses (Mann-Kendal statistic and Pearson statistic) was carried out to test the statistical significance of the long-term trends in the mean annual water balance ratios and to investigate the correlation between changes in the water balance and land use change.

3.4. Results

3.4.1. Land use change analysis

Table 3.2 shows the relative areas of each land use class for the years 1990, 2000, 2006 and 2013. In general, there was an increasing human influence in the Samin catchment through the development of settlements and agriculture area (i.e. paddy field and dryland farm). Figure 3.5 shows that in the period 1994-2000, agriculture area was converted to settlements in the downstream area and evergreen forest area was converted to mixed garden in the upstream area. However, these changes do not significantly reduce the tree-dominated area because the conversion of evergreen forest into mixed garden area basically is a change from a state forest into a community forest. A dramatic change has occurred in the period 2000-2006 when agriculture land was converted to mixed garden area in the upstream area that resulted in a major decrease of tree-covered area in the Samin catchment. At the same time, the settlement area significantly increased in the downstream area converting agricultural area into impervious area. In the period 2006–2013, the development of settlement areas expanded in the upstream part of the catchment at the expense of mixed garden area resulting in major deforestation. In total, during the period 1994-2013, deforestation by decreasing evergreen forest area and mixed garden area was about 30%. At the same time, settlements and agricultural area have increased by 25% and 6%, respectively. It should be noted that the average overall accuracy and Kappa coefficient of land use maps used for this study were 89% and 87%, respectively, which is sufficient for producing a land use map (Congalton, 1991). Detailed descriptions of each land use class are available in Marhaento et al. (2017b).

Land use class	1994	2000	2006	2013
1. Forest area	48.7	42.2	26.0	16.9
a. Evergreen forest	16.1	3.0	2.4	2.3
b. Mixed garden	32.6	39.2	23.6	14.6
2. Agriculture area	39.2	36.9	44.3	45.4
a. Paddy field	30.2	28.2	28.9	28.0
b. Dryland farm	9.0	8.7	15.4	17.4
3. Settlements	9.8	16.3	25.6	34.3
4. Shrub	2.0	4.3	3.8	3.1
5. Water body	0.3	0.3	0.3	0.3

Table 3.2. Area percentage (%) of each land use class in the year 1994, 2000, 2006 and 2013



Figure 3.5. Land use distribution in the Samin catchment in the years 1994 (a), 2000 (b), 2006 (c) and 2013 (d). Numbers in Figure 3.5a show the sub-catchment identity number.

3.4.2. Model calibration and simulations

3.4.2.1. Model calibration

Six SWAT parameters were calibrated, namely CN2, SOL_AWC, ESCO, CANMX, GW_DELAY and GW_REVAP. CN2 is the curve number parameter, which controls the fraction of water to infiltrate into the soil or to generate surface runoff by overland flow. A large CN value causes less infiltration and results in more surface runoff. SOL_AWC is the available water capacity of the soil layer. It controls the soil water storage between soil moisture conditions at field capacity and permanent wilting point. GW_DELAY is the ground water delay time. It controls the lag time between water leaving the soil layers and entering the shallow aquifer. A larger GW_DELAY will enable more evaporation from the

unsaturated zone. GW_REVAP is the ground water revap coefficient, a SWAT term to describe the movement of water into overlying unsaturated layers as a function of water demand for evapotranspiration. The GW_REVAP parameter controls the amount of water in the capillary fringe that separates the unsaturated zone and the saturated zone to move upward (to fulfill evaporative demand). A larger GW_REVAP value results in a larger transfer rate from the shallow aquifer to the unsaturated zone. ESCO is the soil evaporation compensation factor and controls the soil evaporative demand that is to be met. When the value of ESCO gets closer to zero, the model will receive more water from the lower soil level to fulfil evaporative demand. CANMX is the maximum canopy storage to intercept precipitation. It controls the density of plant cover so that it significantly affects infiltration and evapotranspiration. A larger CANMAX results in a larger plant canopy capacity. CANMX was calibrated only for forested areas (i.e., FRSE and FRST).

These parameters were calibrated in different ways. For parameters CN2 and SOL_AWC, a scaling value at the HRU level was calibrated so that the default parameter value was multiplied by (1+ calibrated scaling value). In this way, the default parameters will not lose their original spatial patterns. For parameters GW_DELAY, GW_REVAP, ESCO and CANMX, parameter values were calibrated at the catchment level so that the calibrated parameter values are homogeneous over the entire catchment. After three rounds of calibration, this study found optimum calibrated parameter values with a NSE value of 0.78 (see Figure 3.6). Table 3.3 shows the calibrated values used in this study.



Figure 3.6. Hydrograph of simulation with calibrated parameter values for the baseline period

Parameters	Units	Prior range	Default value	Calibrated values
CN2	-	-0.3 – 0	0	- 0.1
SOL_AWC	mm	0 – 1	0	0.6
ESCO	-	0 – 0.5	0.95	0.1
GW_DELAY	Days	30 – 80	31	55
GW_REVAP	-	0.02 - 0.1	0.02	0.05
CANMX	mm	0 - 10	5	9

Table 3.3. Values of calibrated Soil Water Assessment Tool parameters

(-) means no units

3.4.2.2. Performance of model simulations with and without land use changes

Table 3.4 shows the model performance for simulations with and without land use changes for the altered periods. The results of the simulations without land use change show that the NSE value has continuously decreased, with a significant decrease in the period 2008–2013 (Figure 3.7). In this period, the calibrated model fails to simulate peak flows, which explains the relatively low value of NSE (<0.6). After land use change was incorporated in the simulations, the NSE values of the altered periods (periods 2, 3 and 4) were higher than the NSE values from simulations without land use change are better than in the simulations without land use change are better than in the simulations without land use change. While the mean NSE value of simulations with land use change is around 0.6, the mean NSE value of simulations with land use change is around 0.7.

Simulations	Baseline period		Altered periods		
	1990–1995	1996-2001	2002-2007	2008–2013	
Without land use update	0.78	0.70	0.65	0.53	
With land use update	0.78	0.72	0.71	0.68	

Table 3.4. NSE values of simulations with and without land use update



Figure 3.7. Hydrograph of simulations with and without land use change for the altered periods.

In order to validate the method, a similar procedure was applied in an independent sub catchment namely Tapan catchment (±200 ha), that is located in the upstream part of the study catchment. The streamflow data of the Tapan catchment were made available by Retnowati (2012) at annual basis from the period 1993 to 2007. A new outlet was created in the simulated catchment that is similar to the outlet location of the Tapan catchment, and then the model was executed with and without land use change. Furthermore, the NSE values between simulations with and without land use change for the period 1993–2007 were compared. The results show that simulation with land use change has a better performance than the simulation without land use change with NSE values of -0.75 and -4.03, respectively (see Figure 3.8). Although the model showed a poor performance, a better model performance for simulations with land use change than simulations without land use change could be an indication of the importance of land use change to changes in streamflow.

3.4.3. Changes in water balance ratios

A good model performance of simulations with land use change confirmed the contribution of land use change to changes in streamflow. The results of the simulation with land use change showed that the water balance of the Samin catchment has changed in the 24-year time period at both catchment and sub-catchment level. At the catchment level, the Mann Kendall trend analysis showed that the changes in the runoff coefficient and ET/P were not significant at a significance level of 5%.



Figure 3.8. Hydrograph simulations with and without land use change for years 1993-2007 in the validation catchment

Between period 1 and period 4, the runoff coefficient has increased from 35.7% to 44.6% and ET/P has decreased from 60% to 54.8%. The results of this study are in line with the findings from Marhaento et al. (2017b) but this study found smaller increases for the mean annual streamflow. Despite changes in runoff coefficient were not statistically significant, changes in Q_s/Q and Q_b/Q were significant at a significance level of 1%. Between period 1 and period 4, Q_s/Q has increased from 26.6% to 37.5% whereas Q_b/Q has decreased from 40% to 31.1%. For Q_t/Q , the long-term changes were not significant at a significance level of 1.4%). Figure 3.9 shows the long-term changes in the water balance ratios for the period 1990–2013.

At the sub-catchment level, long-term trends were consistent for three water balance ratios namely runoff coefficient, Q_s/Q and Q_b/Q . It was found that the runoff coefficient and Q_s/Q increased while Q_b/Q decreased for all sub-catchments. However, long-term trends were not consistent for the other two water balance ratios ET/P and Q_t/Q . Several sub-catchments experienced a positive trend of ET/P as well as Q_t/Q while other sub-catchments showed negative trends. The opposite directions of ET/P and Q_t/Q at sub-catchment level have balanced each other so that the differences of ET/P and Q_t/Q at catchment level were minor. As a result, changes in ET/P were not equal to changes in Q/P at catchment level indicating that there was a change in the water storage (Δ S). Table 3.5 shows the results of the Mann Kendall trend analysis of the water balance ratios at sub-catchment level.



Figure 3.9. (a) Simulated runoff coefficient (Q/P) and ET/P and (b) simulated Q_s/Q , Q_t/Q and Q_b/Q including their long-term trend for the period 1990-2013 in the Samin catchment

 Sub-catchment	Area (km ²)	ET/P	Q/P	Q _s /Q	Qı/Q	Q _b /Q	
1	2.5	-	+	+	-	-	
2	8.2	-	+	+	-	-	
3	13.7	-	+	+	-	-	
4	4.6	+	+	+	+	-	
5	0.1	+	+	+	+	-	
6	9.5	+	+	+	+	-	
7	21.3	-	+	+	-	-	
8	17.8	-	+	+	-	-	
9	60.8	-	+	+	-	-	
10	55.4	-	+	+	+	-	
11	84.0	-	+	+	-	-	

Table 3.5. Trend directions from the Mann Kendall trend analysis for the water balance ratios per sub-catchment for the period 1990-2013.

A (-) shows a decreasing trend, while a (+) shows an increasing trend. Shading means that the trend is significant at a significance level of 5%.

3.4.4. Effect of land use change on the water balance

In order to relate changes in the water balance and changes in land use, Pearson correlation tests were performed at the sub-catchment level. The results showed that changes in the water balance were significantly correlated with changes in forest area (i.e. combination of evergreen forest and mixed garden) and settlements, whereas the other land use classes did not show a correlation with the water balance ratios at a significance level of 5%. Figure 3.10 shows the changes in the water balance ratios as a function of changes in forest and settlement area. Changes in the forest area had a positive correlation with the runoff coefficient and a negative correlation with ET/P at a significance level of 1%. In addition, changes in the settlement area had a positive correlation with the runoff coefficient at a significance level of 1%, but relatively low impacts on ET/P. Changes in the runoff coefficient and ET/P at sub-catchment level were higher than changes at catchment level. For the fractions of streamflow, changes in the Q_s/Q could be related to changes in the settlement area with a positive correlation at a significance level of 5%. Changes in Q_b/Q had the same magnitude but in an opposite direction than changes in Q_s/Q. The contribution of changes in forest area to changes in Q_{2}/Q and Q_{1}/Q were minor in particular for the latter. Changes in forest area significantly affected changes in Q/Q with a positive correlation at a significance level of 5%.

3.5. Discussion

Based on hydrological modelling, this study provides a strong indication that changes in land use altered the water balance in the Samin catchment. Findings show that land use change due to deforestation and expansions of settlement area (i.e. urbanization) have reduced the mean annual evapotranspiration and increased the mean annual streamflow. Moreover, the fraction of the streamflow originating from surface runoff has significantly increased compensated by a decrease in base flow. The directions of changes in the water balance by land use change (i.e. deforestation and urbanization) in this study are in line with hydrological studies in tropical regions from Bosch and Hewlett (1982), Bruijnzeel (2004) and Brown et al. (2005). It is well known that a reduction of forest area may cause not only a reduction in the tree stands which significantly reduces evapotranspiration from both canopy interception and plant transpiration, but also diminishes ground vegetation and leaf litter. The absence of ground vegetation and leaf litter affects top soils to lose protection from raindrop splashes and to have less biotic activity (Guevara-Escobar et al., 2007). Thus, processes in the top soil resulting in lower organic matter contents cause a lower permeability and storage capacity resulting in a lower infiltration capacity.



Figure 3.10. Changes in water balance components as a function of changes in forest area and settlements, including Pearson correlation coefficients (R). ET/P = ratio of evapotranspiration to precipitation; Q_b/Q = ratio of base flow to streamflow; Q_l/Q = ratio of lateral flow to streamflow; Q_s/Q = ratio of surface runoff to streamflow

As a result, a larger fraction of precipitation is transformed into surface runoff. In the present study, it is found that changes in forest area significantly contributed to changes in ET/P and Q_I/Q whereas changes in the Q_s/Q and Q_b/Q were profoundly affected by the expansion of settlement area. A rapid increase of settlement area (see Table 3.2) is probably the main reason of the significant impacts on the water balance. Expansion of settlement area will reduce vegetation coverage thus leading to a decrease in evapotranspiration. Expansion of impervious area (e.g. buildings, roads) can alter both surface and subsurface flow as a result of reduced infiltration. As impervious area increases, less water infiltrates into the soil resulting in lower soil water storage. Several studies have investigated the correlation between urbanization and the water balance in tropical regions (Wagner et al., 2013; Gumindoga et al., 2014a; Remondi et al., 2016) and results are similar to ours. In this study, changes in the water balance due to land use change occurred at both catchment and sub-catchment level. However, it was found that changes at sub-catchment level were diverse where some sub-catchments showed different directions of change in the water balance ratios compared to others. Apparently, the positive and negative trends on ET/P and Q_i/Q at sub-catchment level have compensated each other, so that long-term changes of those variables at catchment level were relatively small. This is similar to the findings from Wagner et al. (2013) and Wilk and Hughes (2002) who argue that the complexity of large catchments that can mask the impacts of land use changes on hydrological processes at the local scale. Furthermore, the land use change impacts on the water balance differ for each sub-catchment. The changes in the downstream area, where land conversion from agriculture to settlement is dominant, are larger than in the upstream area, where land conversion from forest to agriculture is dominant. The rapid increase of impervious area is probably the main cause for changes in the overall water balance. This study agrees with Wagner et al. (2013) who argue the significant impacts of urban expansion to changes in the surface runoff and streamflow. In addition to these factors. Van Dijk et al. (2007) and Bruijnzeel (2004) mention slope variation, soil types and geological conditions which could contribute to the impact of land use change on the water balance. Nonetheless, changes in climate can also accelerate changes in the water balance of the study catchment. Aldrian and Diamil (2008) argue that the climate in East Java has changed over the past decades with a decrease in the accumulated monthly precipitation and annually, an increase of the ratio of total precipitation in the wet season to the total annual precipitation and an increase of the dry spell period. However, climate changes in East Java likely occurred in the low altitude area closer to the seashore and relatively less in the mountainous area due to regular orographic effects. Marhaento et al. (2017b) argue that the contribution of climate change to changes in the streamflow in the study catchment is less significant than the contribution of land use change but not negligible. As can be seen in Figure 3.2, the runoff coefficient of the years 2000 and 2005 was higher than in other years, which is probably due to climatic variability (e.g. higher precipitation intensity) rather than land use. Thus, combination of land use change and climate change might result in more significant hydrological changes than either driver acting alone (Hejazi & Moglen, 2008). A more thorough study, however, need to be done in the future focusing on the role of climate change (e.g. changes in the precipitation intensity, extreme precipitation events) to changes in the water balance of the study catchment.

The SWAT model used in the present study has been successful to attribute changes in the water balance to land use changes. However, a good model performance only was found at the catchment level, for which the model calibration has been carried out. At the sub catchment level, the model errors increased. A significant deterioration of the model performance at smaller scale sub catchments probably result from model parameters that cannot represent detailed hydrological processes at a finer scale such as flow concentration during heavy downpours. In addition, accumulation of errors from the model structure (e.g. model assumptions, equations) as well as data inputs (e.g. lack of relevant spatial and temporal variability of data inputs from precipitation, soils, land uses and DEM) was increasing when the model was applied under changing conditions (e.g., land use change; Ewen et al., 2006)

It should be noted that parameterization is the key yet most challenging part in the model simulation. Identification of parameters to be calibrated was essential for the subsequent steps. This study followed the guideline from Abbaspour et al. (2015) to select parameters to be calibrated based on the model performance of the default simulation. However, these parameters were different from the parameters that were found by the global sensitivity analysis in the SWAT-CUP package (CN2, CH N2, GWQMN, GW DELAY, GW REVAP and CNCOEF). Calibration of both parameter sets resulted in a comparable model performance in terms of NSE, where parameters from the global sensitivity analysis resulted in an average NSE value of 0.77 for simulations with land use change and 0.6 for simulations without land use change. Although a similar conclusion could be drawn by using the parameter set from the global sensitivity analysis, it was found that the model simulated a (very) low ratio of catchment averaged evapotranspiration to precipitation (in mm) with an average annual value of 34%. According to Bonell et al. (2005), in forested tropical catchments with distinct wet and dry seasons, the average annual ET is more than 50% of the annual precipitation. Findings of this study resemble Griensven et al. (2012), who reported the equifinality problem in SWAT simulations for tropical regions where satisfactory simulated discharges does not guaranty a good performance of the ET simulation.

The SWAT model has been developed for temperate regions and the default SWAT database is not applicable to other regions including the tropics. The SWAT model considers dormancy for the perennial trees (Neitsch et al., 2011), which is not the case in a tropical forest. Because of the dormancy, the Leaf Area Index (LAI) for the tree species is set to a minimum value during the dormancy period and this result in a low annual ET. Kilonzo (2013) argues to adjust several crop SWAT parameters (initial Leaf Area Index, initial biomass, heat units for perennial trees and plant-harvest schedule for crops) necessary to simulate streamflow in forested areas of tropical regions. In addition, Griensven et al. (2014) argue that parameters of the maximum canopy storage (CANMX) and maximum stomatal conductance at high solar radiation and low vapor
pressure deficit (GSI) are sensitive to evapotranspiration and therefore it is necessary to adjust the values in order to increase ET. Adjusting the crop parameters and management database as seen in Table 3.1 prior to model calibration has resulted in a more realistic simulation as demonstrated in this article. This adjustment caused an increase of ET by 10% compared to default crop parameters.

For further research on the impacts of land use change on the water balance in tropical regions, this study suggests to include ET in the calibration process. The use of remote sensing products to generate spatially and temporally distributed ET can contribute to this process (Rientjes et al., 2013 and Griensven et al., 2014).

3.6. Conclusions

This study was able to attribute changes in the water balance to land use change in the Samin catchment using the approach described in this paper. The results show that land use change was the dominant driver for changes in the water balance as shown by the better model performance for a simulation with land use change compared to a simulation without land use change. The simulation without land use change generally failed to simulate peak flows where the error was increasing in time and thus resulted in a worsening model performance.

Land use change in the study catchment was mainly shown as an increase in settlement area and a decrease in forested area. At catchment level, these land use changes caused an increase in the runoff coefficient and decrease in the ratio of evapotranspiration to precipitation. The ratio of surface runoff to streamflow substantially increased and the ratio of base flow to streamflow substantially decreased. At the sub-catchment level, the impacts of land use changes on the water balance were diverse for the ratio of evapotranspiration to precipitation and the ratio of lateral flow to steam flow. Positive and negative trends at sub-catchment level have compensated each other, so that long-term changes of those variables at catchment level were relatively small. In addition, changes in forest and settlement area have resulted in different impacts on the water balance. A reduction in the forest area significantly contributed to a higher runoff coefficient and lower ratios of evapotranspiration to precipitation and lateral flow to streamflow. An increase in the settlement area significantly contributed to a higher runoff coefficient and ratio of surface runoff to streamflow and a lower ratio of base flow to streamflow.

Chapter 4

Hydrological response to future land use change and climate change in a tropical catchment³

Abstract

Hydrological response to expected future changes in land use and climate in the Samin catchment (278 km²) in Java, Indonesia, has been simulated using the Soil and Water Assessment Tool model. Changes between the baseline period 1983–2005 and the future period 2030-2050 under both land use change and climate change are analyzed. This study uses the output of a bias-corrected Regional Climate Model and the outputs of six Global Climate Models to include climate model uncertainty. The results show that land use change and climate change individually will cause changes in the water balance components, but that more pronounced changes are expected if the drivers are combined, in particular for changes in annual streamflow and surface runoff. The findings of this study will be useful for water resource managers to mitigate future risks associated with land use and climate changes in the study catchment.

³ This chapter is based on a paper that has been submitted as: Marhaento, H., Booij, M. J., & Hoekstra, A. Y. Hydrological response to future land use change and climate change in a tropical catchment.

4.1. Introduction

Climate change and land use change are key factors determining changes in hydrological processes in catchments. Numerous studies have been carried out to evaluate the impacts of land use and climate change on water resources (Legesse et al., 2003; Li et al., 2009; Mango et al., 2011; Wang, 2014; Marhaento et al., 2017a; 2017b; Shrestha & Htut, 2016; Zang et al., 2016). Most findings show that changes in land use and climate affect hydrological processes such as evapotranspiration, interception and infiltration, resulting in spatial and temporal alterations of surface and subsurface flows patterns (Legesse et al., 2003; Bruijnzeel, 2004; Thanapakpawin et al., 2007; Khoi & Suetsugi, 2014; Marhaento et al., 2017a). According to Wohl et al. (2012), hydrological processes in the humid tropics differ from other regions in having greater energy inputs and faster rates of change, including human-induced changes, and therefore require additional study. IPCC (2007) reported that tropical regions, including Indonesia, are one of the most vulnerable areas for future water stress due to extensive land use and climate changes.

With a population of more than 130 million (in 2010), Java, Indonesia, is one of the most densely populated islands of the world. Over the last century, land use on Java has changed rapidly, following the rapid growth of human population (Verburg and Bouma, 1999), which has resulted in significant changes in the water system. Bruijnzeel (1989) observed higher flows during rainy seasons and lower flows during dry seasons after a fair proportion of forest area was transformed into settlements and agriculture in the Konto catchment (233 km²) in East Java. Studies from Remondi et al. (2016) and Marhaento et al. (2017a) present similar results, showing that land use change due to deforestation and expansions of settlement area have reduced mean annual evapotranspiration and increased mean annual streamflow. In addition, the fraction of streamflow originating from surface runoff significantly increased, compensated by a decrease in base flow.

Besides land use change, Java has experienced climate change in the past decades. Aldrian & Djamil (2008) found a significant change in the spatial and temporal climate variability over the Brantas catchment (12,000 km²) in East Java over the period 1955–2005. They found a decrease in annual precipitation, an increase of the precipitation intensity during the wet season and an increase of the dry spell period. More pronounced changes likely occurred in the low altitude area closer to the seashore. Their findings resemble those of Hulme and Sheard (1999), who found that most islands of Indonesia have become warmer since 1900, reflected in an increase of the annual mean temperature by about 0.3°C. Moreover, the mean annual precipitation has likely declined in the

southern regions of Indonesia including Java, resulting in a significant change in the water availability.

Changes in land use and climate in Java have threatened local and regional socio-economic development. Amien et al. (1996) and Naylor et al. (2007) argue that land use and climate change in Java have caused a decrease in the rice production resulting from a warming climate as well as a decreasing farming area. In addition, many reservoirs have failed, having a lower life span and water supply capacity than expected due to sedimentation from deforested upstream areas (Moehansyah et al., 2002). Furthermore, the frequency of disastrous events related to land use and climate change (e.g. droughts and floods) has increased, resulting in major economic losses in Java during the last decades (Marfai et al., 2008). Without taking any mitigation measures, Java is projected to have a severe food crisis by the year 2050 due to land use and climate changes (Syaukat, 2011).

Land use planning can be an effective way to mitigate future risks associated with changes in land use and climate (Memarian et al., 2014). Numerous studies argue that different types of land use have different water use and water storage characteristics (Bruijnzeel, 1989, 2004; Legesse et al., 2003; Memarian et al., 2014). However, it is a challenge to measure the effectiveness of land use planning for improving availability of water resources due to climatic interference. Complex interactions between land use and climate changes may not only result in accelerating changes in hydrological processes (Legesse et al., 2003; Khoi & Suetsugi, 2014), but may also offset each other (Zhang et al, 2016), which requires further study.

In order to provide insight in land use and climate change impacts on hydrological processes, coupled models are typically used. For example, Zhang et al. (2016) used a combination of a Markov chain model and a Dynamic Conversion of Land Use and its Effects (Dyna-CLUE) model to simulate future land uses, climate change scenarios to predict future climate variability and the Soil and Water Assessment Tool (SWAT) model to simulate hydrological processes in order to quantify the hydrological impacts of land use and climate changes. However, very few studies have assessed hydrological impacts due to land use and climate changes in tropical regions, which is mainly due to limited hydrological data in tropical regions for model calibration and validation purposes (Douglas, 1999). Furthermore, the results are often contradictory and inconsistent, in particular for large catchments (>100 km²) where interference from climate change becomes more important (Calder et al., 2001; Beck et al., 2013).

The present study aims to assess future hydrological response to changes in land use and climate in the Samin catchment (278 km^2) in Java, Indonesia. The

Samin catchment is selected following prior studies from Marhaento et al. (2017a; 2017b), who argue that historic land use change in the Samin catchment has significantly affected the hydrological processes in this catchment. Central Java Provincial Government (2010) has introduced a spatial land use planning for the Samin catchment through a regional regulation, which motivated this study to assess its effectiveness to mitigate future risks of water resources under different climatic circumstances. This study assesses the separate and combined effects of land use change and climate change on water balance components for the period 2030–2050 through plausible future land use and climate change scenarios.

4.2. Study Area and Data Availability

4.2.1. Catchment description

The Samin catchment (278 km²) is part of the Bengawan Solo catchment, the largest catchment in Java, Indonesia, which plays an important role to support life of more than half a million people within its area. It is located between 7.6° -7.7° southern latitude and 110.8°-111.2° eastern longitude. The highest part of the catchment is the Lawu Mountain (3,175 meter above sea level) and the lowest part is near the Bengawan Solo river (84 meter above sea level; see Figure 4.1). The upper part of the Samin catchment is characterized by steep terrain (>25%) and predominantly covered by evergreen forest. A less undulated terrain is in the middle part of the catchment, which is mostly covered by mixed garden, agricultural crops and settlements. In the downstream part, agriculture (mainly paddy fields) and settlements are dominant. According to the soil map from the Harmonized World Soil Database (FAO et al., 2012), the soil distribution of the Samin catchment is predominantly Luvisols (a leafy humus soil) and Andosols (a volcanic soil of the Mount Lawu), which cover 57% and 43% of the study area, respectively. Geologically, the Samin catchment is located in the depression zone filled by Mount Lawu volcano's deposits that resulted in deep and fertile soils and thus are suitable for agriculture.

The Samin catchment experiences a tropical monsoon climate with distinct dry and wet seasons, whereby the former generally extends from May to October and the latter from November to April. Mean annual precipitation can be 1,500 mm in dry years and reaches 3,000 mm in wet years. The spatial precipitation pattern likely follows the orography, with a larger amount of precipitation in the upstream than in the downstream area. The mean daily temperature is approximately 26 ^oC, with a mean daily minimum of 21.5 ^oC and a mean daily maximum of 30.5 ^oC. The annual potential evapotranspiration in the catchment ranges from 1,400 to 1,700 mm (Marhaento et al., 2017b). According to the Indonesia Statistical Bureau (BPS, 2017), the population size at the sub-district level in the Samin catchment is about 800.000 inhabitants, with an average annual population growth over the period 1994–2010 of about 0.8%. In the same period, land use has significantly changed, with an increase in the settlement area and a decrease in the forest area (Marhaento et al., 2017a). Population growth in the Samin catchment has been projected to decrease over time, reaching 0.1% per year in 2035, as a result of a successful birth control programme as well as a transmigration programme (BPS, 2013), which may affect future land use change in the study catchment.



Figure 4.1. Samin catchment in Java, Indonesia, with the locations of hydrological gauges

4.2.2. Data Availability

To set up the hydrological model, spatial and non-spatial data were used. For the spatial data, land use maps with 30-meter spatial resolution for the years 1994, 2000, 2006 and 2013 were available for the study area from Marhaento et al. (2017a). A land use spatial planning map of the study area for the period 2009–2029 was available from Central Java Provincial Government (2010). The topographic map contains information related to elevation, roads and locations of public facilities (e.g. hospitals, schools, and offices) and is available from the Geospatial Information Agency of Indonesia at 1:25,000 scale. A soil map at 30 arc-second spatial resolution was taken from FAO et al. (2012).

For the non-spatial data, daily water level data for the period 1990-2013 were available from the Bengawan Solo River Basin organization and converted into daily discharge data using the rating curves provided by the data provider. Daily climate data were available from eleven precipitation stations and three climate stations within the surrounding catchments for the period 1983-2013. Future climate data for the period 2030-2050 for different emission scenarios were made available from SEACLID/CORDEX Southeast Asia (CORDEX-SEA), a consortium consisting of experts from 14 countries and 19 institutions that aim to downscale a number of Global Climate Models (GCMs) from the Fifth Coupled Model Inter-comparison Project (CMIP5) for the Southeast Asian region. For Indonesia, the Indonesian Agency for Meteorological, Climatological and Geophysics (BMKG) provided the Regional Climate Model (RCM) RegCM4 data at 25×25 km² resolution (Ngo-Duc et al., 2016), downscaled from the CSIRO Mk3.6.0 GCM. The precipitation and maximum and minimum temperature data are available on a daily basis for the period 1983-2005 to represent the baseline period and for 2030-2050 to represent the future period. Two scenarios for the radiative forcing of future greenhouse gas emissions were applied, namely Representative Concentration Pathway (RCP) 4.5 and 8.5 to represent low emission and high emission scenarios, respectively. In addition to the RCM dataset, this study also used six additional GCMs from different sources in order to include the effect of climate model uncertainty. This study used GCMs rather than other RCMs to include climate model uncertainty because other RCMs are not available for the study catchment. Table 4.1 shows the characteristics of each GCM used in this study. These GCMs were selected after comparison of the mean annual precipitation in the study catchment from 26 GCMs listed in CMIP5, which showed significantly different changes in precipitation (even in the direction of change) between the baseline and future periods resulting in a large uncertainty band in future climate variability.

4.3. Methods

4.3.1. Land use change model

Future land use distributions in the study catchment are based on two land use scenarios namely a business-as-usual (BAU) scenario and a controlled (CON) scenario. The BAU scenario represents a future situation where no measures are taken to control land use change in the study catchment, whereas the CON scenario represents idealized land use conditions that follow the spatial planning regulation.

Model	Resolution	Country	Signs*	References
	(Lon x Lat)	origin	•	
GFDL-ESM2M	$2.5^{\circ} \times 2.0^{\circ}$	USA		Dunne et al. (2012)
CanESM2	$2.81^{\circ} \times 2.79^{\circ}$	Canada		Arora et al. (2011)
CSIRO-Mk3.6.0	1.875 ⁰ × 1.86 ⁰	Australia	-	Collier et al. (2011)
MPI-ESM-LR	$1.875^{\circ} \times 1.85^{\circ}$	Germany	+	Giorgetta et al. (2013)
MIROC5	1.41 [°] × 1.39 [°]	Japan	+ +	Watanabe et al. (2010)
NorESM1-M	$2.5^{\circ} \times 1.875^{\circ}$	Norway	+ + +	Bentsen et al. (2013)

Table 4.1. List of GCMs used and their characteristics.

*Differences between baseline and future conditions, where the (-) sign indicates that the mean annual precipitation in the future period is smaller than in the baseline period and the (+) sign shows that the mean annual precipitation in the future period is larger than in the baseline period

4.3.1.1. Business as usual (BAU) scenario

Attempts were made to ensure that future land use in the study catchment under the BAU scenario is in accordance with the ongoing trends of land use change. An integration of a Markov Chain and a Cellular Automata model with Multi Criteria Evaluation (MCE) was used to predict land use changes in the catchment for the future period (i.e. 2030-2050). A combination between a Markov Chain and a Cellular Automata model, later called CA-Markov, has been widely used to simulate land use changes throughout the world (Myint & Wang, 2006; Hyandye & Martz, 2017). Compared to other models with a similar aim (e.g. GEOMOD, CLUE), the CA-Markov model has a high ability to simulate multiple land use covers and complex patterns with less data and computational efforts (Eastman, 2012). Along with CA-Markov, a multi criteria decision-making (MCE) technique was used to support the decision processes of land allocations using different criteria of land use suitability (Behera et al., 2012). MCE uses factors and constraints for each land use category. Factors indicate the relative suitability of a specific land use type that is generally based on a measured dataset (e.g. slope gradient, elevation, and road distances), whereas constraints are used to exclude certain areas from consideration (e.g. protected area and water bodies).

Factors and constraints of each land use type were selected based on the available spatial data (see Table 4.2). In order to make factors and constraints spatially comparable, the selected factors and constraints were standardized using fuzzy membership functions with a range between 0-1, where a value closer to 1 indicates a stronger membership.

Land use class	Factors	Membership functions	Control points	Constraints
Dryland Slope gradient		MIS	MIS a=0, b=15, c=25, d=45	
	Distance from road	MDL	a=0, b=maximum	 Water body
				 Existing settlement
Forest	Slope gradient	MIL	a=25, b=maximum	 Existing settlement
	Distance from road	MIL	a=500, b=maximum	 Existing paddy field
	Elevation (DEM)	MIL	a=1500, b=maximum	 Water body
Mixed garden	Slope gradient	MIS	a=8, b=25, c=25, d=45	 Protected area
	Distance from road	MDL	a=100, b=maximum	 Water body
	Distance from existing settlement	MDL	a=100, b=maximum	 Existing settlement
Paddy field	Slope gradient	MIS	a=0, b=15, c=25, d=45	 Protected area
	Distance from road	MDL	a=0, b=maximum	 Water body
	Distance from existing settlement	MDL	a=0, b=maximum	 Existing settlement
Settlement	Distance from road	MDL	a=0, b=maximum	 Protected area
	Distance from existing settlement	MDL	a=0, b=maximum	 Water body
	Slope gradient	MDL	a=0, b=45	
	Distance from urban facilities	MDL	a=0, b=maximum	
Shrub	Slope gradient	MIL	a=15, d=maximum	 Protected area
	Distance from road	MIL	a=500, b=maximum	 Water body
	Distance from existing settlement	MIL	a=500, b=maximum	 Existing settlement
Water body	Existing water body	-	-	-

Table 4.2. Factors, membership functions, control points and constraints of the different land use classes.

MIL = monotonically increasing linear function, MDL = monotonically decreasing linear function, MIS = monotonically increasing symmetric function

Three types of fuzzy membership functions were used, namely a monotonically increasing linear function, a monotonically decreasing linear function and a monotonically increasing symmetric function. Subsequently, four fuzzy control points were determined in which the first point marks the location where the membership function begins to rise above 0, the second point indicates where it reaches 1, the third point indicates the location where the membership function drops again below 1, and the fourth point marks the point where it returns to 0. The four control points of fuzzy membership function for each land use class used in this study were adapted from Hyandye & Martz (2017). However, some changes were made for this study considering the local conditions (e.g. agriculture area was divided into paddy field and dry land farm) and data availability (e.g. population density was not included as a factor).

In this study, the Markovian transition area matrix was generated using two recent land use maps for the years 2006 and 2013. However, a boundary condition was applied for the settlement area by determining the settlement area in 2050 based on a linear relation between the changes in settlement area estimated from historical land use maps (Marhaento et al., 2017a) and population size in the same years (see Figure 4.2a). Population in 2050 was predicted using a second order polynomial function that can represent a decrease in the population growth in the future, following the prediction from the Indonesia Statistical Bureau (BPS, 2013) (Figure 4.2b). The results show that the settlement area in the study catchment in 2050 will be 50.2% of the study area.



Figure 4.2. A linear relation between the settlement area (in %) and population size (a). A second order polynomial function of the population growth in the Samin catchment (b).

This boundary condition was applied in the simulation through a modification of the time lags of the model simulation until the model closely predicted a settlement area in 2050 of about 50% of the study area. The predicted annual land use distributions from 2030 to 2050 resulting from the fitted model were used as land use inputs in the SWAT simulations under the BAU scenario. For this study, spatial data preparation and land use simulations were executed using IDRISI Selva v.17 software (Eastman, 2012).

4.3.1.2. Controlled scenario

In this scenario (i.e. CON scenario), future land use in the study catchment was assumed to follow the spatial planning map. A pre-processing analysis has been carried out in order to convert a printed map of the spatial planning map into a digital map. Furthermore, the classification of the spatial planning map was changed to be comparable with the land use classification from Marhaento et al. (2017a). Figure 4.3 shows the spatial planning map of the study catchment with reclassified land use types. According to the spatial planning map, in the year 2029, agriculture area (i.e. paddy field and dryland farm) is the dominant land use type in the study catchment, covering 55% of the study area, followed by forest area (i.e. evergreen forest and mixed garden, 31%) and settlements (14%). Forest area is dominant in the upstream area, whereas agriculture area and settlements are dominant in the downstream area. This study used the predicted land use distribution in 2029 from the spatial planning map and the land use map for the year 2013 as inputs in the CA-Markov model in order to simulate annual land use distributions in the future period (i.e. 2030-2050). Output of the model was used as land use input in the SWAT simulations under the CON scenario.



Figure 4.3. Predicted land use map of the Samin catchment in the year 2029 according to the spatial planning map from the Central Java Provincial Government.

4.3.2. Climate change model

GCM and RCM outputs are generally biased, which hampers the direct use of GCM or RCM data to assess the impact of climate change on hydrological processes (Teutschbein and Seibert, 2012). Thus, there is a need to correct these outputs before they can be used for regional impact studies. In this study, different bias correction methods were used to correct RCM and GCM output. For the RCM data, the distribution mapping method and the variance scaling method were used to correct biases in precipitation and maximum and minimum temperature, respectively. These methods were selected after performing an accuracy assessment based on a split sample test (Klemeš, 1986) for different bias correction methods (i.e. linear scaling, power transformation, distribution mapping). It was found that these methods outperformed other methods based on their coefficient of determination (R²) against the observed precipitation and maximum and minimum temperature. For precipitation, the distribution mapping method resulted in a R^2 of 0.77, whereas the linear scaling method and power transformation method resulted in a R² of 0.53 and 0.56, respectively. For the maximum and minimum temperature, the variance scaling method resulted in a R^2 of 0.62, whereas the linear scaling method and distribution method resulted in a R² of 0.51 and 0.48, respectively. However, it should be noted that all bias correction methods improved the raw RCM-simulated precipitation and maximum and minimum temperature. For the six additional GCMs, a delta change method was applied to correct biases of the GCM-simulated precipitation and maximum and minimum temperature. Rather than using the GCM simulations of future conditions directly, the delta change method uses the differences between GCMsimulated historic and future conditions (anomalies) for a perturbation of observed data. For a detailed description of each method, references are made to Teutschbein & Seibert (2012) and Fang et al. (2015).

4.3.3. Hydrological model

The Soil and Water Assessment Tool (SWAT) model (Arnold et al., 1998) was used to simulate hydrological processes in the study catchment. The SWAT model is a physically based semi-distributed model that divides a catchment into sub-catchments and then further into hydrological response units (HRUs) for which a land-phase water balance is calculated (Neitsch et al., 2011). Runoff from each HRU (i.e. combinations of land use, soil and slope) is aggregated at sub-catchment level and then routed to the main channel for which the catchment water balance is calculated. In a previous study, Marhaento et al. (2017a) showed the suitability of SWAT to attribute changes in the water balance to land use change in the Samin catchment. The SWAT model set up in the

present study used the same settings as in Marhaento et al. (2017a). Since details of the model setup and parameterization are available in the previous study, this study only presents a brief summary of the model calibration results.

The calibration period is 1990–1995 assuming that this period is a reference period when land use change and climate change had small impacts on hydrological processes (Marhaento et al., 2017a). Six SWAT parameters were calibrated on a monthly basis. The initial parameter ranges were determined based on minimum and maximum values allowed in SWAT. Four iterations were performed where each iteration consisted of 1,000 simulations with narrowed parameter ranges in the subsequent calibration rounds. The Nash Sutcliffe Efficiency (NSE) was used as objective function, similarly as in for instance Setegn et al. (2011) and Zhang et al. (2015). The results of the model calibration show that the simulated mean monthly discharge in the calibration period agrees well with the observed records with a NSE value of 0.78. In the validation period (1996–2013), the NSE model performance of the calibrated SWAT model is 0.7 (Marhaento et al., 2017a).

4.3.4. Future hydrological responses to land use and climate change scenarios

Future hydrological processes in the study catchment were simulated using the validated SWAT model, using inputs from the bias-corrected RCM and GCMs for the period 2030-2050 and the predicted land use distributions. For the baseline conditions, this study used the output of the SWAT simulations forced by the bias-corrected RCM data and the observed data, where both data sets cover the period 1983-2005. The baseline condition simulated with the bias-corrected RCM data was used as a baseline condition for the future simulation forced by the RCM dataset, whereas the baseline condition simulated by the observed data was used as a baseline condition for the future simulation forced by the GCM dataset. In addition, two land use distributions for the years 1994 and 2000 were used in the simulation to represent land use conditions in the baseline period. The Land Use Update (LUP) tool in ArcSWAT (Marhaento et al., 2017a) was used to incorporate land use change in the SWAT simulations, whereby land use was initially assumed as the land use reported for 1994, and updated by the land use for 2000 when the simulation date entered 1st January 2000. The LUP tool was also used to incorporate land use change in the simulations for future period (i.e. 2030-2050). Table 4.3 shows the scenario simulations executed in this study including the assessment of the effects of land use change only (LUC), climate change only (CC) and the combined land use and climate change (LUC+CC) scenario on hydrological processes. For each

scenario, several hydrological components namely precipitation (P), potential evapotranspiration (PET), streamflow (Q), actual evapotranspiration (ET), surface runoff (Q_s), lateral flow (Q_l) and base flow (Q_b) were calculated and compared to the baseline conditions.

Table 4.3. Future land use and climate change scenarios as inputs for the SWAT model.

Scenarios		Land use change			
		NLUC	BAU	CON	
Climate change	NCC	Baseline	BAUNC ¹	CONNC ¹	
	RCP 4.5	NC4.5 ²	BAU4.5 ³	CON4.5 ³	
	RCP 8.5	NC8.5 ²	BAU8.5 ³	CON8.5 ³	

NLUC = No land use change, NCC = No climate change, BAU = Business as usual land use scenario, CON = Controlled land use scenario, RCP 4.5 = Climate change scenario under Representative Concentration Pathway 4.5, RCP 8.5 = Climate change scenario under Representative Concentration Pathway 8.5. ¹LUC scenarios, ²CC scenarios, ³LUC+CC scenarios

4.4. Results

4.4.1. Land use change

As shown in Table 4.4, land use in the study catchment in 2050 under the BAU scenario is predominantly settlements (50.2%), followed by paddy field (22.2%), dryland farm (18.2%), mixed garden (4.9%), evergreen forest (2.3%), shrub (1.9%) and water body (0.3%). In comparison to the year 2000, forest area (i.e. evergreen forest and mixed garden) and shrub have decreased by 35% and 2.4%, respectively, while agriculture area (i.e. paddy field and dryland farm) and settlements have increased by 3.5% and 33.9%, respectively. The settlement area significantly increased in the down- and middle stream area converting mixed garden and paddy field area into impervious area. In the upstream area, the dryland farm has mainly converted mixed garden and shrub area, as shown in Figure 4.4b.

Under the CON scenario, land use in 2050 is predominantly paddy field (34%), followed by dryland farm (18.1%), settlements (17.3%), mixed garden (17.1%), evergreen forest (13.2%) and water body (0.3%). In comparison to the year 2000, mixed garden and shrub have decreased by 22.1% and 4.3%, respectively, while agriculture area (i.e. paddy field and dryland farm), evergreen forest and settlements have increased by 15.2%, 10.2% and 1%, respectively. Figure 4.4c shows that under the CON scenario, forests (i.e. evergreen forest and mixed garden) are located in the upstream area, while agriculture area (i.e.

dryland farm and paddy field) and settlement are in the down- and middle stream area. It should be noted that within this scenario the evergreen forest area will increase but at the other hand the mixed garden area will decrease resulting in a net decrease in the forest area by 11.9%.

Land use class		Reference	Future (year 2050)				
		(year 2000)	BAU Scenario		CON Scenario		
		LU 2000	LU 2050	Δ	LU 2050	Δ	
Forest area		42.2	7.2	-35	30.3	-11.9	
a.	Evergreen forest	3.0	2.3	-0.7	13.2	10.2	
b.	Mixed garden	39.2	4.9	-34.4	17.1	-22.1	
Agriculture area		36.9	40.4	3.5	52.1	15.2	
a.	Paddy field	28.2	22.2	-6	34	5.8	
b.	Dryland farm	8.7	18.2	9.5	18.1	9.4	
3. 5	Settlement	16.3	50.2	33.9	17.3	1	
4. 5	Shrub	4.3	1.9	-2.4	0	-4.3	
5. Water body		0.3	0.3	0	0.3	0	

Table 4.4. Land use distribution (in %) in the years 2000 and 2050 for different scenarios.



Figure 4.4. Land use cover of the Samin catchment in 2000 (a) and 2050 under business as usual (BAU) scenario (b) and controlled (CON) scenario (c).

4.4.2. Climate change

Figure 4.5 shows the changes in mean annual precipitation from the RCM and GCMs under RCP 4.5 and RCP 8.5 in the future period (2030-2050) relative to the baseline period (1983-2005). Based on the RCM, the mean annual precipitation in the study catchment is predicted to decrease by 412 mm (-20%) under RCP 4.5 and 505 mm (-25%) under RCP 8.5. The direction of change in the mean annual precipitation for the RCM is similar to the directions of change from three GCMs (i.e. GFDL-ESM2M, CanESM2 and CSIRO-Mk3.6.0), which predicted a decrease in mean annual precipitation ranging from 8 mm (-0.4%) to 193 mm (-18%) under RCP 4.5 and from 22 mm (-1%) to 343 mm (-17%) under RCP 8.5. On the contrary, three GCMs, namely MIROC5, MPI-ESM-LR and NorESM1-M, show an opposite direction of change in the mean annual precipitation over the study area. These GCMs predict an increase in mean annual precipitation in the future ranging from 91 mm (+5%) to 384 mm (+19%) under RCP 4.5 and from 87 mm (+4%) to 287 mm (+15%) under RCP 8.5. Although the RCM-simulated precipitation was downscaled from the CSIRO Mk.3.6.0 GCM, the magnitude of change of the two is different, in particular under the RCP 4.5 scenario. This is probably due to systematic biases in the RCM, which makes it independent from its driving datasets because of its physics parameterization (Murphy et al., 2004), and application of the biascorrection method (Themeßl et al., 2012).



Figure 4.5. Changes in mean annual precipitation in the Samin catchment in the future period (2030-2050) relative to the baseline period (1983–2005) under RCP 4.5 and RCP 8.5 scenarios.

While different climate models show inconsistent directions of change for the future precipitation, all climate models predict a similar direction of change for the maximum and minimum temperature, although with slightly different magnitudes. For the RCM, the mean annual maximum and minimum temperatures increase for both scenarios, by $+1.0^{\circ}$ C and $+1.3^{\circ}$ C, respectively, under RCP 4.5, and by $+0.7^{\circ}$ C and $+1.4^{\circ}$ C, respectively, under RCP 8.5. All GCMs predicted an increase of about $+1^{\circ}$ C in the mean annual maximum and minimum temperature in the future period compared to the baseline period under both RCP 4.5 and RCP 8.5 scenarios.

4.4.3. Future hydrological response

4.4.3.1. Effects of land use change

In comparison to the baseline conditions (1983–2005), simulations under land use change scenarios indicate a decrease in the mean annual ET and an increase in the mean annual Q and Q_s (Figure 4.6a). However, the directions of change were different for Q₁ and Q_b. Under BAU, Q₁ increased and Q_b decreased while under CON, Q₁ decreased and Q_b increased. It was found that under BAU the mean annual ET decreases by 15%, which is larger than the decrease of 7.4% under CON. A decrease in the mean annual ET under both scenarios is compensated by an increase in the mean annual Q with the same magnitude. More pronounced changes occur in the fraction of Q that becomes Q_s, Q₁ and Q_b. Under BAU, the mean annual Q_s and Q₁ increase by 40% and 20%, respectively, at the expense of Q_b, which decreases by 3%. Under CON, the mean annual Q_s increases by 16% and the mean annual Q_b by 6%, while the mean annual Q₁ was relatively constant.



Figure 4.6 (a) Change in the mean annual water balance components under different land use change scenarios compared to the baseline period. (b) Mean monthly streamflow under different land use change scenarios.

At a monthly scale (Figure 4.6b), the simulated Q under both land use change scenarios is higher than in the baseline period during the wet season and similar to the baseline period during the dry season. The mean monthly Q under the BAU and CON scenarios reaches its peak in February, with a value of 175 mm (+41%) and 149 mm (+20%), respectively.

4.4.3.2. Effects of climate change

There is a large uncertainty in the future mean annual P for both RCP 4.5 and RCP 8.5, which causes a large variation in the predicted water balance components (Figure 4.7). Projected changes in the mean annual P under different RCP scenarios range from -20% to +19% under RCP 4.5 and from -25% to +15% under RCP 8.5. Together with an increase in the simulated PET of about 3%, changes in the mean annual P from both RCP scenarios cause a significant change in the mean annual Q as well as in the fractions of flow that become Q_s, Q_l and Q_b, while changes in ET are relatively minor. Under RCP 4.5, projections of the mean annual Q and ET range from -35% to +43% and from a -7% to +5%, respectively. Pronounced changes can be observed for the mean annual Q_s (ranging from -33% to +50%), Q_l (ranging from -32% to +38%) and Q_b (ranging from -38% to +39%). Under RCP 8.5, projections of the mean annual Q and ET range from -47% to +32% and from -10% to +5%, respectively, whereas changes in the mean annual Q_s , Q_l and Q_h range from -44% to +36%, from -41% to +29%, and -56% to +28%, respectively. The large ranges in the projections for the mean annual water balance components likely result from the differences in the predicted P from the different climate models. It should be noted that the RCM downscaled from a CSIRO Mk.3.6.0 GCM projected a (very) significant decrease in the mean annual P for both scenarios, resulting in pronounced changes in the mean annual Q, Q_s, Q_l and Q_b.

At a monthly scale, most climate models under RCP 4.5 and RCP 8.5 project a decrease in ET during the dry season and an increase during the wet season (Figure 4.8). A large uncertainty is found for the dry season, for which some climate models under RCP 8.5 project a significant decrease in the mean monthly ET, in particular for the simulation using the RCM data. For Q, a large uncertainty is found during the wet season, because different climate models project different directions of change, in particular under RCP 4.5. Under RCP 8.5, the direction of change is evident, with most climate models projecting a decrease of Q from April to December.



Figure 4.7. Changes in the water balance components (in %) compared to the baseline period under RCP 4.5 and RCP 8.5 scenarios for different climate models.



Figure 4.8. Mean monthly ET and Q from different climate models under RCP 4.5 and RCP 8.5 scenarios.

4.4.3.3. Combined effects of land use change and climate change

When land use and climate change scenarios are combined, the effects on the water balance components are pronounced as shown in Figure 4.9. Mean annual ET decreases, with the highest decrease under the BAU8.5 scenario (ranging from -10% to -25%) and the lowest decrease under the CON4.5 scenario (-1% to -14%). The direction of changes in the mean annual ET under combined climate and land use change scenarios are similar with the ET changes found when consider land use change only scenario. The mean annual Q is mostly predicted to increase under the BAU4.5 scenario (from -7% to +64%), whereby only the simulation with RCM data predicted a decrease in Q. Under the BAU8.5 and CON4.5 scenarios, the changes in the mean annual Q range from -18% to +53% and -19% to 52%, respectively. The largest uncertainty in change in the mean annual Q is found under the CON8.5 scenario, in which mean annual Q will change by -30% to +42%. An increase in the mean annual Q is observed under the BAU4.5 scenario, although some climate models predict a decrease in the mean annual precipitation. Changes in the mean annual Q due to land use change are amplified under wetting climate scenarios (prediction of an increase in the future precipitation from the climate models), where the mean annual Q increases by about 64% under scenario BAU4.5, which is about 40% more than under the land use change only scenario.

Similarly, there is a significant increase in the mean annual Q_s in particular under the BAU4.5 and BAU 8.5 scenarios, although the RCM and some GCMs predict a decrease in the mean annual precipitation. The mean annual Q_s is predicted to increase by +21% to +102% and by +2% to +91% under the BAU4.5 and BAU8.5 scenarios, respectively. Changes in the mean annual Q_s are pronounced for wetting climate scenarios, where the mean annual Q_s increases by about 100% compared to the baseline conditions, which is 60% more than under the land use change only scenario. For Q_b , it is observed that scenarios show increases as well as decreases in the mean annual Q_b , resulting in a large uncertainty. The directions of change in the mean annual Q_b under different scenarios are similar to those under the climate change only scenario. The CON4.5 scenario predicts the largest change in mean annual Q_b with a range from -30% to +46%, subsequently followed by the CON8.5, BAU4.5 and BAU8.5 scenarios.



Figure 4.9. Predicted mean annual water balance components for different combinations of climate change and land use change scenarios.

At a monthly scale, most of the scenarios project an increase in the mean monthly Q from January to March, where changes in the mean monthly Q range from +60% to +120% compared to the baseline period (Figure 4.10). During the dry season, a large uncertainty is found for all scenarios, which predict decreases as well as increases in the mean monthly Q. At the start of the wet season (i.e. November), most scenarios predict a decrease in the mean monthly Q, whereby the largest decrease is found under the CON85 scenario (ranging from -84% to +14%), subsequently followed by BAU85, CON45 and BAU45.



Figure 4.10. Mean monthly streamflow for different combinations of land use change and climate change scenarios

4.5. Discussion

In the Samin catchment, changes in land use are mainly controlled by internal factors, such as expansions of agriculture and settlements, whereas changes in climate variability are considered as externally driven. Both play a significant role in changing hydrological processes of the catchment. Using a modelling approach, this study assesses individual and combined effects of future land use and climate change on the water balance components in the Samin catchment.

4.5.1. Effects of future land use change

Findings show that when future land use changes under a business as usual (BAU) scenario, mean annual evapotranspiration will decrease, resulting in an increase in the mean annual streamflow. The mean annual surface runoff and

lateral flow are projected to increase almost twofold, but base flow will decrease. A massive land use conversion from vegetated area into settlement area as suggested by the BAU scenario will significantly reduce canopy interception and soil infiltration capacity, resulting a large fraction of precipitation being transformed into surface runoff (Bruijnzeel, 2004; Valentin et al., 2008; Marhaento et al., 2017a). As a result, under BAU, the Samin catchment will face an increasing risk for disastrous events such as flash floods, landslides and severe soil erosion. Those risks can be potentially reduced by introducing and enforcing a land use planning regulation. It was simulated that the allocation of forest area to the upstream part of the catchment and agriculture to the midstream part, while maintaining the settlement area, as suggested by the CON scenario, will significantly reduce the mean annual surface runoff and lateral flow, while the mean annual base flow will increase. A large fraction of vegetated area (i.e. evergreen forest) may lead to an increase in the water storage capacity of the soil due to greater root penetration and biotic activity in the upper soil layers, resulting in a larger infiltration rate and ground water recharge (Bruijnzeel 1989; 2004; Guevara-Escobar et al., 2007). With more water infiltrated and stored into the soil, a more balanced water distribution between wet and dry seasons can be expected. The directions of change in the water balance components by land use change in this study are in line with other studies in the tropics from Bruijnzeel (1989), Valentin et al. (2008), Remondi et al. (2016) and Marhaento et al. (2017a).

4.5.2. Effects of future climate change

When considering only climate change, findings show that changes in the mean annual precipitation may have large impacts on the water availability of the Samin catchment. An increase (or decrease) in the mean annual precipitation may result in a large increase (or decrease) in the streamflow and in the fractions of flow that become surface runoff, lateral flow and base flow, while changes in ET were relatively minor. However, small increases in precipitation are likely to have little impacts on the water balance. More significant impacts on the water balance can be observed if the changes in the mean annual precipitation are larger than 10%, where the impacts on streamflow and on the fractions of flow that become surface runoff, lateral flow and base flow can be twice the changes in precipitation. These findings are similar to the results of Mango et al. (2011) and Liu et al. (2011). The rising temperature causes an increase in the potential evapotranspiration, which may affect the mean annual ET due to an increase in the available energy (Budyko, 1974). However, this study finds that changes in annual ET are likely attributed to changes in annual precipitation, where the variations of ET follow the variations of precipitation. Thus, an increase in ET is found during the wet season while a significant decrease in ET is found during the dry season. These findings are in line with Budyko (1974), who argues that changes in evapotranspiration are determined by the balance between precipitation and evaporative demands.

4.5.3. Combined effects of future land use change and climate change

The findings show that both land use change and climate change contribute to changes in the water balance components, but each driver has a specific contribution to the water balance alteration. Under combinations of land use and climate change scenarios, changes in the annual evapotranspiration are likely attributed to land use change, while changes in the annual base flow are likely attributed to climate change. It should be noted that this study used two completely different land use scenarios, where the BAU scenario represents deforestation and the CON scenario (partly) reforestation. Thus, changes in annual evapotranspiration under different land use change scenarios can be clearly observed, since trees are generally known to have higher evapotranspiration rates than other land uses (Bruijnzeel, 1989; 2004; Marhaento et al. 2017a). The presence of forests can increase the annual evapotranspiration from both canopy interception and plant transpiration, so that under the BAU scenario there will be a significant decrease in the annual evapotranspiration. However, one cannot neglect the role of climate variability to control changes in the annual evapotranspiration, since evapotranspiration is largely influenced by precipitation (Budyko, 1974; Liu et al., 2013). At a monthly scale, evapotranspiration is close to its potential value during the wet season, since precipitation supplies sufficient water, whereas in the dry season, the evapotranspiration capacity is mainly determined by the antecedent soil moisture and influenced by different land cover types (Liu et al., 2013).

The magnitude of changes in annual base flow is mainly determined by climate change (i.e. changes in precipitation), whereby the directions of change in annual Q_b under different scenarios are similar to those under the climate change only scenario. This study finds a small effect of land use change on the changes in the annual base flow, whereby a larger base flow is found under the CON scenario. However, the magnitude of changes in base flow under different land use scenarios was smaller than expected. According to llstedt et al. (2007), the presence of forests can result in more base flow due to an enhanced groundwater recharge. Apparently, in this catchment, the effects of land use change on the base flow are offset by climate change. Thus, more research is needed to assess the attribution of changes in the base flow to land use change and climate change.

Combined land use change and climate change have more pronounced effect on the streamflow and surface runoff. A projected increase in precipitation accompanied by the deforestation scenario (BAU scenario) will have more significant impacts on streamflow and surface runoff than land use change or climate change acting alone (Legesse et al., 2003; Hejazi & Moglen, 2008; Khoi & Suetsugi, 2014). Moreover, at a monthly scale, the streamflow originating from surface runoff significantly increases during the wet season and decreases during the dry season, indicating that more extreme events (i.e. droughts and floods) can potentially occur in the future. Under the land use planning scenario (CON scenario), the effects can be reduced, but only by less than 20%. Thus, more measures (e.g. soil conservation) are required in addition to land use planning in order to enhance infiltration and aquifer recharge and subsequently reduce risks due to land use and climate change impacts.

4.5.4. Limitations and uncertainties

This study couples a land use change model, climate change model and hydrological model, whereby each model can be a source of uncertainty affecting the results. For the land use change model, an integration of the CA-Markov and MCE techniques was used to project future land use distributions. However, this method predicts land use in the future based on extrapolation of trends, while future trends obviously are uncertain due to uncertainties in e.g. future population growth. For this reason, this study used a control condition in the simulations to constrain the predictions of the future land use distributions. For the climate change model, large model uncertainties occur due to model choices, where different climate models result in different directions of change in the projected precipitation. For this reason, this study employs six additional GCMs that were selected based on different directions of change in future precipitation to include climate model uncertainty. However, uncertainties from the climate models can be larger due to other sources of uncertainty, for instance from the model structure (e.g. choices of spatial resolution, the set of processes included in the model, basic assumptions on parameterization) and application of statistical downscaling methods (Murphy et al., 2004). For the hydrological model, sources of uncertainty can be present due to the model structure (e.g. model assumptions, equations) as well as data inputs (e.g. lack of relevant spatial and temporal variability of data on precipitation, soils, land uses and topography). Moreover, the scenario settings used in this study follow a singlefactor fixed method, which does not take into account atmospheric feedbacks of land surface processes (Blöschl et al., 2007).

4.6. Conclusions

This study assessed the separate and combined impacts of future changes in land use and climate on the water balance components in the Samin catchment. Results based on the SWAT modelling approach show that both land use and climate change can result in changes in the water balance components, but that more pronounced changes will occur if the drivers are mutually considered, this in particular for the mean annual stream flow and surface runoff. When combining the RCP 4.5 climate scenario and BAU land use scenario, the mean annual streamflow and surface runoff are expected to change by -7% to +64% and +21% to +102%, respectively, which is 40% and 60% more than when land use change is acting alone. When the BAU land use scenario is replaced by the CON land use scenario, the mean annual streamflow and surface runoff reduce by up to 10% and 30%, respectively, while the mean annual base flow and evapotranspiration increase by about 8% and 10%, respectively. The findings show that land use planning can be one of the measures to reduce future waterrelated risks in the Samin catchment. However, it remains a challenge to accurately predict the hydrological changes due to land use and climate change since there are various uncertainties, in particular associated with future climate change scenarios.

Chapter 5

Sensitivity of streamflow characteristics to different spatial land use configurations in a tropical catchment⁴

Abstract

This paper analyses the impact of the spatial configuration of specific forms of land use, like forest and settlements, on streamflow in the Samin catchment, Indonesia. Land use distributions in the past have been reconstructed based on satellite images and used to assess land use pattern characteristics for the years 1982, 1994, 2000, 2006 and 2013. Hydrological processes of the study catchment are simulated using a validated Soil and Water Assessment Tool (SWAT), taking land use distribution for the mentioned years as inputs. Next, a correlation analysis between changes in land use pattern characteristics and simulated streamflow characteristics was carried out. A land use scenario analysis was done to assess the sensitivity of streamflow characteristics to different future land use patterns. The results show that changes in the streamflow characteristics of the study catchment can be attributed to changes in the percentages of different land use types and changes in the physical connectivity between patches of similar land use types. When the relative presence of different land use types is fixed but the physical connectivity of patches is changed, it is found that an increase in the settlement connectivity can result in an increase in the ratio of surface runoff to streamflow and a decrease in the ratio of dry-season streamflow to wet-season streamflow, and vice versa, while changes in the forest connectivity have less impact on streamflow characteristics. The results suggest that land use pattern management can be an important component in water management.

⁴ This chapter is based on a paper that has been submitted as: Marhaento, H., Booij, M. J., Rientjes, T. H. M., & Hoekstra, A. Y. Sensitivity of streamflow characteristics to different spatial land use configurations in a tropical catchment.

5.1. Introduction

In the last decades, the impacts of land use changes on catchment hydrology have received a lot of attention. Numerous studies argue that different types of land use have different water use and water storage characteristics, thus different land use distributions may result in different distributions of water in space and time (Bruijnzeel, 1989; 2004; Sahin & Hall, 1996; Brown et al., 2005; Romanowicz & Booij, 2011; Gumindoga et al., 2014a; Zhang et al., 2016; Marhaento et al., 2017a). Studies in tropical regions have shown that changes from vegetated areas into settlements may significantly reduce canopy interception and soil infiltration capacity, resulting in a large fraction of precipitation being transformed into surface runoff (Bruijnzeel, 2004; Valentin et al., 2008; Marhaento et al., 2017a). In addition, the spatial distribution of different land uses may affect the hydrological response of catchments, for example the velocity and volume of surface flow entering a stream (Wheater & Evans, 2009; Su et al., 2014). Hence, land use management can be used as a measure in water management practice for flood prevention and drought mitigation (Wheater & Evans, 2009 and Zhang et al., 2013).

Given the rapid growth of populations and associated demand for settlements and agricultural lands, land use policy can be a relevant factor in water management (Carter et al., 2005). Increased demands for settlements and farming areas can be in conflict with an increased demand for water resources. For instance, agricultural intensification and settlement development can lead to a reduction in soil infiltration rates and base flow and an increase of surface runoff and soil erosion. Therefore, future land allocation and land use management should be designed based on a proper understanding of how land use changes may affect catchment hydrology (see Carter et al., 2005).

Most studies assessing the effects of land use change on hydrological processes focused on the impact of changes in the relative presence of different land use types. Less attention has been given to the impacts of changes in spatial land use configurations (e.g. shape and connectivity of land use types). The impact of the spatial land use pattern on water resources, however, is relevant when selecting a particular land use management strategy (Azevedo et al., 2005; O'Connell et al., 2007; Bakker et al., 2008). Relevant questions are, for example: what is the effect of choosing different locations for certain new land uses (e.g. high vs. low elevations), and what is the effect of different spatial configurations (e.g. small-scattered vs. large-clustered) of forest planting and harvesting strategies? A better understanding will give insight in the potential effects of alternative land use decisions on water availability at catchment scale (Lin et al., 2007). Figure 5.1 depicts different spatial land use patterns which may result in different hydrological responses.



Figure 5.1. Six grids of 6 x 6 cells showing differences in abundance of a certain land use type (e.g. settlements) and differences in spatial configuration of this land use type in the landscape. Grids 1 to 3 show an increase in abundance, from 3/36 to 6/36 to 9/36 of the total area, respectively. Grids 4 to 6 show changes in the spatial patterns (e.g. shape and connectivity) while the fraction of patches to the total area is maintained at 9/36.

The impact of land use patterns on hydrological processes is often studied by means of a paired catchment approach, in which land use in the control catchment is held constant while land use in the treatment catchment, having similar physical conditions, is changed. Measurements then focus on the differences in hydrological response between the control and treatment catchment (Fohrer et al., 2005). Survatmojo et al. (2011) used this approach to assess the impacts of different forest harvesting strategies (i.e. selective logging) on the streamflow characteristics in Central Kalimantan, Indonesia, and found that applications of different forest harvesting strategies resulted in different hydrological responses (i.e. peak flows). Sahin and Hall (1996) and Brown et al. (2005) reviewed the results of numerous paired catchment studies worldwide and found that changes in land use through deforestation, afforestation, regrowth and forest conversion can affect annual streamflow, which is likely to increase with the percentage of forest removed. Although a paired catchment study can provide relevant information of changes in hydrological processes due to changes in land use, this method is only applicable to small catchments (<25 km²), for which uncertainties due to spatial heterogeneity are relatively small (Fohrer et al., 2005; Brown et al., 2005).

For larger catchments (>100 km²), a modelling approach is typically used. Zhang et al. (2013) generated various hypothetical land use patterns using a land use change model and used these as input for a grid-based hydrological model to assess the impacts of land use patterns on streamflow in the Yong Ding catchment (300 km²), China. They found that land use fragmentation may affect streamflow characteristics at different spatial resolutions. At fine scale (30x30 m²), fragmented grassland positively correlated with peak flow and total streamflow, while at coarse scale $(1,200 \times 1,200 \text{ m}^2)$, fragmented forest positively correlated with peak flow and total streamflow. Li and Zhou (2015) assessed the correlation between various land use pattern characteristics and hydrological variables (i.e. streamflow and sediment yield) in the Yanhe catchment (7,725 km²), China, and found that changes in the land use patterns significantly changed the sediment yield but did not significantly change the streamflow. Bormann et al. (2009) used different hydrological models (i.e. WASIM, TOPLATS and SWAT) to assess the sensitivity of water balance components to changes in different land use patterns in the Dill catchment (693 km²), Germany, and found that land use redistribution slightly affected the water balance components. These studies show that the relationships between land use patterns and hydrological processes at the catchment scale are sometimes contradictory, requiring further study.

This paper aims to assess the impacts of land use patterns on streamflow characteristics of the Samin catchment (278 km²) in Java, Indonesia. Hydrological processes of the study catchment were simulated using the semidistributed physically based Soil and Water Assessment Tool (SWAT) (Arnold et al., 1998). In the first part of this study, land use patterns over the period 1982-2013 have been reconstructed. Landsat images from the years 1982, 1994, 2000, 2006 and 2013 were used to assess past land use patterns. Different landscape metrics were calculated using FRAGSTAT (McGarigal & Marks, 1995) to define spatial land use pattern characteristics for each year. In the second part of this study, correlations between land use metrics and streamflow characteristics were determined. Two streamflow characteristics, the ratio of surface runoff to streamflow (Q_{s}/Q) and the ratio of streamflow in the dry season to streamflow in the wet season (Q_{drv}/Q_{wet}) were simulated using a validated SWAT model (Marhaento et al., 2017a) with different land use distributions. For the third part of this study, hypothetical land use scenarios were developed in order to assess the sensitivity of streamflow characteristics to different spatial land use configurations. In this scenario analysis, land use planning for the study catchment available from the Central Java Provincial Government (2010) was used as a baseline scenario and its spatial configuration was changed while maintaining the percentage of each land use type. Finally, this study assessed the potential hydrological impacts of different land use patterns in the study catchment. The study area and data used are described in Section 5.2. Section 5.3 covers the methods used in this study. Section 5.4 presents the results, Section 5.5 discusses the key findings, and Section 5.6 draws the conclusions from this study.

5.2. Study area and data availability

5.2.1. Catchment description

The Samin river is a tributary of the Bengawan Solo river, the longest river in Java, Indonesia. The source of the river is the Lawu Mountain (3,175 m above sea level). The Samin catchment area is about 278 km², situated between 7.6⁰-7.7° southern latitude and 110.8°–111.2° eastern longitude. The mean elevation of the catchment is 380 m, the mean slope 19.8%, and the stream density around 2.2 km/km². The soil composition of the Samin catchment is predominantly Luvisols, a leafy humus soil that can be mainly found in the midstream and downstream areas, and Andosols, a volcanic soil that can be mainly found in the upstream area near the Lawu Mountain. These soils cover 57% and 43% of the study area, respectively. The study catchment has a tropical monsoon climate with a distinct wet season (November to April) and dry season (May to October), with January-March being the wettest period and July-September the driest period of the year. In the period 1990-2013, the mean daily temperature varied between 21.5 °C and 30.5 °C, the annual precipitation varied between 1,500 mm and 3,000 mm, the annual potential evapotranspiration varied between 1,400 mm and 1,700 mm, and the annual streamflow varied between 500 mm and 1,200 mm (Marhaento et al., 2017b). Figure 5.2 shows the Samin catchment with its topography and the location of hydro-meteorological gauges.

5.2.2. Data availability

Various datasets are available for modelling hydrological processes of the study catchment, such as a Digital Elevation Model (DEM), land use data, soil data and climate data. The DEM was generated from a contour map with a contour interval of 12.5 meters, which was made available by the Indonesia Geospatial Information Agency. Furthermore, the DEM was used to delineate the catchment and sub-catchment boundaries and to generate a slope map. Land use maps with a spatial resolution of 30 m for the years 1994, 2000, 2006 and 2013 were available for the study area from Marhaento et al. (2017a). An additional land cover image for September 11, 1982 was acquired from the Landsat satellite (USGS, 2017) to represent land use for the year 1982.



Figure 5.2. Location of the Samin catchment in Java, Indonesia, with locations of hydrometeorological gauges.

A soil map at 30-arc second spatial resolution was available from the Harmonized World Soil Database (FAO et al., 2012). A land use spatial planning map of the study area for the period 2009–2029 was available with a spatial scale of 1:500.000 from the Central Java Provincial Government (2010). Daily climate data were available from eleven precipitation stations and three meteorological (see Figure 5.2) stations for the period 1983–2013, provided by the Bengawan Solo River Basin office and the Adi Sumarmo Airport. The Bengawan Solo River Basin office provided daily water level data in the Samin catchment for the period 1990–2013, including the rating curve to convert the water level data into discharge data for model calibration. This study used the same climate and discharge dataset as Marhaento et al. (2017b), who have filled and corrected time series of the daily climatological and discharge data of the Samin catchment.

5.3. Method

5.3.1. Reconstruction of past land use change

Land use patterns over the period 1982-2013 have been reconstructed based on Landsat images from the years 1982, 1994, 2000, 2006 and 2013, following the approach of Marhaento et al. (2017a). The Landsat images have been processed through two steps: pre-processing and image classification. Preprocessing included a non-systematic geometric correction to avoid distortion of map coordinates and a masking analysis to remove the area outside the study area. Image classification was performed using a maximum likelihood approach based on a thousand ground-control points (GCPs) from Marhaento et al. (2017a). For this study, land use classes in Marhaento et al. (2017a) are reclassified into four land use classes: forest area (combining evergreen forest and mixed garden), agricultural area (combining paddy field and dry land farming), settlements and other areas (combining shrub, bare land and water body). The latter covers less than 5% of the catchment area.

In order to characterize land use patterns for respective years, landscape metrics for each land use distribution were calculated using FRAGSTATS (McGarigal et al., 2012). Landscape metrics are quantitative indices that describe spatial aspects (e.g. size and edge, shape, connectivity) of landscapes based on spatial data (Kupfer, 2012). FRAGSTATS is software that can be used to calculate numerous landscape metrics simultaneously within a Geographical Information System. It has been widely used in landscape analysis including its relations with hydrological processes (Lin et al., 2007; Zhang et al., 2013; Li and Zhou, 2015).

There are different spatial levels at which landscapes can be analyzed, namely cell, patch, class, and landscape level. The selection of the appropriate level for analysis depends on the level of heterogeneity for the question under consideration (McGarigal et al., 2012). This study focuses on the class level since this study aims to analyze the spatial patterns for individual land use types within a catchment. Since there are numerous landscape metrics available with redundant information, this study selected fifteen metrics that represent a wide range of land use pattern characteristics and were previously reported having a relationship with hydrological processes (Lin et al., 2007; Zhang et al., 2013; Li and Zhou, 2015). These metrics include five metrics that describe landscape size and edge, namely percentage of land use type (PL), number of patches (NP), patch density (PD), largest patch index (LPI) and edge density (ED), five metrics that describe landscape shape (i.e. geometric complexity of patch types), namely perimeter-area ratio (PAR), shape index (SHI), fractal dimension index (FDI), related circumscribing circle (RCC) and contiguity index (CI), and five metrics that describe landscape aggregation (i.e. tendency of patch types to be spatially

aggregated), namely euclidian nearest neighborhood distance (ENN), proportion of like adjacencies (PLA), splitting index (SPI), landscape shape index (LSI) and patch cohesion index (PCI). A detailed description of these landscape metrics including their equations can be found in McGarigal and Marks (1995).

5.3.2. Hydrological model simulations

Hydrological processes were simulated using the SWAT model (Arnold et al., 1998), a physically based semi-distributed hydrological model operating at a daily time step, which has proven its suitability for hydrologic impact studies around the world (Wagner et al., 2013; Memarian et al., 2014, Marhaento et al., 2017a). The SWAT model divides a catchment into sub-catchments and further divides each sub-catchment into hydrological response units (HRUs), at which level a land-phase water balance is calculated (Neitsch et al., 2011). A HRU is defined as a lumped area within a catchment assumed to have uniform hydrological behavior and is characterized by a dominant land use type, soil type and slope (Arnold et al., 1998). A daily water balance is computed for each HRU for each sub-catchment, and runoff is routed through channels to the catchment outlet where the water balance of the catchment is calculated. A SWAT model calibrated and validated for the Samin catchment was available from a previous study (Marhaento et al., 2017a). In the current study, equal SWAT model settings were applied as in the previous study. However, since this study used a coarser land use classification, thus crop parameters for each land use class used in this study are averages (e.g. forest is a combination of evergreen forest and mixed garden). Here, a brief summary of the model setup and parameterizations is presented; a detailed description can be found in Marhaento et al. (2017a).

Eleven sub-catchments ranging in size from 0.12 km² to 83 km² have been generated from the DEM. A land use map with four classes (i.e. forest, settlement, agriculture and other areas), a slope map with five classes (i.e. 0-8%, 8-15%, 15-25%, 25-45%, >45%) and a soil map with three classes (i.e. luvisols, andosols and vertisols) were used to create Hydrological Response Units (HRUs), by spatially overlying maps of land use, soil and slope classes. The Penman-Monteith method and the Soil Conservation Service Curve Number (SCS CN) were used to calculate reference evapotranspiration and surface runoff, respectively. For flow routing, this study used the Muskingum method that models the storage volume as a combination of wedge and prism storages (Neitsch et al., 2011). Six SWAT parameters were calibrated based on the observed discharge using the Latin Hypercube Sampling approach from the Sequential Uncertainty Fitting version 2 (SUFI-2) in the SWAT-Calibration and Uncertainty Procedure (SWAT-CUP) package on a monthly basis. The

calibration period is 1990–1995 and the validation period is 1996-2013. The Nash Sutcliffe Efficiency (NSE) was used as objective function. The NSE values for model calibration and validation were 0.78 and 0.70, respectively.

5.3.3. Changes in land use patterns and streamflow characteristics

Effects of past land use changes on hydrological processes were simulated using the validated SWAT model, using as inputs land use distributions in the years 1982, 1994, 2000, 2006 and 2013 and observed meteorological time series in the period 1983-2013, with a 2-years "warming-up" period. Each land use distribution was simulated separately with the same meteorological input so that the differences obtained from the simulations are only due to the differences in land use. For each simulation, the ratio of mean annual surface runoff to streamflow (Q_{s}/Q) and the ratio of streamflow in the dry season to streamflow in the wet season (Q_{dry}/Q_{wet}) is determined. Following the mean monthly precipitation from 1983-2013, it was observed that July-September (JAS) is the driest period of the year and January-March (JFM) the wettest. For this reason, streamflow in the period JAS and JFM were accumulated to represent streamflow in the dry season and the wet season, respectively. While Q₂/Q is often used as an indicator for determining the severity of land degradation of the catchment, Q_{drv}/Q_{wet} can be used as a measure to represent the seasonal balance of water availability within the catchment. Furthermore, the relationships between changes in land use metrics and streamflow characteristics were analyzed using Pearson's correlation statistic.

5.3.4. Simulation of land use scenarios

In order to explore changes in streamflow characteristics in relation to changes in land use patterns, hydrological processes were simulated under different land use scenarios. In this scenario analysis, the spatial patterns of the land use planning of the study area available from the Central Java Provincial Government (2010) were changed, while the percentages of different land use types remained unchanged (as depicted in Figure 5.1). The land use planning is used as a baseline scenario and provides spatial constraints for the land use scenarios. Land use planning was selected because it is assumed to be the future direction of land use management in the study area. Developing land use scenarios based on the land use planning can support the exploration of potential hydrological impacts of alternative land use planning. Land use scenarios were developed at a 250×250 m² spatial resolution following the spatial scale of the land use planning.
In line with the land use planning, settlements are located in the elevation range from 0 to 1,000 meter above sea level and slope range from 0% to 15%, agriculture is located in the elevation range from 0 to 1000 meter above sea level and slope range from 0 to 20%, and forest is located in the elevation range from 100 to 3,125 meter above sea level and slope range from 0% to 50%. Based on these spatial constraints, locations above 1,000 meter above sea level and slope >20% are only allocated to forest area, while other locations can be for every land use type, thus giving room to develop land use pattern scenarios. Within these constraints, two opposite land use scenarios were created to represent a wide range of possible configurations. In the "clustered land use scenario" each land use class was combined at sub-catchment level following the guideline from the Presidential Regulation of the Republic of Indonesia number 32 (1990), where settlements are grouped and positioned at low elevations and flat to middle slopes, agriculture is grouped and positioned at middle elevations and flat to middle slopes, and forest is grouped and positioned at high elevations and steep slopes. In the "scattered land use scenario" each land use class was split at sub-catchment level into several patches and spread within the catchment. It should be noted that in the land use redistribution process, the mean elevation and slope of each land use class were maintained so that land use classes in this scenario have a similar mean elevation and slope as in the land use planning.

5.4. Results

5.4.1. Past land use evolution

Figure 5.3 shows the land use distributions in the Samin catchment in different years in the period 1982-2013. The land use change over this period is characterized by a significant decrease in forest area (-36.1%), which mainly converted into settlement area (+29.6%) and agriculture area (+7.7%), while changes in other land use types were relatively small (-1.2%). The forest area decreased by 4.2% over the period 1982-1994, by 6.6% over 1994-2000, by 16.1% over 2000-2006 and by 9.2% over 2006-2013, while the settlement area increased by 5%, 6.5%, 9.4% and 8.7% over the four respective periods. Agriculture area increased by 1.5% in the period 1982-1994, then decreased by 2.4% in the period 1994-2000, and increased again by 8.6% in the period 2000-2013.

Changes in the distribution of land use types in the period 1982-2013 were accompanied by changes in spatial land use patterns. Table 5.1 shows the landscape metric values for 1982, 1994, 2000, 2006 and 2013 per land use type.



Figure 5.3. Land use distribution of the Samin catchment in 1982 (a), 1994 (b), 2000 (c), 2006 (d) and 2013 (e).

The decrease in the percentage of forest, is accompanied by a decrease in the largest patch index (LPI) and the patch cohesion index (PCI), while there is no clear relation with other metrics. LPI quantifies the percentage of the landscape comprised by the largest patch, while PCI quantifies the physical connectivity of patches from the corresponding land use type. The LPI value for forest was 41.5 in 1982, but decreased to 38.2 (1994), 17.2 (2000), 5.1 (2006) and 2.3 (2013), while the PCI value for forest was 99.8 in 1982, which decreased to 99.7 (1994), 99.2 (2000), 98.2 (2006) and 97 (2013). The increase in the percentage of settlement area, is accompanied by an increase in the edge density (ED) and the PCI values, while there is no clear relation with other metrics. ED quantifies the ratio of edge segment length to total area. The ED value for settlement was 10.8 in 1982 and increased to 24.3 (1994), 39.9 (2000), 41.9 (2006) and 53.9 (2013), while the PCI value of settlement was 91.1 in 1982 and increased to 91.3 (1994), 92.5 (2000), 96.2 (2006), 98.8 (2013). For agriculture, there is no clear relation between changes in the percentage of land use and changes in the other metrics.

Metrics*	PL	NP	PD	LPI	ED	PAR	SHI	FDI	RCC	CI	ENN	PLA	SPI	LSI	PCI
Agriculture															
1982	37.8	390	1.4	28.9	49.1	508	1.5	1.1	0.6	0.6	140	90.2	12	33.3	99.7
1994	39.2	764	2.7	27.6	60.1	639	1.4	1.0	0.5	0.5	104	88.5	13	39.9	99.6
2000	36.9	791	2.8	13.5	55.8	635	1.4	1.1	0.5	0.5	102	88.7	36	38.3	99.1
2006	44.3	702	2.5	19.6	49.9	599	1.3	1.0	0.5	0.5	106	91.5	23	31.2	99.2
2013	45.5	690	2.5	31.7	60.5	609	1.4	1.0	0.5	0.5	99	90.0	10	37.3	99.6
Settlement															
1982	4.7	262	0.9	0.3	10.8	438	1.4	1.1	0.6	0.6	247	82.8	24,185	20.6	91.1
1994	9.7	856	3.1	0.9	24.3	604	1.2	1.0	0.5	0.5	140	81.2	6,409	32.5	91.3
2000	16.1	997	3.6	0.8	39.9	611	1.4	1.1	0.5	0.5	108	81.4	3,933	41.4	92.5
2006	25.4	629	2.3	1.4	41.9	462	1.4	1.1	0.5	0.6	110	87.6	837	34.6	96.2
2013	34.2	821	3.0	10.4	53.9	536	1.3	1.0	0.5	0.5	110	88.2	76	38.4	98.8
Forest															
1982	52.9	655	2.4	41.5	52.0	580	1.3	1.0	0.5	0.5	118	92.6	6	29.8	99.8
1994	48.7	720	2.6	38.2	55.7	579	1.3	1.0	0.5	0.5	124	91.4	7	33.2	99.7
2000	42.2	698	2.5	17.2	62.4	497	1.5	1.1	0.5	0.6	110	88.9	24	40.0	99.2
2006	25.9	661	2.4	5.1	36.8	553	1.3	1.0	0.5	0.5	132	89.4	167	30.1	98.2
2013	16.8	711	2.6	2.3	29.4	623	1.3	1.0	0.5	0.5	145	86.9	545	29.9	97.0

Table 5.1. Landscape metrics for 1982, 1994, 2000, 2006 and 2013 at class level.

* PL = Percentage of land use type (%), NP = Number of patches (-), PD = Patch density (number per 100 hectare), LPI = Largest patch index (%), ED = Edge density (meter/hectare), PAR = Perimeter-area ratio (-), SHI = Shape index (-), FDI = Fractal dimension index (-), RCC = Related circumscribing circle (-), CI = Contiguity index (-), ENN = Euclidian nearest neighborhood distance (meter), PLA = Proportion of like adjacencies (%), SPI = Splitting index (-), LSI = Landscape shape index (-), PCI = Patch cohesion index (-).

5.4.2. Correlations between land use patterns and streamflow characteristics

Figure 5.4 shows that land use changes in the period 1982-2013 have affected simulated streamflow characteristics. The long-term average ratio of mean annual surface runoff to streamflow (Q_s/Q) increased from 0.28 in 1982 to 0.31 (1994), 0.33 (2000), 0.36 (2006) and 0.39 (2013), whereas the long-term average ratio of dry-season to wet-season streamflow (Q_{dry}/Q_{wet}) decreased from 0.13 in 1982 to 0.12 (1994), 0.11 (2000), 0.1 (2006) and 0.09 (2013). A consistent increase of Q_s/Q and a consistent decrease of Q_{dry}/Q_{wet} is in line with the direction of changes in forest and settlement area, where forest has significantly decreased and settlement has significantly increased in the period 1982-2013.



Figure 5.4. (a) Box-whisker plot of the ratio of mean annual surface runoff to streamflow (Q_s/Q) and (b) box-whisker plot of the ratio of mean annual streamflow in the dry season to streamflow in the wet season (Q_{dry}/Q_{wet}) from simulations using land use distributions from different years.

Based on the results of Pearson's correlation analysis between various landscape metrics and streamflow characteristics, it is found that the percentage of forest (PL_f) negatively correlates and the percentage of settlement (PL_s) positively correlates with Q_s/Q , at a significance level of 1%. The cohesion of forest area (PCI_f) negatively correlates and the cohesion of settlement (PCI_s) positively correlates with Q_s/Q , at a significance level of 1%. For Q_{dry}/Q_{wet} , the percentage of forest (PL_f) positively correlates, while the percentage of settlement (PL_s) and the cohesion of settlement (PCI_s) negatively correlate, at a significance level of 1%. Figure 5.5 shows the estimated Pearson's correlation coefficients between landscape metrics and streamflow characteristics. It should

be noted that the correlation analysis was performed at a significance level 1% due to the small size of the samples (n=5).



Figure 5.5. Pearson's correlation coefficient for relations between landscape metrics and the ratio of mean annual surface runoff to streamflow (Q_s/Q) (a), and the ratio of streamflow in the dry season to streamflow in the wet season (Q_{dry}/Q_{wet}) (b). Bars with orange color show landscape metrics with a significance level of 1%. Letters a, f and s refer to agriculture, forest and settlement, respectively.

5.4.3. Changes in streamflow characteristics under different land use scenarios

The correlation analysis shows that the percentages of forest and settlement (PL_s and PL_f) and the patch cohesion index of forest and settlement (PCI_s and PCI_f) were the metrics that best correlated to streamflow characteristics (α =1%). It was expected that PL_s and PL_f , would significantly affect streamflow characteristics based on previous studies for the Samin catchment (Marhaento et al., 2017a; 2017b). The relations of PCI_s and PCI_f , with the streamflow

characteristics independent of the percentage of settlement and forest area was assessed using the scenario analysis. Figure 5.6 shows the land use distribution of the baseline scenario (i.e. land use planning), the clustered land use scenario and the scattered land use scenario, and Figure 5.7 shows the differences of PCI values for each land use scenario compared to the baseline scenario. It is shown that the clustered scenario has larger PCI values and the scattered scenario smaller PCI values than the baseline scenario.



Figure 5.6. Land use distribution of the baseline scenario (i.e. land use planning) (a), the clustered scenario (b), and the scattered scenario (c). The land use scenarios represent different cohesion values while the percentages of the different land use types are fixed.



Figure 5.7. Patch cohesion index (PCI) of each land use type for different land use scenarios with each the same percentages of the different land use types.

At catchment scale, the results of the simulations show that the clustered and scattered scenarios reduce the long-term average of Q_s/Q . In comparison to the baseline scenario, the long-term average of Q_s/Q reduced from 0.37 to 0.35 under the clustered land use scenario and from 0.37 to 0.36 under the scattered land use scenario. The long-term average of Q_{dry}/Q_{wet} increased under the clustered land use scenario and reduced under the scattered scenario. Figure 5.8 shows that simulations using different land use scenarios may result in different hydrological responses. However, the changes in the long-term average of Q_s/Q and Q_{dry}/Q_{wet} at catchment scale under different scenarios were relatively small (i.e. ranged from -0.02 to +0.01).



Figure 5.8 (a) Changes in the ratio of surface runoff to total runoff (Q_s/Q) and (b) the ratio of streamflow in the dry season to streamflow in the wet season (Q_{dry}/Q_{wet}) under different scenarios relative to the baseline scenario, including coefficient of variation of annual precipitation for the period 1985-2013.

Clear effects of changes in the patch connectivity on the streamflow characteristics can be observed at the sub-catchment level, in particular for changes in PCI_s. As can be seen in Figure 5.9, Q_s/Q has increased up to +0.1 following an increase in PCI_s up to +0.9 under the clustered land use scenario, whereas Q_s/Q has decreased up to -0.18 following a decrease in PCI_s up to -2.4 under the scattered land use scenario. Conversely, Q_{dry}/Q_{wet} has increased up to +0.02 following a decrease in the PCI_s up to -2.4 under the scattered scenario, while less clear impacts on Q_{dry}/Q_{wet} under the clustered land use scenario can be observed. While the effects of changes in PCI_s on the streamflow characteristics are discernible, there is no clear relation between PCI_f and the streamflow characteristics.



Figure 5.9. Relationship between changes in the patch cohesion index for settlement (PCI_s) and forest (PCI_f) and changes in streamflow characteristics relative to the baseline scenario.

5.5. Discussion

The study shows that changes in the streamflow characteristics of the study catchment can be attributed to both changes in the percentages of the different land use types and changes in the physical connectivity between patches of similar land use types. A decrease of vegetation and an increase of impervious areas were likely the cause of substantial changes in the streamflow generation. A consistent decline of forest area in the catchment due to conversion into settlement area has decreased the infiltration rate resulting in reduced groundwater recharge and water storage in the soil. As a result, a larger volume of precipitation was transformed into surface runoff. With less water infiltrated and stored into the soil, an increased fraction of flow becoming surface runoff and a less balanced water distribution between the wet and dry seasons can be expected. These findings regarding the relationship between changes in the percentages of different land use types and changes in the streamflow characteristics confirm those in earlier studies (e.g. Bruijnzeel, 1989; 2004; Brown et al., 2005; Wagner et al., 2013; Remondi et al., 2013; Gumindoga et al., 2014b; Marhaento et al. 2017a; 2017b).

When the percentages of different land use types are fixed but the physical connectivity of patches is changed, changes in streamflow characteristics occur at both catchment and sub-catchment level. However, changes at catchment level are very small, where different patch cohesion indices from different land use scenarios hardly affect the streamflow characteristics. A clearer relationship between streamflow characteristics and patch connectivity can be observed at sub-catchment scale, where Qs/Q is positively correlated and Qdrv/Qwet is negatively correlated with the patch connectivity of settlements. Apparently, the opposite impacts of changes in the patch connectivity at sub-catchment level compensate each other at catchment level. It was expected that the clustered forest area in the upstream part would reduce Q_s/Q and increase Q_{drv}/Q_{wet} while the clustered settlement area in the downstream part would increase Q_s/Q and decrease Q_{drv}/Q_{wet}. With different directions of changes in streamflow characteristics between upstream and downstream areas within the catchment, long-term changes of those variables at catchment level canceled each other out. This is similar to the findings from Wilk and Hughes (2002) and Wagner et al. (2013) who argue that the net hydrological result at catchment level can mask the impacts of land use changes on hydrological processes at sub-catchment scale.

The impacts of changes in the patch connectivity of settlement on streamflow characteristics are more pronounced than the impacts of changes in the patch connectivity of forest. This study agrees with Mejía & Moglen (2009), Wagner et al. (2013) and Su et al. (2014) who argue that urban imperviousness patterns

within catchments can play an important role in determining changes in streamflow characteristics, particularly because of changes in the fraction of flow becoming surface runoff. When the settlement area is clustered, peak flows will increase without much affecting the flow volume compared to scattered settlement (Corbett et al., 1997). When the urban development is located in the downstream area near the catchment outlet (as simulated in this study under the clustered scenario), more concentrated impacts can be expected compared to settlement scattered over the catchment (Su et al., 2014; Wheater & Evans, 2009).

It was simulated that the percentages of different land use types strongly affect the runoff generation of the study catchment, while the patch connectivity for a certain land use type may affect surface runoff and the seasonal balance of flows by accelerating or decelerating runoff responses. However, it should be noted that the spatial heterogeneity of hydrological response over catchments is scale-dependent, where the landscape metrics may change with changing scale (McGarigal & Marks, 1995; McGarigal et al., 2012) as well as dominant hydrological processes (Blöschl et al., 2007; Zhang et al., 2013). This study uses a semi-distributed SWAT model that simulates hydrological processes in each Hydrological Response Unit (HRU) within sub-catchments. Thus, the water balance simulation will be at a HRU, which is a combination of land use type, soil type and slope class. With the coarse scale of land use scenarios used in this study, spatial detail of changes in the land use patterns might be lost. As shown in Figure 5.9, both land use scenarios provide relatively small changes in the patch connectivity at sub-catchment scale. Although the results show relatively clear signals for changes in the streamflow characteristics due to changes in the patch connectivity, a more discernible signal can be expected if the land use change scenarios were developed at a finer scale. SWAT model is able to simulate streamflow only at the sub-catchment level so that the spatial flow variation between HRUs within a sub-catchment is not taken into account. With this model limitation, changes in hydrological processes due to changes in land use patterns cannot be presented at a smaller scale than at sub-catchment scale. To further investigate the impacts of different land use metrics on streamflow characteristics, the use of a fully-distributed model, e.g. the Gridded Surface Subsurface Hydrologic Analysis (GSSHA) model (Zhang et al., 2013) can be considered.

Although the effects of climate variability are isolated in the simulations by forcing the model with the same meteorological input for different land use inputs, the contribution of climate variability to changes in streamflow characteristics cannot be neglected. As can be seen in Figure 5.8, the hydrological response to different land use scenarios varies across years, likely

due to precipitation variability. Marhaento et al. (2017c) found for the same study area that changes in the precipitation variability may have large impacts on the water availability of the study catchment.

Findings of this study offer an opportunity to include hydrological impact considerations in developing land use plans. For the study catchment, the existing land use planning has been projected to reduce the mean annual streamflow and surface runoff, and increase the mean annual base flow and evapotranspiration (Marhaento et al., 2017c). By altering the spatial patterns of the land use plan, e.g. by scattering settlement area in the downstream area and clustering forest area in the upstream area, surface runoff can be reduced and a more balanced distribution of streamflow between the dry and wet season can be achieved. Since land use planning always occurs at different scales (e.g. national level, provincial level and district level), multi-scale analysis of the impacts of land use patterns on streamflow characteristics is recommended.

5.6. Conclusions

This study has assessed the effects of changes in land use patterns on streamflow characteristics in the Samin catchment and found that changes in the ratio of surface runoff to streamflow (Q_{a}/Q) and the ratio of dry-season to wetseason streamflow (Q_{drv}/Q_{wet}) are significantly affected by the percentage of forest and settlement (PLf and PLs) and the patch cohesion index of forest and settlement (PCI_f and PCI_s). Simulations with the SWAT model indicate that a decrease of PL_f and PCI_f and an increase of PL_s and PCI_s cause an increase of Q_s/Q and a decrease of Q_{drv}/Q_{wet}. Independently, changes in PCI may affect the streamflow characteristics, where clear relationships were found at the subcatchment level. When simulating the impact of two hypothetical land use scenarios, a clustered scenario and a scattered scenario, it was found that an increase in PCI_s with PL_s maintained may result in an increase of Q_s/Q and a decrease of Q_{drv}/Q_{wet}, and vice versa, while changes in PCI_f with PL_f maintained have less impact on streamflow characteristics. The findings show that, particularly for settlements, altering the spatial configuration can be an effective measure to achieve a more balanced distribution of streamflow between the dry and wet season.

Chapter 6 Conclusions

6.1. Overview of the main findings

Analysis of streamflow time series for the period 1990-2013 for the Samin catchment show that annual streamflow has increased by an average of 12.1 mm/year. Chapter 2 shows that the contributions of changes in land use and climate to this change in streamflow were estimated to be about 72% and 28%, respectively. This result has been corroborated by land use change analysis and statistical trend analysis for the mean annual streamflow, precipitation and potential evapotranspiration. These analyses show that an increase in the annual streamflow during 1990-2013 was in line with a decrease in forest area from 48.7% to 16.9%, an increase in agriculture area from 39.2% to 45.4% and an increase in settlement area from 9.8% to 34.3%, while changes in climate variables (i.e. precipitation and potential evapotranspiration) are minor. In addition, land use change has affected other water balance components, such as actual evapotranspiration and the fractions of precipitation that become surface runoff, lateral flow and base flow.

In Chapter 3, the Soil and Water Assessment Tool (SWAT) was used to simulate hydrological processes in the study catchment. It was found that, at the catchment level, land use changes have resulted in an increase of the ratio of streamflow to precipitation from 35.7% to 44.6%, a decrease in the ratio of evapotranspiration to precipitation from 60% to 54.8%, an increase in the ratio of surface runoff to streamflow from 26.6% to 37.5% and a decrease in the ratio of base flow to streamflow from 40% to 31.1%. At the sub-catchment level, the impacts of land use changes on the water balance were diverse, depending on the extent of changes in forest and settlement areas. It was observed that positive and negative changes in the water balance components at sub-catchment level have counterbalanced each other, resulting in relatively small long-term changes of those variables at catchment level. Although it was found that changes in the hydrological processes can be mainly attributed to land use change rather than climate change, the contribution of climate change to changes in hydrological processes cannot be neglected.

In Chapter 4, the future climate variability (2030-2050) of the study catchment was assessed based on the output of a bias-corrected Regional Climate Model and the outputs of six Global Climate Models to include climate model uncertainty. Under two climate change scenarios namely Representative Concentration Pathway (RCP) 4.5 and 8.5, it was projected that climate change

may significantly affect water availability in particular if the changes in the mean annual precipitation are larger than 10%. The impacts of future climate conditions on water availability can be larger when combined with the future land use distributions under the business-as-usual scenario. It was observed that the combination of the RCP 4.5 climate scenario and the business-as-usual land use scenario may result in changes in mean annual streamflow and surface runoff ranging from -7% to +64% and from +21% to +102%, respectively, which is about 40% and 60% more than when land use change is acting alone. In this chapter, the effectiveness of governmental land use planning were also estimated. It was projected that under the RCP 4.5 climate scenario, land use planning reduces the mean annual surface runoff up to 30% and increases the mean annual base flow up to 8%. The effectiveness of applied land use planning to mitigate future water-related risks can be improved if the spatial land use distribution is optimized.

In Chapter 5, the effects of land use patterns on streamflow characteristics were assessed. It was found that two land use pattern characteristics, namely the percentage of different land use types and the physical connectivity between patches of similar land use types within the catchment, are dominant drivers for changes in the streamflow characteristics. While effects of changes in the percentages of different land use types on streamflow characteristics are expected (based on the results of Chapter 2, 3 and 4), it is interesting to know how changes in the physical connectivity between patches of similar land use types may affect streamflow characteristics as well. Based on the SWAT model simulations, using two hypothetical land use scenarios as inputs, a clustered scenario and a scattered scenario, an increase in the settlement connectivity can result in an increase in the ratio of surface runoff to streamflow and a decrease in the ratio of streamflow in the dry season to streamflow in the wet season, and vice versa, while changes in the forest connectivity have less impacts on streamflow characteristics. Findings of this study further suggest that land use management can be an important component of water management in the study catchment.

6.2. Limitations and future outlook

To achieve the research objective, this study uses various datasets such as spatial data (e.g. satellite images, elevation data, soil data, administrative maps) and non-spatial data (e.g. hydro-meteorological data) acquired from governmental and non-governmental agencies. In addition, this study employs a hydrological model and outputs from land use change and climate models to assess past and future impacts of land use change and climate change on hydrological processes. The data and models used can be a source of uncertainty. For the data, it is a challenge to obtain long-term and reliable hydrometeorological data for the study catchment. This is typically for South-East Asian countries including Indonesia where meteorological gauge networks generally include a limited number of stations which, commonly, are not well distributed over catchments (Douglas, 1999). In addition, for the study catchment most of the hydro-meteorological data were manually operated resulting in potential recording errors. For these reasons, data checks for errors and data corrections using well-established methodologies as shown in this study (Chapter 2) can be useful to construct more reliable datasets. As input for the SWAT model, this study used a soil map from FAO et al. (2012) with a coarse resolution. This global soil map was used because of limited information on soil characteristics in the local soil map available in the study catchment. Because of this soil generalization, several details important for hydrological processes for different soil conditions (e.g. surface runoff from different combinations of soils and slopes) might be obscured. For the models used in this study (Chapters 3, 4 and 5), sources of uncertainty can be present due to model choice and structure (e.g. model assumptions, equations, parameterization), data inputs (e.g. lack of relevant spatial and temporal variability of data on precipitation, soils, land uses and topography) and assumptions used to predict and project future conditions (e.g. population, climatic forcing). In addition, for future simulations it was assumed that land use change is independent of climate change so that mutual interactions between land use change and climate change due to local atmospheric feedbacks of land surface processes are not taken into account. Nevertheless, future changes in land use and climate are hard to predict since these changes are controlled by many policy-related, economic, and environmental factors which were not fully considered in this study. Considering all these uncertainties, accumulation of errors can be substantial affecting the results. Consequently, conclusions drawn in this study should be interpreted taking into account the uncertainty in the results (e.g. relatively large uncertainties in changes in the simulated water balance components as a result of land use change as can be seen in Chapter 4).

For future research, four main points are suggested for consideration. First, it is essential to use a reliable hydro-meteorological dataset. As mentioned earlier, the data-based approach applied in Chapter 2 heavily relies on the quality of hydro-meteorological data in particular for the streamflow data. Applying data corrections to the streamflow data (as conducted in Chapter 2) will affect the attribution results and the uncertainties can be considerable. In addition, the meteorological data obtained from the field gauges were the main input for the SWAT model (Chapter 3, 4 and 5), so that the spatial and temporal distribution of the meteorological gauges may affect the simulation results (see Cho et al.,

2009). Calibrating a model using observed, erroneous streamflow data will lead to misinterpretations of the overall water balance (see Montanari & Baldassarre, 2013; Van den Tillaart et al., 2013). For these reasons, it is expected that this study can provide more convincing results if hydro-meteorological datasets are available for a long time period (at least 30 years) and extracted from reliable and well distributed gauges over the study catchment. Second, it is suggested to use a soil map with a higher resolution as input for the hydrological model. This study used a global soil map with a coarse resolution as input for the SWAT model rather than a local soil map available for the study catchment. The global soil map was selected because it provides information on the soil characteristics required by the model (e.g. soil texture, soil depth, soil permeability, etc.). In order to use the local soil map as input for the SWAT model, field measurements including laboratory experiments can be conducted to provide information on the soil characteristics of the local soil map. The third point is related to the robustness of the hydrological model (i.e. SWAT model). Although this study has shown the ability of the SWAT model to achieve the research objective, improvements on model structures can be considered for future research. It should be noted that the SWAT model is able to simulate streamflow only at the sub-catchment level (i.e. it is a semi-distributed model) and neglects the spatial flow variation between Hydrological Response Units (HRUs) within a subcatchment. Consequently, this limits the impact assessment of different land use configurations on water availability within a given sub-catchment (Chapter 5). For this reason, the use of a fully-distributed model e.g. SWATgrid, a grid-based version of the SWAT model developed by Rathjens & Oppelt (2012), can be considered for future research. In addition, the model robustness can be potentially improved by the use of additional variables besides streamflow (e.g. actual evapotranspiration) in model calibration and validation. Remote sensing products such as MODIS data can be used for this purpose to generate spatially and temporally distributed actual evapotranspiration (see Rientjes et al., 2013). Finally, it should be mentioned that this study uses only one single catchment. For this reason, more research in different catchments with various hydrological and climatic conditions is needed to test the general applicability of the methods and reliability of the results.

6.3. Contribution to scientific advancement

This study provides insight in the impacts of land use change and climate change on water availability in the study catchment for past and future conditions. In addition, this thesis contributes to the advancement of the field of hydrology in four ways. First, a more thorough understanding of the methods available to attribute changes in hydrological processes of a tropical catchment

to land use change and climate change was gained. The methods used in this thesis to separate the impacts of land use change and climate change on hydrological processes (in Chapters 2 and 3) were applied for the first time in a tropical region. Moreover, the methods used for the attribution analysis were extended, thus advancing the methods. Second, this thesis offers an assessment of the effectiveness of land use planning to mitigate future risks associated with land use change and climate change (in Chapters 4 and 5). The methods and results of the thesis provide practical guidelines which can bridge the gap between land use planning and water resource management. As a result, this study can be useful for supporting the authorities to design appropriate land use management measures to achieve water sustainability and to improve water management. Third, this thesis shows the extents and directions of land use change and climate change impacts on water availability in a tropical catchment. Although this research is conducted in a single catchment (i.e. Samin catchment, Java, Indonesia), it is thought to represent problems characteristic for the hydrology in tropical regions having similar climate and land use conditions. Fourth, while most of the research focuses on the impacts of different land use types on water resources, this thesis successfully assessed the impacts of different land use configurations on the water availability as well (Chapter 5). For tropical regions, this assessment is one of the first studies investigating the relations between land use patterns and water resources. Ultimately, I would like to encourage and motivate colleagues in the same field to carry out more research in this very interesting yet challenging topic.

References

- Abbaspour, K. C., Rouholahnejad, E., Vaghefi, S., Srinivasan, R., Yang, H., & Kløve, B. (2015). A continental-scale hydrology and water quality model for Europe: Calibration and uncertainty of a high-resolution large-scale SWAT model. *Journal of Hydrology*, 524, 733-752.
- Alansi, A. W., Amin, M. S. M., Abdul Halim, G., Shafri, H. Z. M., & Aimrun, W. (2009). Validation of SWAT model for streamflow simulation and forecasting in Upper Bernam humid tropical river basin, Malaysia. *Hydrology and Earth System Sciences Discussions*, 6(6), 7581-7609.
- Aldrian, E., & Djamil, Y. S. (2008). Spatio-temporal climatic change of rainfall in East Java Indonesia. International Journal of Climatology, 28(4), 435-448.
- Amien, I., Rejekiningrum, P., Pramudia, A., & Susanti, E. (1996). Effects of interannual climate variability and climate change on rice yield in Java, Indonesia. Water, Air, and Soil Pollution, 92(1-2), 29-39.
- Anderson, J. R. (1976). A land use and land cover classification system for use with remote sensor data (Vol. 964). US Government Printing Office: Washington, DC, USA.
- Andréassian, V. (2004). Waters and forests: from historical controversy to scientific debate. *Journal of Hydrology*, 291(1), 1-27.
- Arnold, J. G., Srinivasan, R., Muttiah, R. S., & Williams, J. R. (1998). Large area hydrologic modeling and assessment part I: Model development. *Journal of the American Water Resources Association*, 34(1), 73-89.
- Arora, V. K., Scinocca, J. F., Boer, G. J., Christian, J. R., Denman, K. L., Flato, G. M., Kharin, V. V., Lee, W. G., Merryfield, W. J. (2011). Carbon emission limits required to satisfy future representative concentration pathways of greenhouse gases. *Geophysical Research Letters*, 38(5).
- Azevedo, J. C., Williams, J. R., Messina, M. G., & Fisher, R. F. (2005). Impacts of the sustainable forestry initiative landscape level measures on hydrological processes. *Water Resources Management*, 19(2), 95-110
- Bakker, M. M., Govers, G., van Doorn, A., Quetier, F., Chouvardas, D., & Rounsevell, M. (2008). The response of soil erosion and sediment export to land-use change in four areas of Europe: the importance of landscape pattern. *Geomorphology*, 98(3), 213-226.
- Beck, H. E., Bruijnzeel, L. A., Van Dijk, A. I. J. M., McVicar, T. R., Scatena, F. N., & Schellekens, J. (2013). The impact of forest regeneration on streamflow in 12 mesoscale humid tropical catchments. *Hydrology and Earth System Sciences*, 17(7), 2613-2635.
- Behera, D., S.N. Borate, S.N. Panda, P.R. Behera & P.S. Roy. (2012). Modelling and analyzing the watershed dynamics using Cellular Automata (CA)–Markov model–A geo-information based approach. *Journal of Earth System Science*, 121(4), 1011-1024.
- Bentsen, M., Bethke, I., Debernard, J. B., Iversen, T., Kirkevåg, A., Seland, Ø., Drange, H., Roelandt C., Seierstad, I. A., Hoose, C., Kristjánsson, J. E. (2013). The Norwegian earth system model, NorESM1-M—Part 1: description and basic evaluation of the physical climate. *Geosci. Model Dev*, 6(3), 687-720.
- Blöschl, G., Ardoin-Bardin, S., Bonell, M., Dorninger, M., Goodrich, D., Gutknecht, D., Matamoros, D., Merz, B., Shand, P. and Szolgay, J. (2007). At what scales do

climate variability and land cover change impact on flooding and low flows?. *Hydrological Processes*, *21*(9), 1241-1247.

- Bonell, M., Hufschmidt, M. M., & Gladwell, J. S. (2005). Hydrology and water management in the humid tropics: hydrological research issues and strategies for water management. Cambridge university press.
- Bormann, H., Breuer, L., Gräff, T., Huisman, J. A., & Croke, B. (2009). Assessing the impact of land use change on hydrology by ensemble modelling (LUCHEM) IV: model sensitivity to data aggregation and spatial (re-) distribution. Advances in Water Resources, 32(2), 171-192.
- Bosch, J. M., & Hewlett, J. D. (1982). A review of catchment experiments to determine the effect of vegetation changes on water yield and evapotranspiration. *Journal of Hydrology*, 55(1), 3-23.
- Botula, Y. D., Van Ranst, E., & Cornelis, W. M. (2014). Pedotransfer functions to predict water retention for soils of the humid tropics: a review. *Revista Brasileira de Ciência do Solo*, 38(3), 679-698.
- BPS. (2013). Karanganyar and Sukoharjo in Figures 2013. Statistical Bureau of Karanganyar and Sukoharjo Regency. Ministry of Home Affairs, Indonesia.
- BPS. (2017). Karanganyar and Sukoharjo in Figures 2017. Statistical Bureau of Karanganyar and Sukoharjo Regency. Ministry of Home Affairs, Indonesia
- Brown, A. E., Zhang, L., McMahon, T. A., Western, A. W., & Vertessy, R. A. (2005). A review of paired catchment studies for determining changes in water yield resulting from alterations in vegetation. *Journal of Hydrology*, *310*(1-4), 28-61
- Bruijnzeel, L. A. (1989). (De)forestation and dry-season flow in the tropics: A closer look. *Journal of Tropical Forest Science*, *1*, 229-243.
- Bruijnzeel, L. A. (1990). Hydrology of moist tropical forests and effects of conversion: a state of knowledge review. Humid Tropics Programme of the International Hydrological Programme of UNESCO, Paris, and Vrije Universiteit, Amsterdam, 224 pp.
- Bruijnzeel, L. A. (2004). Hydrological functions of tropical forests: not seeing the soil for the trees?. *Agriculture, Ecosystems & Environment, 104*(1), 185-228.
- Budyko, M. I. (1974). Climate and Life, Academic press, New York, USA.
- Calder, I.R. (1998). *Water Resources and Land Use Issues*. SWIM Paper 3. Colombo, Sri Lanka: International Water Management Institute
- Calder, I.R., Young, D., Sheffield, J. (2001). Scoping study to indicate the direction and magnitude of the hydrological impacts resulting from land use change on the Panama Canal Watershed: Newcastle upon Tyne, UK, Centre for Land Use and Water Resources Research.
- Carter, N., Kreutzwiser, R. D., & de Loë, R. C. (2005). Closing the circle: linking land use planning and water management at the local level. *Land Use Policy*, 22(2), 115-127.
- Central Java Provincial Government (2010). The regional regulation of Central Java province number 6 (year 2010): spatial planning of Central Java province 2009 2029. http://peraturan.go.id [access in May 14th 2017]
- Cho, J., Bosch, D., Lowrance, R., Strickland, T., & Vellidis, G. (2009). Effect of spatial distribution of rainfall on temporal and spatial uncertainty of SWAT output. *Transactions of the ASABE*, *52*(5), 1545-1556.

- Collier, M. A., Jeffrey, S. J., Rotstayn, L. D., Wong, K. K., Dravitzki, S. M., Moseneder, C., Hamalainen, C., Syktus, J. I., Suppiah, R., Antony, J., El Zein, A. Atif, M. (2011). The CSIRO-Mk3. 6.0 Atmosphere-Ocean GCM: participation in CMIP5 and data publication. In 19th International Congress on Modelling and Simulation– MODSIM. Australia
- Congalton, R. G. (1991). A review of assessing the accuracy of classifications of remotely sensed data. *Remote Sensing of Environment*, *37*(1), 35-46.
- Corbett, C. W., Wahl, M., Porter, D. E., Edwards, D., & Moise, C. (1997). Nonpoint source runoff modeling A comparison of a forested watershed and an urban watershed on the South Carolina coast. *Journal of Experimental Marine Biology and Ecology*, 213(1), 133-149.
- Costa, M. H., Botta, A., & Cardille, J. A. (2003). Effects of large-scale changes in land cover on the discharge of the Tocantins River, Southeastern Amazonia. *Journal* of *Hydrology*, 283(1), 206-217.
- D'Almeida, C., Vörösmarty, C. J., Hurtt, G. C., Marengo, J. A., Dingman, S. L., & Keim, B. D. (2007). The effects of deforestation on the hydrological cycle in Amazonia: a review on scale and resolution. *International Journal of Climatology*, 27(5), 633-647.
- Douglas, I. (1999). Hydrological investigations of forest disturbance and land cover impacts in South-East Asia: A review. *Philosophical Transactions of the Royal Society B: Biological Sciences, 354*(1391), 1725-1738
- Dunne, J. P., John, J. G., Adcroft, A. J., Griffies, S. M., Hallberg, R. W., Shevliakova, E., Stouffer, R. J., Cooke, W., Dunne, K. A., Harrison, M. J., Krasting, J. P., Malyshev, S. L., Milly, P. C. D., Phillipps, P. J., Sentman, L. T., Samuels, B. L., Spelman, M. J., Winton, M., Wittenberg, A. T., Zadeh, N. (2012). GFDL's ESM2 global coupled climate–carbon earth system models. Part I: Physical formulation and baseline simulation characteristics. *Journal of Climate*, 25(19), 6646-6665.
- Eastman, J. R. (2012). IDRISI Selva Tutorial, Manual Version 17. Clark University. Massachusetts.
- ERSDAC. (2016). USGS and Japan Aster Program, Aster scene AST_L1B_00306162004025925_20070620103614_3374 level 1B.
- Ewen, J., O'Donnell, G., Burton, A., & O'Connell, E. (2006). Errors and uncertainty in physically-based rainfall-runoff modelling of catchment change effects. *Journal of Hydrology*, 330(3), 641-650.
- Fang, G., Yang, J., Chen, Y. N., & Zammit, C. (2015). Comparing bias correction methods in downscaling meteorological variables for a hydrologic impact study in an arid area in China. *Hydrology and Earth System Sciences*, 19(6), 2547-2559.
- FAO/IIASA/ISRIC/ISSCAS/JRC. (2012). Harmonized World Soil Database (version 1.2). FAO, Rome, Italy and IIASA, Laxenburg, Austria.
- Fohrer, N., Haverkamp, S., & Frede, H. G. (2005). Assessment of the effects of land use patterns on hydrologic landscape functions: development of sustainable land use concepts for low mountain range areas. *Hydrological Processes*, 19(3), 659-672.
- Fulazzaky, M. A. (2014). Challenges of integrated water resources management in Indonesia. Water, 6(7), 2000-2020.
- Gallo, E. L., Meixner, T., Aoubid, H., Lohse, K. A., & Brooks, P. D. (2015). Combined impact of catchment size, land cover, and precipitation on streamflow and total dissolved nitrogen: A global comparative analysis. *Global Biogeochemical Cycles*, 29(7), 1109-1121.

- Giorgetta, M. A., et al. (2013). Climate and carbon cycle changes from 1850 to 2100 in MPI-ESM simulations for the Coupled Model Intercomparison Project phase 5. *Journal of Advances in Modeling Earth Systems*, *5*(3), 572-597.
- Guevara-Escobar, A., Gonzalez-Sosa, E., Ramos-Salinas, M., & Hernandez-Delgado, G.
 D. (2007). Experimental analysis of drainage and water storage of litter layers. Hydrology and Earth System Sciences, 11(5), 1703-1716.
- Gumindoga W., Rientjes, T.H.M., Shekede, M.D., Rwasoka, D.T., Nhapi, I., and Haile, A.T. (2014a). Hydrological Impacts of Urbanization of Two Catchments in Harare, Zimbabwe *Remote Sensing*, 6, 12544-12574
- Gumindoga, W., Rientjes, T. H. M., Haile, A. T., & Dube, T. (2014b). Predicting streamflow for land cover changes in the Upper Gilgel Abay River Basin, Ethiopia: A TOPMODEL based approach. *Physics and Chemistry of the Earth, Parts 2*, 76, 3-15.
- Hansen, M. C., Stehman, S. V., Potapov, P. V., Arunarwati, B., Stolle, F., & Pittman, K. (2009). Quantifying changes in the rates of forest clearing in Indonesia from 1990 to 2005 using remotely sensed data sets. *Environmental Research Letters*, 4(3), 034001.
- Hejazi, M. I., & Moglen, G. E. (2008). The effect of climate and land use change on flow duration in the Maryland Piedmont region. *Hydrological Processes*, 22(24), 4710-4722
- Huisman, J. A., Breuer, L., Bormann, H., Bronstert, A., Croke, B. F. W., Frede, H., Graft, T., Hubrechts, L., Jakeman, A.J., Kite, G., Lanini, J., Leavesley, G., Lettenmaier, D.P., Lindstrom, G., Seibert, J., Sivapalan, M., Viney, N., R Willems, P. (2009). Assessing the impact of land use change on hydrology by ensemble modeling (LUCHEM) III: Scenario analysis. *Advances in Water Resources*, *32*(2), 159-170
- Hulme, M and N. Sheard. (1999). Climate Change Scenarios for Indonesia. Climatic Research Unit, Norwich, UK, 6pp.
- Hyandye, C., & Martz, L. W. (2017). A Markovian and cellular automata land-use change predictive model of the Usangu Catchment. *International Journal of Remote Sensing*, *38*(1), 64-81.
- Ilstedt, U., Malmer, A., Verbeeten, E., & Murdiyarso, D. (2007). The effect of afforestation on water infiltration in the tropics: a systematic review and meta-analysis. *Forest Ecology and Management*, 251(1), 45-51.
- IPCC. (2007). Summary for Policymakers. In: Climate Change 2007: Mitigation. Contribution of Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change [B. Metz, O.R. Davidson, P.R. Bosch, R. Dave, L.A. Meyer (eds)], Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- IPCC. (2012). Summary for Policymakers. In: Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Mastrandrea, K.J. Mach, G.-K. Plattner, S.K. Allen, M. Tignor, and P.M. Midgley (eds.)]. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK, and New York, NY, USA, pp. 1-19.
- Jiang, C., Xiong, L., Wang, D., Liu, P., Guo, S., & Xu, C. Y. (2015). Separating the impacts of climate change and human activities on runoff using the Budyko-type equations with time-varying parameters. *Journal of Hydrology*, *5*22, 326-338.

- Khoi, D. N., & Suetsugi, T. (2014). The responses of hydrological processes and sediment yield to land-use and climate change in the Be River Catchment, Vietnam. *Hydrological Processes*, 28(3), 640-652.
- Kilonzo, F. N. (2013). Assessing the Impacts of Environmental Changes on the Water Resources of the Upper Mara, Lake Victoria Basin. Ph.D, Vrije Universiteit Brussel.
- Klemeš, V. (1986). Operational testing of hydrological simulation models. *Hydrological Sciences Journal*, *31*(1), 13-24.
- Kupfer, J. A. (2012). Landscape ecology and biogeography: rethinking landscape metrics in a post-FRAGSTATS landscape. *Progress in Physical Geography*, 36(3), 400-420.
- Legesse, D., Vallet-Coulomb, C., & Gasse, F. (2003). Hydrological response of a catchment to climate and land use changes in Tropical Africa: case study South Central Ethiopia. *Journal of Hydrology*, 275(1), 67-85
- Li, J., & Zhou, Z. X. (2015). Coupled analysis on landscape pattern and hydrological processes in Yanhe watershed of China. Science of The Total Environment, 505, 927-938.
- Li, Z., Liu, W. Z., Zhang, X. C., & Zheng, F. L. (2009). Impacts of land use change and climate variability on hydrology in an agricultural catchment on the Loess Plateau of China. *Journal of Hydrology*, 377(1), 35-42.
- Lin, Y. P., Hong, N. M., Wu, P. J., Wu, C. F., & Verburg, P. H. (2007). Impacts of land use change scenarios on hydrology and land use patterns in the Wu-Tu watershed in Northern Taiwan. *Landscape and Urban Planning*, 80(1), 111-126.
- Liu, Y., Zhang, X., Xia, D., You, J., Rong, Y., & Bakir, M. (2011). Impacts of land-use and climate changes on hydrologic processes in the Qingyi River watershed, China. *Journal of Hydrologic Engineering*, 18(11), 1495-1512.
- Lørup, J. K., Refsgaard, J. C., & Mazvimavi, D. (1998). Assessing the effect of land use change on catchment runoff by combined use of statistical tests and hydrological modelling: case studies from Zimbabwe. *Journal of Hydrology*, 205(3), 147-163.
- Mango, L. M., Melesse, A. M., McClain, M. E., Gann, D., & Setegn, S. G. (2011). Land use and climate change impacts on the hydrology of the upper Mara River Basin, Kenya: results of a modeling study to support better resource management. *Hydrology and Earth System Sciences*, 15(7), 2245.
- Marfai, M. A., King, L., Singh, L. P., Mardiatno, D., Sartohadi, J., Hadmoko, D. S., & Dewi, A. (2008). Natural hazards in Central Java Province, Indonesia: an overview. *Environmental Geology*, 56(2), 335-351.
- Marhaento, H., Booij, M. J., Rientjes, T. H. M., & Hoekstra, A. Y. (2017a). Attribution of changes in the water balance of a tropical catchment to land use change using the SWAT model. *Hydrological Processes*, *31*(11), 2029-2040.
- Marhaento, H., Booij, M. J., & Hoekstra, A. Y. (2017b). Attribution of changes in streamflow to land use change and climate change in a mesoscale tropical catchment in Java, Indonesia. *Hydrology Research*, 48(4), 1143-1155.
- Marhaento, H., Booij, M. J., & Hoekstra, A. Y. (2017c). Hydrological response to future land use change and climate change in a tropical catchment. Under review in *Hydrological Sciences Journal*
- McGarigal, K., Cushman, S. A., & Ene, E. (2012). FRAGSTATS v4: Spatial Pattern Analysis Program for Categorical and Continuous Maps. Computer software program produced by the authors at the University of Massachusetts, Amherst.

Available at the following web site: http://www.umass.edu/landeco/research/fragstats/fragstats.html

- McGarigal, K., Marks, B. J. (1995). FRAGSTATS: Spatial pattern analysis program for quantifying landscape structure. Gen. Tech. Rep. PNW-GTR-351. Portland, OR: U.S. Department of Agriculture, Forest Service, Pacific Northwest Research Station. 122 p.
- Mejía, A. I., & Moglen, G. E. (2009). Spatial patterns of urban development from optimization of flood peaks and imperviousness-based measures. *Journal of Hydrologic Engineering*, 14(4), 416-424.
- Memarian, H., Balasundram, S. K., Abbaspour, K. C., Talib, J. B., Boon Sung, C. T., & Sood, A. M. (2014). SWAT-based hydrological modelling of tropical land-use scenarios. *Hydrological Sciences Journal*, 59(10), 1808-1829.
- Menzel, L., Koch, J., Onigkeit, J., & Schaldach, R. (2009). Modelling the effects of landuse and land-cover change on water availability in the Jordan River region. Advances in Geosciences, 21, 73.
- Ministry of Environment, Republic of Indonesia. (2010). *Indonesia Second National Communication under the UNFCCC*. http://unfccc.int/resource/docs/natc/indonc2.pdf [access in October 19th 2017]
- Ministry of Forestry (1987). A guideline of land rehabilitation and soil conservation. Ministry of Forestry, Jakarta, Indonesia. Available in Bahasa Indonesia.
- Moehansyah, H., Maheshwari, B. L., & Armstrong, J. (2002). Impact of land-use changes and sedimentation on the Muhammad Nur Reservoir, South Kalimantan, Indonesia. *Journal of Soils and Sediments*, 2(1), 9-18.
- Montanari, A., & Di Baldassarre, G. (2013). Data errors and hydrological modelling: The role of model structure to propagate observation uncertainty. *Advances in Water Resources*, *51*, 498-504.
- Murphy, J. M., Sexton, D. M., Barnett, D. N., Jones, G. S., Webb, M. J., Collins, M., & Stainforth, D. A. (2004). Quantification of modelling uncertainties in a large ensemble of climate change simulations. *Nature*, 430(7001), 768-772.
- Myint, S. W., & Wang, L. (2006). Multicriteria decision approach for land use land cover change using Markov chain analysis and a cellular automata approach. *Canadian Journal of Remote Sensing*, 32(6), 390-404.
- Naylor, R. L., Battisti, D. S., Vimont, D. J., Falcon, W. P., & Burke, M. B. (2007). Assessing risks of climate variability and climate change for Indonesian rice agriculture. *Proceedings of the National Academy of Sciences*, *104*(19), 7752-7757.
- Neitsch, S. L., Arnold, J. G., Kiniry, J. R., & Williams, J. R. (2011). Soil and water assessment tool theoretical documentation version 2009. Texas Water Resources Institute.
- Ngo-Duc T, Tangang FT, Santisirisomboon J, Cruz F, Trinh-Tuan L, Nguyen-Xuan T, Phan-Van T, Juneng L, Narisma G, Singhruck P, Gunawan D, Aldrian E. (2016). Performance evaluation of RegCM4 in simulating extreme rainfall and temperature indices over the CORDEX-Southeast Asia region. *Int. J. Climatol.* 37(3): 1634–1647.
- Niehoff, D., Fritsch, U., & Bronstert, A. (2002). Land-use impacts on storm-runoff generation: scenarios of land-use change and simulation of hydrological response in a meso-scale catchment in SW-Germany. *Journal of Hydrology*, 267(1-2), 80-93

- Nobre, C. A., Sampaio, G., Borma, L. S., Castilla-Rubio, J. C., Silva, J. S., & Cardoso, M. (2016). Land-use and climate change risks in the Amazon and the need of a novel sustainable development paradigm. *Proceedings of the National Academy* of Sciences, 113(39), 10759-10768.
- O'Connell, P. E., Ewen, J., O'Donnell, G., & Quinn, P. (2007). Is there a link between agricultural land-use management and flooding?. *Hydrology and Earth System Sciences*, *11*(1), 96-107.
- Paulhus, J. L. H. & Kohler, M. A. (1952). Interpolation of missing precipitation records. *Monthly Weather Review*, 80 (5), 129-133
- Pike, J. G. (1964). The estimation of annual run-off from meteorological data in a tropical climate. *Journal of Hydrology*, 2(2), 116-123.
- Putra, N. A., & Han, E. (Eds.). (2014). Governments' Responses to Climate Change: Selected Examples From Asia Pacific. Springer.
- Rathjens, H., & Oppelt, N. (2012). SWATgrid: An interface for setting up SWAT in a gridbased discretization scheme. *Computers & Geosciences*, *45*, 161-167.
- Refsgaard, J.C. & Storm, B. (1995). MIKE SHE, in: Computer Models of Watershed Hydrology, edited by: Singh, V.P., Water Resources Publications, Highland Ranch, Colorado, 809-846
- Refsgaard, J.C., Alley, W.M., Vuglinsky, V.S. (1989). Methods for Distinguishing between Man's Influence and Climatic Effects on the Hydrological Cycle. IHP-III Project 6.3. Unesco, Paris.
- Remondi, F., Burlando, P., & Vollmer, D. (2016). Exploring the hydrological impact of increasing urbanisation on a tropical river catchment of the metropolitan Jakarta, Indonesia. Sustainable Cities and Society, 20, 210-221.
- Renner, M., Brust, K., Schwärzel, K., Volk, M., & Bernhofer, C. (2014). Separating the effects of changes in land cover and climate: a hydro-meteorological analysis of the past 60 yr in Saxony, Germany. *Hydrology and Earth System Sciences*, 18, 389-405.
- Renner, M., Seppelt, R., & Bernhofer, C. (2012). Evaluation of water-energy balance frameworks to predict the sensitivity of streamflow to climate change. *Hydrology* and Earth System Sciences, 16(5), 1419-1433.
- Resosudarmo, B., Nawir, A.A., Resosudarmo, I.A., & Subiman, N.L. (2012). Forest Land Use Dynamics in Indonesia. Working Paper No. 2012/01. Arndt Corden Department of Economics Crawford School of Economics and Government, ANU College of Asia and the Pacific.
- Retnowati S. (2012). Impact of Land Use Changes to Hydrological Condition, MSc Thesis. Graduate School of Universitas Sebelas Maret. Surakarta. Indonesia. Available in Bahasa Indonesia. https://digilib.uns.ac.id/. [access in February 5th 2017]
- Rientjes, T. H. M., Haile, A. T., Kebede, E., Mannaerts, C. M. M., Habib, E., & Steenhuis, T. S. (2011). Changes in land cover, rainfall and streamflow in Upper Gilgel Abbay catchment, Blue Nile basin–Ethiopia. *Hydrology and Earth System Sciences*, 15(6), 1979-1989.
- Rientjes, T.H.M., Muthuwatta, L.P., Bos, M.G., Booij, M.J., Bhatti, H.A. (2013). Multivariable calibration of a semi-distributed hydrological model using streamflow data and satellite-based evapotranspiration, *Journal of Hydrology*, 505, 276-290
- Rodriguez Suarez, J. A., Diaz-Fierros, F., Perez, R., & Soto, B. (2014). Assessing the influence of afforestation with *Eucalyptus globulus* on hydrological response from

a small catchment in northwestern Spain using the HBV hydrological model. *Hydrological Processes*, *28*(22), 5561-5572.

- Romanowicz, R. J., & Booij, M. J. (2011). Impact of land use and water management on hydrological processes under varying climatic conditions. *Physics and Chemistry* of the Earth, Preface, 36(13), 613-614.
- Saha, S., et al. (2010). NCEP Climate Forecast System Reanalysis (CFSR) 6-hourly Products, January 1979 to December 2010. Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory. http://dx.doi.org/10.5065/D69K487J. [access in March 4th 2016].
- Sahin, V., & Hall, M. J. (1996). The effects of afforestation and deforestation on water yields. *Journal of Hydrology*, *178*(1-4), 293-309.
- Saxton, K., & Willey, P. (2005). The SPAW model for agricultural field and pond hydrologic simulation. *Mathematical Modeling of Watershed Hydrology*, 401-408.
- Setegn, S.G., Rayner, D., Melesse, A.M., Dargahi, B. & Srinivasan, R. (2011). Impact of climate change on the hydroclimatology of Lake Tana basin, Ethiopia. *Water Resources Research*, 47, W04511, doi:10.1029/2010WR009248.
- Shrestha, S., & Htut, A. Y. (2016). Land Use and Climate Change Impacts on the Hydrology of the Bago River Basin, Myanmar. *Environmental Modeling &* Assessment, 21(6), 819-833.
- Slater, R et al. (2007). Climate Change, Agricultural Policy and Poverty Reduction How Much Do We Know? Natural Resource Perspectives #109 of the Overseas Development Institute, UK.
- Smith, M., & Allen, R. G. (1992). Report on the Expert Consultation on Revision of FAO Methodologies for Crop Water Requirements: Held in FAO, Rome, Italy, 28-31 May 1990. Food and Agriculture Organization of the United Nations, Land and Water Development Division.
- Su, W., Ye, G., Yao, S., & Yang, G. (2014). Urban land pattern impacts on floods in a new district of China. *Sustainability*, *6*(10), 6488-6508.
- Subarna, D., Purwanto, M.Y.J., Murtilaksono, K., Wiweka. (2014). The relationship between monthly rainfall and elevation in the Cisangkuy watershed Bandung regency. *International Journal of Latest Research in Science and Technology*, 3(2), 55-60.
- Suryatmojo, H., Masamitsu, F., Kosugi, K., & Mizuyama, T. (2011). Impact of selective logging and intensive line planting system on runoff and soil erosion in a Tropical Indonesia rainforest. *Proceedings of River Basin Management VI*, 288-300.
- Syaukat, Y. (2011). The Impact of climate change on food production and security and its adaptation programs in Indonesia. *Journal of the International Society for Southeast Asian Agricultural Sciences*, 17(1), 40-51.
- Teutschbein, C., & Seibert, J. (2012). Bias correction of regional climate model simulations for hydrological climate-change impact studies: Review and evaluation of different methods. *Journal of Hydrology*, 456, 12-29.
- Thanapakpawin, P., Richey, J., Thomas, D., Rodda, S., Campbell, B., & Logsdon, M. (2007). Effects of landuse change on the hydrologic regime of the Mae Chaem river basin, NW Thailand. *Journal of Hydrology*, 334(1), 215-230
- Themeßl, M. J., Gobiet, A., & Heinrich, G. (2012). Empirical-statistical downscaling and error correction of regional climate models and its impact on the climate change signal. *Climatic Change*, 112(2), 449-468.

- Thomson, A. M., Calvin, K. V., Chini, L. P., Hurtt, G., Edmonds, J. A., Bond-Lamberty, B., Frolking, S., Wise, M. A., & Janetos, A. C. (2010). Climate mitigation and the future of tropical landscapes. *Proceedings of the National Academy of Sciences*, 107(46), 19633-19638.
- Tomer, M. D., & Schilling, K. E. (2009). A simple approach to distinguish land-use and climate-change effects on watershed hydrology. *Journal of Hydrology*, 376(1), 24-33.
- USGS. (2016). Landsat products courtesy of the U.S. Geological Survey. http://landsat.usgs.gov [access in January 19th 2016]
- USGS. (2017). Landsat products courtesy of the U.S. Geological Survey. http://landsat.usgs.gov [access in July 9th 2017]
- Valentin, C., Agus, F., Alamban, R., Boosaner, A., Bricquet, J. P., Chaplot, V. T. de Guzman, A. de Rouw, J.L. Janeau, D. Orange, K. Phachomphonh, Do Duy Phai, P. Podwojewski, O. Ribolzi, N. Silvera, K. Subagyono, J.P. Thiébaux, Tran Duc Toan, T. Vadari. (2008). Runoff and sediment losses from 27 upland catchments in Southeast Asia: Impact of rapid land use changes and conservation practices. *Agriculture, Ecosystems & Environment*, 128(4), 225-238.
- Van den Tillaart, S. P., Booij, M. J., & Krol, M. S. (2013). Impact of uncertainties in discharge determination on the parameter estimation and performance of a hydrological model. *Hydrology Research*, 44(3), 454-466.
- Van Dijk, A. I., Hairsine, P. B., Arancibia, J. P., & Dowling, T. I. (2007). Reforestation, water availability and stream salinity: A multi-scale analysis in the Murray-Darling Basin, Australia. *Forest Ecology and Management*, 251(1), 94-109.
- Van Griensven, A., Ndomba, P., Yalew, S., & Kilonzo, F. (2012). Critical review of SWAT applications in the upper Nile basin countries. *Hydrology and Earth System Sciences*, 16(9), 3371-3381.
- Van Griensven, A., Maharjan, S., & Alemayehu, T. (2014). Improved simulation of evapotranspiration for land use and climate change impact analysis at catchment scale. International Environmental Modelling and Software Society (iEMSs) 7th International Congress on Environmental Modelling and Software.
- Verburg, P. H., & Bouma, J. (1999). Land use change under conditions of high population pressure: the case of Java. *Global Environmental Change*, *9*(4), 303-312.
- Wagner, P. D., Kumar, S., & Schneider, K. (2013). An assessment of land use change impacts on the water resources of the Mula and Mutha Rivers catchment upstream of Pune, India. *Hydrology and Earth System Sciences*, 17(6), 2233-2246.
- Wang, X. (2014). Advances in separating effects of climate variability and human activity on stream discharge: an overview. Advances in Water Resources, 71, 209-218.
- Watanabe, M., Suzuki, T., O'ishi, R., Komuro, Y., Watanabe, S., Emori, S., Takemura, T., Chikira, M., Ogura, T., Sekiguchi, M., Takata, K., Yamazaki, D., Yokohata, T., Nozawa, T., Hasumi, H., Tatebe, H., Kimoto, M. (2010). Improved climate simulation by MIROC5: mean states, variability, and climate sensitivity. *Journal* of Climate, 23(23), 6312-6335.
- Wei, X., & Zhang, M. (2010). Quantifying streamflow change caused by forest disturbance at a large spatial scale: a single watershed study. Water Resources Research, 46(12).
- Wheater, H., & Evans, E. (2009). Land use, water management and future flood risk. Land Use Policy, 26, S251-S264.

- Wigmosta, M. S., Vail, L. W., & Lettenmaier, D. P. (1994). A distributed hydrologyvegetation model for complex terrain. *Water Resources Research*, 30(6), 1665-1679.
- Wilk, J., & Hughes, D. A. (2002). Simulating the impacts of land-use and climate change on water resource availability for a large south Indian catchment. *Hydrological Sciences Journal*, 47(1), 19-30.
- Williams, J. R. (1995). Chapter 25: The EPIC model. 909-1000. In V.P. Singh (ed). Computer models of watershed hydrology. Water Resources Publications, Highland Ranch, CO.
- Wittenberg, H. (1994). Nonlinear analysis of flow recession curves. IAHS Publications-Series of Proceedings and Reports-Intern Assoc Hydrological Sciences, 221, 61-68.
- Wohl, E., Barros, A., Brunsell, N., Chappell, N. A., Coe, M., Giambelluca, T., Goldsmith, S., Harmon, R., Hendrickx, J.M.H., Juvik, J., McDonnell, J., Ogden, F. (2012). The hydrology of the humid tropics. *Nature Climate Change*, 2(9), 655-662
- Yan, R., Huang, J., Wang, Y., Gao, J., & Qi, L. (2015). Modeling the combined impact of future climate and land use changes on streamflow of Xinjiang Basin, China. *Hydrology Research*, 47(2), 356-372
- Ye, X., Zhang, Q., Liu, J., Li, X., & Xu, C. Y. (2013). Distinguishing the relative impacts of climate change and human activities on variation of streamflow in the Poyang Lake catchment, China. *Journal of Hydrology*, 494, 83-95.
- Zhan, C., Niu, C., Song, X., & Xu, C. (2013). The impacts of climate variability and human activities on streamflow in Bai River basin, northern China. *Hydrology Research*, 44(5), 875-885.
- Zhang, G., Guhathakurta, S., Dai, G., Wu, L., & Yan, L. (2013). The control of land-use patterns for stormwater management at multiple spatial scales. *Environmental Management*, 51(3), 555-570
- Zhang, L., Nan, Z., Xu, Y., & Li, S. (2016). Hydrological impacts of land use change and climate variability in the headwater region of the Heihe River Basin, Northwest China. *PloS one*, *11*(6), e0158394.
- Zhang, M., Wei, X., Sun, P., & Liu, S. (2012). The effect of forest harvesting and climatic variability on runoff in a large watershed: The case study in the Upper Minjiang River of Yangtze River basin. *Journal of Hydrology*, 464, 1-11.
- Zhang, X., Booij, M.J. and Xu, Y.P. (2015). Improved simulation of peak flows under climate change: post-processing or composite objective calibration? *Journal of Hydrometeorology*, 16, 2187-2208.
- Zhang, Y., Guan, D., Jin, C., Wang, A., Wu, J., & Yuan, F. (2014). Impacts of climate change and land use change on runoff of forest catchment in northeast China. *Hydrological Processes*, 28(2), 186-196.



Hero Marhaento was born in Yogyakarta, Indonesia on April 5th, 1982. He finished his elementary school in SD Tarakanita in 1994, junior high school in SMPN 5 Pawitikra in 1997 and senior high school in SMAN 3 Padmanaba in 2000. Afterward, he continued his studies in Forest Management at Universitas Gadjah Mada (UGM), Indonesia. In 2005 he obtained his BSc degree *cum laude* with a final project on the topic of "land suitability analysis for forest rehabilitation using remote sensing and GIS". In 2007 he obtained his MSc degree in

Environmental Studies focusing in Geo-Information for Disaster Management from the joint-program between UGM and International Institute for Geo-Information Science and Earth Observation (ITC), The Netherlands. For his final MSc project, he worked on the topic of "landslide hazard analysis using heuristicstatistic method in combination with multi-temporal landslide data".

Hero worked for the Center for Disaster Studies (PSBA) UGM in 2005 and Ministry of Environment for Java Region (PPLH Regional Jawa) in 2006. Since 2009, Hero has been working as a lecturer in the Forest Resources Conservation department, Faculty of Forestry, UGM. In September 2013, he started his PhD in the Water Engineering and Management group, University of Twente, The Netherlands funded by DIKTI-scholarship from the Directorate General of Higher Education, Ministry of Research, Technology and Higher Education of the Republic of Indonesia. While doing his PhD, he chaired the Universitas Gadjah Mada Alumni Association in the Netherlands (KAGAMA-NL) in 2015-2017. He will return to his position as a lecturer in the Faculty of Forestry, UGM after finishing his PhD.

List of publications

Papers in peer-reviewed journal

- Marhaento, H., Booij, M. J., Rientjes, T. H. M., & Hoekstra, A. Y. (2017). Attribution of changes in the water balance of a tropical catchment to land use change using the SWAT model. *Hydrological Processes*, *31*(11), 2029-2040. DOI: 10.1002/hyp.11167
- Marhaento, H., Booij, M. J., & Hoekstra, A. Y. (2017). Attribution of changes in streamflow to land use change and climate change in a mesoscale tropical catchment in Java, Indonesia. *Hydrology Research*, 48(4), 1143-1155. DOI: 10.2166/nh.2016.110
- Marhaento, H., Booij, M. J., & Hoekstra, A. Y. Hydrological response to future land use change and climate change in a tropical catchment. *Submitted*
- Marhaento, H., Booij, M. J., Rientjes, T. H. M., & Hoekstra, A. Y. Sensitivity of streamflow characteristics to different spatial land use configurations in a tropical catchment. *Submitted*

Conferences

- Marhaento, H., Booij, M. J., & Hoekstra, A. Y. (2015). Quantifying the contribution of land use and climate change to stream flow alteration in tropical catchments. Presented at European Geosciences Union (EGU) General Assembly 2015, 12-17 April, Vienna Austria, Vienna, Austria.
- Marhaento, H., Booij, M. J., Rientjes, T.H.M., & Hoekstra, A. Y. (2017). Simulation of land use change impacts on hydrological processes in a tropical catchment. Presented at European Geosciences Union (EGU) General Assembly 2017, 23-28 April, Vienna Austria, Vienna, Austria.
- Marhaento, H., Booij, M. J., & Hoekstra, A. Y. (2017). Simulation of future land use change and climate change impacts on hydrological processes in a tropical catchment. Presented at American Geosciences Union (AGU) General Assembly 2017, 11-15 December, New Orleans, United States of America
- Marhaento, H., Booij, M. J., Rientjes, T.H.M., & Hoekstra, A. Y. Simulation of land use configuration impacts on stream flow characteristics of a tropical catchment. European Geosciences Union (EGU) General Assembly 2018, 8-13 April, Vienna Austria, Vienna, Austria. Submitted
- Schipper, T.C., Booij, M.J., & Marhaento, H. Distinguishing climate and land use change impacts on streamflow for 472 catchments in the United States and Australia. European Geosciences Union (EGU) General Assembly 2018, 8-13 April, Vienna Austria, Vienna, Austria. Submitted

Land use and climate changes are key factors determining changes in hydrological processes in catchments. For tropical countries like Indonesia, the effects of changes in land use and climate have been projected to cause a food crisis and eventually increase the degree of poverty in the future. In order to mitigate future water-related risks, knowledge on the extent and directions of land use change and climate change impacts on water availability is essential. This book provides insight in the impacts of land use and climate changes on water availability in the Samin catchment in Indonesia for past and future conditions.

ISBN: 978-90-365-4491-7 UNIVERSITY OF TWENTE.